

Michiel Th. Voûte

Financial support for the publication of this thesis was kindly provided by

Stichting Lijf en Leven Afdeling Heelkunde Erasmus MC Erasmus Universiteit Rotterdam Medtronic Inc. Vascutek Nederland W. L. Gore & Associates B.V. **Krijnen Medical Innovations** Sigma Medical B.V. Chipsoft Covidien Nederland BV Norgine Angiodynamics Nederland C.R. Bard Inc. Cook Medical Endologix Maquet Vascular Olympus

ISBN: 978-94-6169-603-8 © Michiel Th. Voûte, 2015

Coverdesign & Lay-out: MT Voûte Printed by: Optima Grafische Communicatie, Rotterdam

No part of this thesis may be reproduced or transmitted in any form or by any means without prior permission of the author, Michiel Th. Voûte, or when appropriate, of the scientific journals in which substantial parts of this thesis have been published.

MEDICAL & TECHNICAL CONSIDERATIONS IN VASCULAR SURGERY

MEDISCHE & TECHNISCHE OVERWEGINGEN IN DE VAATCHIRURGIE

PROEFSCHRIFT

ter verkrijging van de graad van doctor aan de Erasmus Universiteit Rotterdam op gezag van de rector magnificus

Prof.dr. H.A.P. Pols en volgens besluit van het College voor Promoties.

De openbare verdediging zal plaatsvinden op woensdag 14 januari 2015 om 15.30 uur

MICHIEL THOMAS VOÛTE

geboren te Amsterdam, 1981

ERASMUS UNIVERSITEIT ROTTERDAM

PROMOTIECOMMISSIE

Promotores:

Prof.dr. H.J.M. Verhagen Prof.dr. R.J. Stolker

Overige leden:

Prof.dr. F. Zijlstra Prof.dr. E.J.G. Sijbrands Prof.dr. F.L. Moll (UMCU)

Financial support by the Dutch Heart Foundation for the publication of this thesis is gratefully acknowledged

CONTENTS

	Preface		7	
	Introdu	ction	8	
Medical Considerations in Vascular Surgery				
	1:	Blood thinners	13	
	2:	Statins	27	
	3:	Beta-blockers	41	
	4:	Vitamin D status	63	
	5:	New-onset arrhythmias	77	
Technical Considerations in Vascular Surgery				
	6:	Carotid stent characteristics	91	
	7:	Aortic stentgraft composition	101	
	8:	Laparoscopic sac fenestration after EVAR	115	
	9:	Need we intervene for type II endoleaks?	127	
Appendices				
	Summary and conclusions			
	Nederlandse samenvatting			
	List of publications			
	Acknowledgements			
	PhD Portfolio			
	Curricu	160		

PREFACE

Evidence based medicine has led to a high demand for researchers to continuously improve health care. The current expectations of an applicant for surgical training have led to a large flock of researchers, all with future surgical training positions in mind.

After successfully writing a thesis or dissertation one acquires the title of *Doctor of Philosophy (PhD)*, from the Ancient Greek *Philosophia* which means "love of wisdom". This doctorate degree is the highest academic degree that one can acquire at a Dutch university.

Are all surgical trainees, in possession of a doctorate, madly in love with acquiring wisdom? Do all of them aspire to an academic career? Is this highest academic degree still a unique achievement, a rarity which deserved the utmost respect?

The first marathon was run in 490 B.C. by a single soldier. Today, over half a million people finish a marathon each year in the U.S. alone. That doesn't mean that each runner does not put in the hours of preparation, make the extensive effort and perform at his best up to the finish line.

I never ran a marathon. I did however, write this thesis.

Thank you for reading.

Michiel Voite

INTRODUCTION

It is estimated that in the year 2020 more than a million non-cardiac, vascular surgery procedures will be performed annually in Europe. The same applies for the US. With this growing application of high-risk procedures, it is important to continuously strive for improved patient safety and surgical outcome.

Regarding patient safety, it is key to recognize a change in patient populations and patient selections. Due to anesthetic and medical improvements as well as surgical innovation, the vascular surgery population is increasing in age, tends to have more comorbidity and more often uses multiple prescription drugs. In other words, it is a fragile population, undergoing high-risk surgery.

Cardiovascular adverse events are responsible for a major proportion of morbidity and mortality in patients undergoing vascular surgery, with an estimated incidence of more than 5% of ischemic complications, and up to 10% new-onset arrhythmias in major vascular surgery. In the first part of this thesis, certain patient safety issues, secondary prevention and future perspectives are discussed, regarding these cardiovascular complications.

Medical considerations

In Chapter 1, a comprehensive outline is provided of the optimal medical management, in regard to the fine balance between bleeding and thrombotic complications in any surgical procedure. A number of drugs are discussed that can help reduce cardiac ischemic complications, such as statins and beta-blockers.

The pros and cons of statin use in high-risk surgery is further elaborated in Chapter 2. Specifically, the clinical applications of the extended release fluvastatin is evaluated.

Chapter 3 is a literature review on the safety of beta-blockade in surgery patients. The aphorism to 'start low and go slow' is presented as a guide to safely introduce beta-blockade in the perioperative period.

In an effort to find new prevention strategies against cardiovascular disease, the relationship between patients' vitamin D status and the severity of arterial disease is reported in Chapter 4.

Having only looked into ischemic complications so far, Chapter 5 presents a pilot study with an insertable cardiac monitor, to reveal the true incidence of new-onset cardiac arrhythmias in vascular surgery patients, atrial fibrillation (AF) in particular. New-onset AF after vascular surgery has a reported incidence of 4-13%, is a known major risk factor for postoperative stroke, myocardial infarction and pulmonary embolism, and it can be treated if diagnosed. Traditional monitoring methods are intermittent, and may possibly fail to detect a proportion of patients at high risk for thromboembolic complications. Can we improve detection, and therefore patient safety and outcome?

Technical considerations

Since the introduction of endovascular techniques, carotid artery disease is often treated with angioplasty and stenting. Literature suggests that the radial force of a carotid stent is related to the clinical outcome, in terms of short term emboli and long-term patency. In Chapter 6 a comparison is made of the radial force of four widely-used carotid stents, by testing them in a purpose-built measuring device at the Technical University in Delft, The Netherlands.

Another application of endovascular surgery is the endovascular repair of abdominal aortic aneurysms (EVAR), first introduced by Dr. Juan Parodi in 1991. New technology brings new complications and clinical entities. The post-implantation syndrome is one of these entities. Why and how often does this flu-like syndrome occur after EVAR? Chapter 7 explores the difference in incidence, depending on the type of graft material that was implanted.

Another complication after EVAR, unknown with open aneurysm repair, is the occurrence of endoleaks. In open repair, branching arteries from the aneurysm are inevitably ligated after clamping the aorta and opening the aneurysm sac, to reduce blood loss and improve sight. In EVAR procedures, the aneurysm sac is not opened, and

side-branches such as lumbar arteries are left untouched. Retrograde flow in these lumbar arteries can therefore reach the excluded sac, a phenomenon called a type II endoleak, and cause expansion of the aneurysm. Hypothetically, this could lead to a reduced seal of the graft and threaten patient safety and surgical outcome.

Between 1999 and 2005, a series of patients underwent a laparoscopic procedure, in which side-branches were clipped and the aneurysm sac was fenestrated, to reduce pressure. The aim was to stop sac growth. Chapter 8 presents long-term follow-up of this somewhat experimental approach.

The importance of a type II endoleak has often been subject of discussion in scientific literature. Conflicting data on the natural history of type II endoleak have been published. There is no consensus on the threshold for treatment of type II endoleak and controversy exists about the optimal treatment modality. Chapter 9 discusses the current evidence behind treating type II endoleak and investigates the need for treatment. Perhaps, in retrospect, there was never any need to perform additional surgery on the patients discussed in the previous chapter?

10

Medical Considerations

Blood thinners

Statins Beta-blockers Vitamin D status New-onset arrhythmias Carotid stent characteristics Aortic stentgraft composition Laparoscopic sac fenestration after EVAR Need we intervene for type II endoleaks?

Published as: **"Optimal medical management around the time of surgery"** *Heart 2010;96:1842-1848*

Michiel T. Voûte Tamara A. Winkel Don Poldermans

ABSTRACT

This paper provides a comprehensive outline of the optimal perioperative medical management, concerning cardiac risk in any surgical population, based on recent guidelines. Special attention is paid to the fine balance between bleeding complications and thrombotic complications, as well as medical therapy to reduce the risk of an oxygen supply-demand mismatch in the perioperative period.

The authors emphasize that knowledge of, and adherence to current guidelines is essential for optimal care and safety of surgical patients.

MANUSCRIPT

INTRODUCTION

Undergoing surgery imposes certain risks on the patient, such as impaired wound healing, bleeding complications, perforations, nerve damage and infection. These are all well-known points of attention when planning a surgical procedure. Apart from these apparent risks directly related to the target area of the surgical procedure, there are other surgery-related factors contributing to the outcome, which tend to be overlooked by the treating physicians. Of all contributors to poor postoperative outcome, cardiac complications are the most important, including myocardial infarction (MI), congestive heart failure (CHF), stroke and arrhythmias. An estimated 10-40% of postoperative mortality is attributed to myocardial infarction. With over 230 million surgical procedures being performed worldwide each year, the perioperative period is also a golden hour to initiate secondary prevention. Identification of risk factors for perioperative cardiac adverse events and risk stratification of surgery patients are the pillars for reducing cardiovascular complication rates.

The overall theme for perioperative care is to find the balance between risk reduction strategies, with a potential delay of the index surgical procedures, and its impact on the operation. For instance, how to handle the controversy between hemorrhagic control and prevention of thrombosis related complications, such as MI and stroke. Patients with antiplatelet therapy represent a surgical challenge, in terms of bleeding risk, while withdrawal can increases the risk of coronary thrombosis. In this paper we aim to provide a comprehensive overview of the optimal medical management around the time of surgery, shedding light on all key issues.

PATHOFYSIOLOGY OF SURGICAL RISKS

Surgery causes physiological changes, affecting multiple organs apart from the primary target of the procedure. These changes can increase myocardial oxygen demand and reduce supply because of thrombosis, both leading to (fatal) myocardial ischemia. The most important contributors to these pathophysiological pathways will be outlined briefly.

Stress and the oxygen mismatch

In the perioperative period, surgical stress induces a catecholamine surge, prompted by incisional tissue injury and mediated by neuroendocrine factors. The surgical stress causes an increase in heart rate and myocardial contractility, leading to an increased myocardial oxygen demand. Subsequently, an oxygen supply-demand mismatch can occur in patients with coronary artery disease. Another pathway of stress causing perioperative ischemia is plaque rupture. The stress-induced, increased mechanical activity of the heart can lead to shear stress on coronary plaques, increasing plaque instability and subsequent rupture or emboli. Plaque rupture and emboli decrease the oxygen supply, increasing the oxygen mismatch, and the risk for myocardial infarction.

Inflammation and the oxygen mismatch

Apart from focal damage, surgery induces an generalized inflammatory response, with an increase of circulating C-reactive protein (CRP), interleukin-6 (IL-6) and tumor necrosis factor-alpha (TNF- α). This response is further increased by the use of general anesthesia. The progression of atherosclerosis can be propelled by increased IL-6 and other inflammatory cytokines, leading to the genesis of more lesions and growth of existing atherosclerotic plaques. Increased levels of inflammation have also been linked to coronary plaque vulnerability and rupture.[1] In short, the surgical inflammatory response can lead to a narrowing of the coronary lumina, and possible plaque rupture, both decreasing the oxygen supply to the myocardium.

Laparoscopy and thrombosis

Laparoscopic procedures cause minimal incisional tissue damage, but have another pathway leading to an increased cardiac risk. The increase in intra-abdominal pressure – due to the pneumoperitoneum that is used – reduces the venous return and therefore decreases cardiac output, and increased systemic vascular resistance. The decreased flow velocity that follows increases the risk of thrombus formation and growth. Taken this into account, minimally invasive surgery patients should be regarded as equal to open surgery patients, in terms of cardiac risk stratification.

Hypercoagulability and thrombosis

Another consequence of surgery is a change in the balance of prothrombotic versus fibrinolytic factors. Platelet activation and aggregation, elevation of coagu-lation factors (e.g. fibrinogen) on one side, and a decrease in fibrinolysis on the other can result in a state of hypercoagulability, increasing the risk of coronary artery thrombosis. This process predisposes for myocardial ischemia and heart failure during and after the surgical procedure.



PREOPERATIVE RISK ASSESSMENT

An adequate risk assessment can serve multiple purposes. The predicted risk guides the preoperative workup, such as initiation of risk reduction strategies, and helps decide the selection of the best surgical and anesthesiological techniques. Preoperatively, both the risk of the surgical procedure as the cardiac risk of the individual patient should be taken into account. A distinction is made between surgical procedures with a high, an intermediate and a low cardiovascular risk. Based on estimates from Boersma et al., high risk surgery has an estimated 30-day postoperative cardiac event rate of >5%, intermediate risk surgery a 1-5% event rate and low risk surgery <1%.[2] The risk of cardiac complications should be assessed for each individual patient. Useful risk factors

were identified by Lee et al. in prospectively gathered data of 2893 patients, undergoing a variety of surgical procedures.[3] Lee's risk index includes ischemic heart disease (e.g. angina pectoris or myocardial infarction), heart failure, stroke, diabetes, renal dysfunction and surgical risk. With equal contributions of each risk factor, the incidence of cardiac complications increases with each added risk factor. In their paper, Lee et al. found that in patients with 0, 1, 2 and \geq 3 risk factors, the incidence of cardiac complications was 0.4%, 0.9%, 7% and 11% respectively. The predicted perioperative cardiac risk guides the initiation or continuation of medical therapies.

ANTIPLATELET MANAGEMENT AND THROMBOSIS

Advances in interventional cardiology, radiology and endovascular surgery have resulted in an increasing number of patients receiving antiplatelet therapy. Recommendations for antiplatelet therapy include treatment regiments in patients with, among others, atrial fibrillation (AF), coronary artery disease (CAD), acute coronary syndrome (ACS), cerebrovascular disease and chronic peripheral arterial disease, as well as patients that have been treated with bare-metal stents (BMS), drug-eluting stents (DES), endovascular prostheses or via carotid endarterectomy (CEA).[4] According to existing guidelines, duration of antiplatelet therapy can vary from 6 weeks (e.g. after BMS placement) to a lifelong continuation for some indications.[5] Guidelines even recommend dual antiplatelet therapy (i.e. aspirin and clopidogrel) for patients with non-ST-segment elevation ACS. High-risk patients with recurrent ischemia, ST-segment depression, troponin release and diabetes may also receive a Gp IIb/IIIa receptor inhibitor, on top of aspirin and clopidogrel, as triple therapy.

Surgeons must beware of the balance between bleeding risk, when performing surgery under antiplatelet therapy, and the risk for thrombotic complications when discontinuing the antiplatelet agents. Perioperative withdrawal from single antiplatelet therapy precedes 10% of cardiovascular events, according to a meta-analysis by Burger et al.[6] Collet et al. found that in patients treated with antiplatelets for having received a DES or BMS, interruption of antiplatelet therapy led to death in up 25 to 50 per cent.[7] Furthermore, surgery-related inflammatory response and hypercoagulability may be especially hazardous in these thrombogenic patients. Therefore, it is essential to have an understanding on how to manage a patient with current antiplatelet therapy, when considering a surgical procedure.

Aspirin

The most widely prescribed antiplatelet agent is acetylsalicylic acid, or aspirin. Aspirin is an antithrombotic agent that acetylates part of cyclooxigenase (COX) 1. This inhibits the release of thromboxane A_2 , which acts as a stimulator of platelet activation. Since platelets are unable to generate new COX 1, the affected platelets are impaired for the duration of their life.

Surgeons tend to instruct patients to stop taking aspirin no less than a week before surgery, concerned of bleeding complications. For example, a 1.5 fold increase in the risk of bleeding complications was reported in a meta-analysis by Burger et al., without an increase in the severity of hemorrhagic complications. However, in patients at risk

for or with ischemic heart disease, perioperative aspirin withdrawal was associated with a 3 fold higher risk for major adverse cardiac events.[8] There are no specific guidelines on treatment of major bleeding in surgical patients under current aspirin treatment, but discontinuation of aspirin and platelet transfusions are possible resorts in case of emergency.

Before planning surgery, the procedural bleeding risk and the individual risk of ischemic events should be carefully weighed. In most cases, including minor and endoscopic surgery, continuing aspirin perioperatively is safe and advisable. Except for prostatectomy and intra-cranial surgery, low-dose aspirin is not associated with an increased severity of bleeding nor perioperative mortality because of bleeding complications. If a planned surgical procedure is expected to have very difficult haemostatic control, withdrawal from aspirin therapy 7 days prior to surgery can be considered. Not much guidance is available on preoperative aspirin withdrawal, nor on restarting aspirin postoperatively. Future publication of the ASPIRIN trial (Antiplatelet Strategies in the Perioperative Period in Patients at Risk of Ischemic Events) may provide definitive guidelines for patients taking aspirin in the perioperative period.[4]

Clopidogrel

Thienopyridines, of which clopidogrel is the most prescribed agent, have a different acting site on platelets, and therefore can be prescribed as double therapy besides aspirin, or as solitary prevention in patients with high thromboembolic risk (e.g. after coronary DES placement). The use of clopidogrel has been implemented in guidelines after percutaneous coronary interventions (PCIs) and (non)ST-segment elevation myocardial infarction (STEMI).[5] These indications illustrate the strong necessity of antiplatelet therapy, at the risk of (stent)thrombosis and cardiac ischemia. Meta-analysis showed that patients who prematurely discontinue clopidogrel treatment after coronary stent insertion are ten times more likely to die or be readmitted during the next year.[6] In comparison with aspirin, this new generation antiplatelet agent is more powerful, and has been called "a surgeon's headache", due to its capacity to cause bleeding.

If a patient under clopidogrel therapy is planned for an operation, surgeons tend to discontinue the clopidogrel treatment. However, consulting a cardiologist is warranted considering the high thrombotic risk of clopidogrel withdrawal. In a report by Wilson et al., early (<6 weeks after coronary stenting) surgery with premature clopidogrel withdrawal was associated with an incidence of death, MI or stent thrombosis of 4.8%, compared to no acute events in patients who underwent surgery >6 weeks after coronary stent placement.[9]

Collet and Montalescot designed an algorithm for patients receiving dual antiplatelet therapy after DES insertion, undergoing surgery.[7] In this approach, bleeding risk and stent thrombosis risk should be assessed by the surgeon, an anesthesiologist and a cardiologist. If there is a great risk for both bleeding and stent thrombosis, surgery should be postponed until the clopidogrel therapy is ended, as was supported by a review from Thachil et al.[10] If delay is inadmissible, clopidogrel should be

discontinued 5 days before surgery and aspirin should be continued. If the bleeding risk is small and there is a major risk of stent thrombosis, surgery should be performed under dual antiplatelet therapy.

If surgery is performed in a patient with current clopidogrel therapy and a severe bleeding occurs, the drug should then be discontinued in agreement with a cardiologist. To reverse the effect of clopidogrel in severe bleeding cases, platelet transfusions can be considered, as well as administration of antifibrinolytic agents or recombinant factor VIIa, although these recourses remain subject of further investigation.

Prasugrel

Prasugrel is a potent novel thienopyridine antiplatelet agent and has been shown in preclinical and clinical studies to achieve faster onset, higher levels of platelet inhibition, and less response variability than clopidogrel. This antiplatelet profile reflects more efficient generation of the active metabolite of prasugrel. As a consequence, the TRITON-TIMI 38 trial found that prasugrel was superior to clopidogrel as measured by the composite endpoint of cardiovascular death, nonfatal MI, or nonfatal stroke in patients with ACS undergoing PCI, but was associated with a minor increase in the risk of bleeding. Future studies may address the additional perioperative bleeding risk of prasugrel, and the thrombotic risk of perioperative withdrawal, in (non)cardiac surgery.

Gp IIb/IIIa inhibitors

The third major class of antiplatelet agents is inhibitors of the Gp IIb/IIIa receptor, which plays a key role in the linking of activated platelets and the formation of platelet thrombi. These agents (i.e. abciximab, cilostazol) have been tested in patients admitted with an acute coronary syndrome (ACS), patients undergoing thrombolytic therapy for acute myocardial infarction and patients undergoing percutaneous coronary intervention (PCI) or coronary artery bypass grafting (CABG).[4] Triple therapy has been reported to improve cardiac outcome and survival in patients with acute STEMI undergoing PCI.[11] Further studies and guidelines on the usefulness of Gp IIb/IIIa inhibitors in patients undergoing non-cardiac surgery have yet to be published.

BOX 1: Antiplatelet therapy and surgery

- Alterations in antiplatelet therapy can best be deliberated between surgeon, anaesthesist and cardiologist to ensure optimal balance between chance of bleeding and risk of ischemia
- Aspirin should be continued in surgery patients, unless there is a severe bleeding risk
- Clopidogrel should be discontinued before surgery
- However, if clinically admissible, delaying surgery until clopidogrel can be terminated safely is warranted
- If severe bleeding occurs in a surgical patients under clopidogrel treatment, discontinue clopidogrel in agreement with a cardiologist
- To counter the antiplatelet effect of clopidogrel, consider platelet transfusions
- More evidence is needed for guidelines on triple antiplatelet therapy around the time of non-cardiac surgery

ANTICOAGULANT MANAGEMENT AND THROMBOSIS

In patients treated with oral vitamin K antagonists (VKA), more commonly known as coumarins (e.g. warfarin, acenocoumarol, phenprocoumon), there is an increased risk of bleeding complications when performing non-cardiac surgery. In surgical procedures with an increased bleeding risk, anticoagulation should be discontinued. However, since these patients benefit from anticoagulant therapy, temporary cessation of these drugs can lead to thrombo-embolic events. There is a need for bridging therapy, consisting of low molecular weight heparin (LMWH) or unfractionated heparin (UFH). Thromboembolic risk is especially considered high in patients with atrial fibrillation (AF), mechanical prosthetic heart valves, biological prosthetic heart valves or mitral valvular repair within the last 3 months or recent venous thromboembolism (<3 months) plus thrombophilia, among other conditions.[12] For bridging therapy, current guidelines prescribe discontinuation of VKA no less than 5 days before surgery, due to the long lasting biological availability of these agents. There are differences between agents in pharmacogenetics. Phenprocoumon for instance has a longer lasting effect, and a more steady INR throughout treatment, than acenocoumarol. Therefore, guidelines on anticoagulant withdrawal and INR control are generally helpful, but INR control in the individual patient is nevertheless warranted.

BOX 2: Anticoagulants and bridging therapy

Low bleeding risk

• Continue anticoagulant therapy with INR in therapeutic range

Low thromboembolic risk, high bleeding risk

- Discontinue anticoagulants 5 days before surgery
- Start LMWH prophylaxis *once daily* or UFH intravenously 1 day after acenocoumarol interruption, and 2 days after warfarin interruption. Administer the last dose of LMWH at least 12 hours before the procedure or give UFH intravenously up to 4 hours to surgery.
- Resume LMWH or UFH at preprocedure dose 1 to 2 days (at least 12 hours) after the procedure according to hemostatic sufficiency. Resume VKA 1 to 2 days after surgery 150% of pre-procedure dose for 2 consecutive days according to hemostatic adequacy.
- Low molecular weight heparin or UFH is continued until the INR returned to therapeutic levels.

High thromboembolic risk, high bleeding risk

- Discontinue anticoagulants 5 days before procedure.
- Start therapeutic LMWH *twice daily* or UFH intravenously 1 day after acenocoumarol interruption, and 2 days after warfarin interruption. Administer the last dose of LMWH at least 12 hours before the procedure or give UFH intravenously up to 4 hours to surgery.
- Resume LMWH or UFH at preprocedure dose 1 to 2 days (at least 12 hours) after the procedure according to hemostatic adequacy. Resume VKA 1 to 2 days after surgery 150% of pre-procedure dosefor 2 consecutive days according to hemostatic sufficiency.
- Low molecular weight heparin or UFH is continued until the INR returned to therapeutic levels.

One day before surgery the INR should have decreased to below 2.0, so that it can be expected to reach a level <1.5 on the day of surgery. Generally, if the international normalized ratio (INR) is <1.5, any type of surgery can be performed safely.[10] Otherwise, an oral dose of 1-2 mg of vitamin K1 can be administered 24 hours before surgery, or considerations should be given to postponing the procedure.

The recommended daily dose of bridging LMWH is 70 anti-Xa U/kg, and should be administered subcutaneously. In high risk patients two daily doses should be administered. Box 2 provides a timeline for discontinuation of VKA, bridging therapy and postoperative restarting of heparin and VKA. As anticoagulation is restarted, extra attention should be paid to possible bleeding complications.

MEDICAL MANAGEMENT AND OXYGEN MISMATCH

Risk reduction in the surgery patient population can be achieved by coronary revascularisation and medication. The focus for intervention is on plaque stabilization and prevention of the oxygen supply/demand mismatch. Some perioperative medication (i.e. beta-blockers, statins) has been widely studied, and a good guideline on their treatment regiments has been published in Education in Heart by Schouten et al.[13] An update on the latest recommendations and developments on these agents will be discussed briefly in the following paragraphs.

Beta-blockers

The use of β -adrenoreceptor antagonists, or beta-blockers, is primarily aimed at reducing the heart rate, counteracting the effect of the catecholamine surge and reducing the oxygen supply-demand mismatch prevents myocardial ischemia to occur. A recent meta-analysis of 12306 patients underlined the effect of beta-blockade in high-risk surgery patients on all-cause mortality (63% decreased risk) and non-fatal myocardial infarction (44% decreased risk). In intermediate surgical-risk patients, 30% reduction of the risk of non-fatal MI was found, at the expense of an increased risk of all-cause mortality, non-fatal stroke and hypotension.[14] This was challenged by a recent RCT in intermediate surgical-risk patients, which showed improved perioperative outcome, without an increased rate of these complications.[15]

The introduction of esmolol, an ultra-short acting beta-blocker, has provided a new tool to better mitigate heart rate but prevent hypotension. A recent meta-analysis including 1765 patients showed a decrease of myocardial ischemia in noncardiac surgery, without an increase in the rate of hypotension or bradycardia.[16] In future guidelines, perhaps esmolol will take a prominent place in perioperative medical management. Dosage and timing of beta-blocker therapy have great influence on the treatment effect. According to guidelines, medical treatment should commence 30 days prior to surgery, with a low starting dose of 2.5 mg bisoprolol or 50 mg metoprolol succinate. Subsequently, dose can be titrated to the preferred range of resting heart rate, which is around 60-70 bpm.[12] Peri- and postoperatively, intravenous beta-blockade is warranted when oral administration is not possible.

BOX 3. Cardioprotective measures: Beta-blockers

- Beta-blockers are recommended in patients with ischemic heart disease or myocardial ischemia on preoperative stress testing
- Beta-blockers are highly recommended in high-risk surgery patients
- Beta-blockers are recommended in intermediate risk surgery patients
- Beta-blockers should be administered 30 days prior to surgery
- Beta-blockers should be titrated to heart rate 60-70 bpm
- Short acting beta-blockers could decrease hypotension and bradycardia rates

Statins

In the process of cholesterol synthesis, the enzyme 3-hydroxy-3-methylglutaryl coenzyme A reductase (HMGR) plays a crucial role. HMGR inhibitors, better known as statins, directly influence this process. Generally prescribed to patients as primary or secondary prevention of cardiac ischemia, statins are renowned for their lipid lowering effect. Additionally, statins have so-called pleiotropic effects, adding to their cardio-protective value. These include decreasing lipid oxidation, inflammation, matrix metalloproteinase and cell death, and increasing tissue inhibitor of metalloproteinase and collagen. It is these effects that may prevent plaque instability and subsequent rupture around the time of surgery.

Several reports on the effects of perioperative statin showed improved cardiac outcome, especially in high-risk surgery patients. Meta-analysis of 18 studies including a total of 799632 patients showed a 30% to 42% reduction of perioperative rates of death or ACS in patients taking statins.[17] However, large prospective studies had not been performed at the time. In the recently published DECREASE III trial by Schouten et al., an RCT including 497 vascular surgery patients, fluvastatin use (80 mg extended release) was associated with 10.8% myocardial ischemia, compared to 19.0% in the placebo group.[18]

Lacking evidence in intermediate and low risk surgery, perioperative statin therapy is not recommended for these types of surgery. However, discontinuation of statins may be harmful, independent of surgery. Therefore, perioperative statin continuation is recommended regardless of the type of surgery. Lacking intravenous substitutes, the use of statins with a prolonged half-life (e.g. atorvastatin or fluvastatin extended release) is preferable, to bridge the immediate postoperative period in which oral statin administration is impaired.[12]

BOX 4. Cardioprotective measures: Statins

- Perioperative continuation of current statin therapy is recommended
- Statins should be started 30 days prior to high-risk surgery
- Statins with prolonged half-life are preferable
- Statins should be restarted as soon as possible, following postoperative withdrawal

Angiotensin-converting enzyme inhibitors

Angiotensin-converting enzyme (ACE) inhibitors can be prescribed in patients with left ventricular (LV) systolic dysfunction (e.g. postmyocardial infarction) as well as for hypertensive patients. ACE-inhibitors have several effects on endothelial function and atherosclerosis, adding to the treatment effect in especially high risk patients. Regarding surgical patients, current studies do not support perioperative ACE-inhibitor therapy to have added cardioprotective value.[19] However, if a patient is receiving ACE-inhibitor therapy preoperatively for LV dysfunction instead of hypertension, this should not be discontinued. If LV dysfunction is discovered preoperatively, it is preferable to start ACE-inhibitor and beta-blocker therapy, and therefore temporarily postpone surgery.[20] Considering the type of surgery, starting perioperative ACE-inhibitor therapy for LV systolic dysfunction in stable patients is more strongly advised in high risk surgery than intermediate risk surgery.[12]

BOX 5. Cardioprotective measures: ACE-inhibitors

- Withdrawal from ACE-inhibitors for hypertension can be considered before non-cardiac surgery
- ACE-inhibitor therapy in stable patients with LV systolic dysfunction should be continued during noncardiac surgery
- Upon discovery of LV systolic dysfunction, ACE-inhibitors should be started before high risk surgery
- Upon discovery of LV systolic dysfunction, starting ACE-inhibitors can be considered before intermediate risk surgery

SUMMARY

Due to technical advancement and improved life expectancy, the surgical patient population is increasing in age and level of comorbidities. In an effort to decrease postoperative cardiac complications and death, optimal medical management is essential. Undergoing surgery has an accelerating effect on coronary atherosclerosis, increases inflammation and induces a state of hypercoagulability in patients. Therefore, cardioprotective measures should be taken, especially in patients with a high risk of cardiac complications after surgery. Furthermore, an increasing number of patients scheduled for surgery is treated with antiplatelet and/or anticoagulant therapy. These agents require strict management around the time of surgery, due to their ability to cause hemorrhage on one hand and the increased cardiac risks of withdrawal on the other hand.

As planning for surgery begins, cardioprotective measures are best initiated. Optimally, 30 days before surgery both beta-blockade and statin therapy are recommended to start. Especially in high cardiac risk patients, these medications have proven to be beneficial in the perioperative period and on long-term follow-up. Additionally, the use of antiplatelet therapy should be assessed. A cardiologist and an anesthesiologist should be consulted if the planned procedure has such high bleeding risk, that

withdrawal from antiplatelet therapy is considered by the surgeon. Antiplatelet withdrawal – especially clopidogrel – is often hazardous to the patient, and surgery should therefore be postponed until clopidogrel therapy has ended, if possible.

Patients with current anticoagulant treatment should discontinue their therapy 5 days before most types of surgery. This will reduce the risk of bleeding during surgery, but it will increase the risk for thrombosis. In general, low molecular weight heparin (LMWH) will be used as bridging therapy to reduce the perioperative thrombotic risk. LMWH therapy should commence one day after acenocoumarol or two days after warfarin, and be continued until 12 hours prior to surgery. One or two days, and certainly no less than 12 hours after surgery, LMWH bridging therapy can be continued. One or two days after surgery, anticoagulant therapy should be restarted at 150% of the preoperative daily dose for two days, and then continued at the preoperative daily dose. Heparin is discontinued when the INR reaches the therapeutic range.

This paper provides a comprehensive outline of the optimal perioperative medical management, concerning cardiac risk in any surgical population, based on recent guidelines. We emphasize that knowledge of, and adherence to current guidelines is essential for optimal care and safety of surgical patients.

The advices about the perioperative use of statins and beta-blockers in this and the two following chapters are in line with the recent new guideline of the American Heart Association.[21] The indication for triple therapy described in Box 1 of this chapter has been narrowed based on a high bleeding risk.[22] The individual perioperative strategy for triple therapy should be based on a multidisciplinary decision. Nonetheless, there is an unchanged need for a high level of evidence to strengthen our perioperative strategies related to triple therapy. The novel factor Xa inhibitors (so-called 'xabans') are discussed in the new guideline on perioperative management of patients undergoing noncardiac surgery.[21]

REFERENCES

- Shah PK. Inflammation and plaque vulnerability. Cardiovascular drugs and therapy / sponsored by the International Society of Cardiovascular Pharmacotherapy. 2009;23(1):31-40.
- [2] Boersma E, Kertai MD, Schouten O et al. Perioperative cardiovascular mortality in noncardiac surgery: validation of the Lee cardiac risk index. The American journal of medicine. 2005;118(10):1134-41.
- [3] Lee TH, Marcantonio ER, Mangione CM et al. Derivation and prospective validation of a simple index for prediction of cardiac risk of major noncardiac surgery. Circulation. 1999;100(10):1043-9.
- [4] O'Riordan JM, Margey RJ, Blake G et al. Antiplatelet agents in the perioperative period. Arch Surg. 2009;144(1):69-76;
- [5] Schunemann HJ, Cook D, Grimshaw J et al. Antithrombotic and thrombolytic therapy: from evidence to application: the Seventh ACCP Conference on Antithrombotic and Thrombolytic Therapy. Chest. 2004;126(3 Suppl):688S-96S.
- [6] Burger W, Chemnitius JM, Kneissl GD et al. Low-dose aspirin for secondary cardiovascular prevention - cardiovascular risks after its perioperative withdrawal versus bleeding risks with its continuation - review and meta-analysis. Journal of internal medicine. 2005;257(5):399-414.
- [7] Collet JP, Montalescot G. Premature withdrawal and alternative therapies to dual oral antiplatelet therapy. Eur Heart J Suppl. 2006;8(suppl G):G45-G52.
- [8] Robless P, Mikhailidis DP, Stansby G. Systematic review of antiplatelet therapy for the prevention of myocardial infarction, stroke or vascular death in patients with peripheral vascular disease. The British journal of surgery. 2001;88(6):787-800.
- [9] Wilson SH, Fasseas P, Orford JL et al. Clinical outcome of patients undergoing non-cardiac surgery in the two months following coronary stenting. Journal of the American College of Cardiology. 2003;42(2):234-40.
- [10] Thachil J, Gatt A, Martlew V. Management of surgical patients receiving anticoagulation and antiplatelet agents. The British journal of surgery. 2008;95(12):1437-48.
- [11] Chen KY, Rha SW, Li YJ et al. Triple versus dual antiplatelet therapy in patients with acute STsegment elevation myocardial infarction undergoing primary percutaneous coronary intervention. Circulation. 2009;119(25):3207-14.
- [12] Poldermans D, Bax JJ, Boersma E et al. Guidelines for pre-operative cardiac risk assessment and perioperative cardiac management in non-cardiac surgery: The Task Force for Preoperative Cardiac Risk Assessment and Perioperative Cardiac Management in Non-cardiac Surgery of the European Society of Cardiology (ESC) and endorsed by the European Society of Anaesthesiology (ESA). European heart journal. Published Online First: 27 August 2009. doi:10.1093/eurheartj/ehp337
- Schouten O, Bax JJ, Poldermans D. Assessment of cardiac risk before non-cardiac general surgery. Heart 2006;92:1866-1872.
- [14] Bangalore S, Wetterslev J, Pranesh S et al. Perioperative betablockers in patients having noncardiac surgery: a meta-analysis. Lancet 2008;372(9654):1962-76.
- [15] Dunkelgrun M, Boersma E, Schouten O et al. Bisoprolol and fluvastatin for the reduction of perioperative cardiac mortality and myocardial infarction in intermediate-risk patients undergoing noncardiovascular surgery: a randomized controlled trial (DECREASE-IV). Annals of surgery. 2009;249(6):921-6.
- [16] Landoni G, Turi S, Biondi-Zoccai G et al. Esmolol Reduces Perioperative Ischemia in Noncardiac Surgery: A Meta-analysis of Randomized Controlled Studies. Journal of cardiothoracic and vascular anesthesia. Published Online First: 1 October 2009. doi:10.1053/j.jvca.2009.07.008

- [17] Kapoor AS, Kanji H, Buckingham J et al. Strenght of evidence for perioperative use of statins to reduce cardiovascular risk: systemic review of controlled studies. BMJ 2006;333(7579):1149.
- [18] Schouten O, Boersma E, Hoeks SE et al. Fluvastatin and perioperative events in patients undergoing vascular surgery. The New England journal of medicine. 2009;361(10):980-9.
- [19] Flu WJ, Hoeks SE, van Kuijk JP et al. Treatment recommendations to prevent myocardial ischemia and infarction in patients undergoing vascular surgery. Current treatment options in cardiovascular medicine. 2009;11(1):33-44.
- [20] Dickstein K, Cohen-Solal A, Filippatos G et al. ESC Guidelines for the diagnosis and treatment of acute and chronic heart failure 2008: the Task Force for the Diagnosis and Treatment of Acute and Chronic Heart Failure 2008 of the European Society of Cardiology. Developed in collaboration with the Heart Failure Association of the ESC (HFA) and endorsed by the European Society of Intensive Care Medicine (ESICM). European heart journal. 2008;29(19):2388-442.
- [21] Fleisher LA, Fleischmann KE, Auerbach AD, et al. 2014 ACC/AHA Guideline on Perioperative Cardiovascular Evaluation and Management of Patients Undergoing Noncardiac Surgery: A Report of the American College of Cardiology/American Heart Association Task Force on Practice Guidelines. J Am Coll Cardiol. 2014 Jul 29. doi: 10.1016/j.jacc.2014.07.944. [Epub ahead of print]
- [22] Moser M, Olivier CB, Bode C. Triple antithrombotic therapy in cardiac patients: more questions than answers. Eur Heart J. 2014;35(4):216-23.

Blood thinners

Statins

Beta-blockers Vitamin D status New-onset arrhythmias Carotid stent characteristics Aortic stentgraft composition Laparoscopic sac fenestration after EVAR Need we intervene for type II endoleaks?

Published as: "Safety of fluvastatin in patients undergoing high-risk noncardiac surgery" Expert Opinion on Drug Safety 2010;9:793-800

Michiel T. Voûte Tamara A. Winkel Don Poldermans

ABSTRACT

Importance of the field

In patients undergoing vascular surgery there is a high incidence of adverse cardiac events, due to sudden coronary plaque rupture. The non-lipid lowering, or pleiotropic effects of statins can help reduce adverse cardiovascular events, associated with vascular surgery.

Areas covered in this review

The evidence for perioperative use of fluvastatin, as well as other statins, in high-risk surgery patients is summarized in this review. Data on pharmacokinetics and metabolism is presented, together with considerations on possible drug interactions in the perioperative period.

What the reader will gain

The reader will gain a comprehensive understanding of the existing safety and efficacy data for fluvastatin and other statins in the perioperative period. The practical considerations of perioperative fluvastatin therapy will be presented, including potential side-effects and management of the early non-oral phase immediately postoperative. Finally, advice on when to initiate therapy and safety recommendations are offered.

Take home message

In patients scheduled for high-risk vascular surgery, fluvastatin improves postoperative outcome, reducing the incidence of myocardial damage by almost 50% in the first 30 days following vascular surgery. In comparison with placebo, fluvastatin was not associated with a rise in liver enzymes or creatine kinase levels. To bridge the non-oral phase, an extended-release formula is recommended.

MANUSCRIPT

INTRODUCTION

Statins are renowned for their ability to reduce low-density lipoprotein (LDL) cholesterol in patients at with hypercholesterolemia.[1-3]Apart from their lipidmodifying properties, other treatment effects of statins, so called pleiotropic effects, have been investigated over the last decade, including anti-inflammatory effects and cardiac risk reduction. among others.[4-6] Fluvastatin (Lescol[®]. Novartis Pharmaceuticals, Switzerland) was first tested in humans in 1986 and was approved for clinical use in the USA since December 1993.[7-8] In this review, we summarize the pharmacology of fluvastatin, evaluate the efficacy and safety in patients undergoing high-risk surgery and comment on possible future developments.

PHARMACOLOGY

Pharmacodynamic properties

Fluvastatin sodium is a white to pale yellow, hygroscopic powder. It is soluble in water, ethanol and methanol. Fluvastatin, like all statins, is an inhibitor of 3-hydroxy-3-methyl-glutaryl-CoA (HMG-CoA) reductase, an enzyme that is rate-controlling in the mevalonate pathway. Mevalonate is a precursor of sterols, including cholesterol. Inhibition of this enzyme leads to a decrease in cholesterol levels in serum, as well as hepatic cells. This leads to an increase in receptors for low-density lipoprotein (LDL), which clear LDL from the bloodstream, adding to the lipid-lowering effect of fluvastatin.

Additionally, statins has been described to have non-lipid lowering effects, or pleiotropic effects. These effects contribute to the reduction of macrovascular risk that is seen in treated patients. Fluvastatin has been described to improve endothelial function, reducing the expression of adhesion molecules in the peripheral and coronary circulation.9 Also, by inhibiting the secretion of metallo-proteases by macrophages, fluvastatin may stabilize atherosclerotic lesions, reducing the risk of plaque rupture. Finally, effects on clotting, fybrinolysis and platelet aggregation can prevent extensive thrombus formation on fissured or ruptured plaques.[9]

Pharmacokinetic properties of fluvastatin

When administered orally, fluvastatin absorption takes place primarily in the small intestine. About 90% of an oral dose is absorbed. Peak concentrations in plasma are reached on average at 0.5 - 1.5 hour in healthy, fasting individuals. Administration with food reduces the rate but not the extent of the absorption. A 50% lower maximum concentration is however reached when taken with food, after twice as much time, when compared to administration 4 hours after a meal. Fluvastatin endures a substantial first-pass effect in the liver after absorption, with a wide variety in nonhepatic bioavailability of ~ 24% on average (range 9-50%) after a 10mg single dose. Liver metabolism of fluvastatin is saturable, so that the systemic bioavailability increases in a non-linear manner after single or multiple doses above 20 mg.

BOX 1. Drug summary.				
Drug name	Fluvastatin			
Phase	Launched			
Approved indication	Hypercholesterolemia, mixed dyslipidemia Heterozygous familial hypercholesterolemia in pediatric patients Secondary prevention of coronary events Atherosclerosis			
Pharmacology description	HMG-CoA reductase inhibitor*			
Route of administration	Oral			
Chemical structure	F N H ₃ C CH ₃ OH OH O- Na +			
	C ₂₄ H ₂₅ FNO ₄ • Na Mol. wt. 433.46			
Pivotal trial(s)	LIPS, LCAS†; ALERT, DECREASE-III§			
*HMG-CoA, 3-hydroxy-3-methylglutaryl-coenzyme A, † Key studies supporting approved indications, § Other large-scale key studies Acronyms: LIPS, Lescol Intervention Prevention Study; LCAS, Lipoprotein and Coronary Atherosclerosis Study; ALERT, Assessment of LEscol in Renal Transplantation; DECREASE, Dutch Echographic Cardiac Risk				

Subject to multiple larger doses, the systemic non-hepatic bioavailability can increase

Pharmacokinetic properties of fluvastatin extended release

The main development rationale for fluvastatin XL was to provide an extended release formulation that allows for once daily dosing of 80 mg of fluvastatin while maintaining low plasma levels. A hydrophilic cellulose matrix, containing the drug, was designed to swell when in contact with intestinal fluids, allowing the drug to diffuse out.[11] The time to peak plasma concentrations was 3-6 hours with fluvastatin XL, within the dose range of 80-320 mg. Steady state concentrations were observed within 7 days with regular daily doses.[13]

Metabolism and excretion

Evaluation Applying Stress Echo study group

to 45-65% in humans.[10-12]

Fluvastatin is rapidly metabolized by several cytochrome (CYP) P450 isoenzymes, mainly CYP2C9 (75% of metabolism), primarily via hydroxylation. This process lead to forming of three main metabolites, 5-hydroxy fluvastatin, 6-hydroxy fluvastatin and des-isopropyl-fluvastatin. Ex vivo, the 5- and 6-hydroxy metabolites have an inhibitory

effect on HMG-CoA (88% and 45% of unmetabolised fluvastatin, respectively), but in vivo there is no clinical relevance to these metabolites due to rapid elimination from the circulation. Fluvastatin is excreted primarily in bile, and leaves the body in feces. Over 90% of fluvastatin is found in feces as metabolites, and less than 2% present as unchanged drug. Only 5% of radiolabeled fluvastatin is recovered in urine.[10-12]

CLINICAL APPLICATION OF FLUVASTATIN IN HIGH-RISK SURGERY

Clinical efficacy in trials

In a report by Boersma et al. surgical procedures were categorized into low, intermediate and high perioperative cardiac risk.14 In later guidelines from the European Society of Cardiology (ESC), high-risk surgery was defined as aortic and peripheral vascular surgery, because there is a high probability that the atherosclerotic process is also present in the coronary arteries. Endovascular procedures are considered to be of intermediate cardiac risk, based on data from several randomized trials with open or endovascular aneurysm repair, as is carotid artery endarterectomy.[3]

Few reports have been published on the use of statins in the perioperative period to reduce the risk for a cardiac event after vascular surgery. Only one paper to date has prospectively investigated the effect of fluvastatin on perioperative cardiac events in high-risk surgery patients. The DECREASE-III trial was a large, placebo-controlled trial that assesses the effect of fluvastatin on the 30-day postoperative outcome in vascular surgery patients.[15] In the DECREASE-III trial, 497 patients (372 male) were enrolled at a median of 37 days prior to vascular surgery. Double-blinded randomization took place between fluvastatin (n=250) and placebo (n=247). No adverse cardiac outcome was reached before surgery. In the first 30 days postoperative, evidence of myocardial ischemia was seen in 27 patients (10.8%) in the fluvastatin group, and 47 patients (19.0%) in the placebo group (P=.016). The secondary endpoint of death from cardiac causes or non-fatal myocardial infarction was reached in 12 patients (4.8%) on fluvastatin and 25 patients (10.1%) on placebo (P=.039). The number needed to treat (NNT) to prevent one occurrence of myocardial ischemia in the first 30 days after vascular surgery was 12; the NNT to prevent one nonfatal MI was 36; and the NNT to prevent one cardiovascular death was 42.[15-16]

SAFETY EVALUATION

Side-effects of fluvastatin treatment

Fluvastatin is generally well tolerated in treated patients. Combining all clinical trials, Lawrence et al. found only 1% of patients to withdraw from fluvastatin therapy due to adverse events, attributed to the study medication.11 Most common are complaints of headache, abdominal symptoms and muscle problems. Statin therapy in general has been reported to lead to myalgia, myositis and myopathy. Although often transient, hepatic abnormalities are also well-known side-effects of statin therapy. Liver function tests should be taken previous to statin administration and regularly thereafter. If a patient has active liver disease or persistent abnormal liver function tests, statin

prescription for continous lipid-lowering purposes is contraindicated.[11] No advice is available on statins as prophylaxis for cardiac events in the perioperative period, in patients with liver disease.

Side-effects of fluvastatin in comparison with other statins seem mild. In a pooled analysis by Novartis Pharmaceuticals in 2002, the proportion of the 8951 patients on fluvastatin (20 or 40 mg) or fluvastatin XL (80 mg) that had a CK level more than five times the upper limit of normal was comparable to placebo-treated patients.[17] A large observational study by Bruckert et al. reported on muscular symptoms in general practice with high-dose statin therapy.[18] The rate of muscular symptoms was 5.1% with fluvastatin, compared to 10.9-18.2 % for other statins. However, all these side-effects were recorded when fluvastatin was prescribed in general practice, not surgery patients.

In the high-risk surgery patients from DECREASE-III, general anesthesia prevented monitoring for muscular complaints, but creatine kinase (CK) levels were measured as part of the safety endpoints. In patients treated with fluvastatin, the median CK level was 141 U/L, compared to 113 U/L for placebo (P=.24), and the proportion of patients that had CK levels > 10 times the upper limit of normal (ULN) was 4.0% in the fluvastatin group versus 3.2% in the placebo group (P=.81). Hepatic toxicity was observed without significant differences between groups. The median levels of alanine aminotransferase (ALT) were 24 U/L for the fluvastatin group and 23 U/L for placebo group (P=.43), and only 3.2% of patients had ALT levels > 3 x ULN in the fluvastatin group, compared to 5.3% for placebo (P=.27).[15]

Safety in special populations

The safety of fluvastatin use in the general population has been subject to research on several occasions. There have been reports on a low incidence of adverse reactions in children and adolescents with familial heterozygous hypercholesterolemia.[19] Animal studies have reported an association of high-dose fluvastatin (12 and 24 mg/kg) with a higher rate of stillbirths and maternal mortality, while a low-dose (2 mg/kg) therapy had no effects on the dam nor the foetus.[20] No data with fluvastatin in pregnant women are available, but statin use should be discontinued in women who become pregnant, and in nursing mothers statin use is strongly discouraged.[21]

With regard to high-risk surgery patients, advanced age and renal insufficiency are common. Age was found not to be of influence on the plasma concentrations for the general population.[10,22] However, the treatment response measured by LDL-level changes was slightly higher in patients ≥65 years of age, than in patients aged under 65 (30% versus 27%). Subject to the large role of the liver in the metabolism of fluvastatin, renal dysfunction is of no large influence on fluvastatin pharmacokinetics.

Drug interactions

When prescribing statins, the possibility of drug interactions is important to consider. This may inhibit normal metabolism, leading to increased plasma levels, or increase normal metabolism, via enzyme induction, leading to a decreased treatment effect. Statin toxicity has been reported to cause myalgia or rhabdomyalysis and statin inhibition can lead to under-treatment of dislipidemia and subsequent increased cardiovascular risk.

As mentioned, the metabolism of fluvastatin takes place via the CYP P450 system. Where other statins are metabolized primarily by the CYP3A4 isoenzyme (atorvastatin, lovastatin and simvastatin) or not by the liver at all (pravastatin), fluvastatin is the only statin that is predominantly metabolized by the CYP2C9 isoenzyme (75%), and to a much lesser extent by CYP3A4 (~20%). There are several known interactions between statins and other drugs that are metabolised by, or are of influence on CYP3A4. Drugs that inhibit this isoenzyme are antigungals, erythromycin and other macrolides, histamine-2 blockers, cyclosporin, calcium channel blockers and grapefruit juice. All these agents lead to an increase in plasma concentration of statins, and the use of pravastatin or fluvastatin may be preferable, since these are not primarily metabolized by CYP3A4. Rifampicin, Phenobarbital, carbamazepine and phenytoin are examples of drugs that induce both CYP3A4 and CYP2C9, and therefore lead to increased metabolism of hepatically-metabolized statins.[23] The lipid-lowering effect of statins, including fluvastatin, can be reduced by concomitant use of these drugs.

Warfarin is metabolized by CYP3A4 and CYP2C9, and there have been few reports that patients on fluvastatin/warfarin are potentially at risk for bleeding complications. Careful monitoring of the INR is advised in patients on warfarin, following the start of or any change in statin use (except pravastatin).[24-25]

Interactions with anesthetic agents

Commonly used agents for general anesthesia may have interactions with statin therapy. Midazolam, a commonly used benzodiazepine, is metabolized in part by the same CYP P450 isoenzyme as most statins, CYP3A4. In theory this could lead to interactions and alterations on the efficacy of either drug or both. However, a recent report showed that statins had no influence on midazolam pharmacokinetics in healthy subjects, dismissing this theory. [26] Among hypnotic agents, propofol is a widespread example, generally used for narcotic induction. It is metabolized primarily by direct glucuronidation in the liver.[27] However, one report suggests that propofol can influence midazolam metabolism by inhibition of CYP3A4. In theory, an interaction between propofol and statins could also be expected based on this report. Finally, narcotic analgesics are used for general anesthesia patients. Fentanyl was introduced in the 1960s, and has been in use ever since. CYP3A4 is the major catalyst in fentanyl metabolism in humans.[28] An interaction with other drugs, metabolized by the same isoenzyme, is hypothesized, but not reported. Alfentanil, a derivate of fentanyl, was investigated in surgery patients with atorvastatin therapy.[29] There was no sign of altered pharmacokinetics of alfentanil.

Safety considerations of statin withdrawal

Caution in prescribing statin to surgery patients is warranted, in light of the mentioned safety evaluations, but statin withdrawal should be given equal consideration. In the first days directly following surgery, oral medication is often inhibited. Since the

pleiotropic effects of statins are often underappreciated, postoperative statin withdrawal is common.[16]

Unfortunately, statin withdrawal can cause a rebound effect, diminishing the treatment benefit of statin administration. Discontinuation of short-acting statins was associated with an increase in inflammatory markers and oxidative stress, and an increase in cardiac events has been observed following acute withdrawal of statins during periods of instability, compared to continuation of statin therapy.[30] In vascular surgery patients, an increased incidence of cardiovascular events was reported, related to statin withdrawal after surgery. In that report, fluvastatin XL was considered a usefull long-acting agent, with a lower rate of adverse cardiac events after withdrawal, than other agents.[31] In guidelines from the European Society of Cardiology the use of a long-acting statin is advised to prevent the withdrawal effect.[3]

USE OF OTHER STATINS IN HIGH-RISK SURGERY

Apart from DECREASE-III, there have only been four other reports on statin use in patients undergoing vascular surgery.[32-35] Two additional reports discuss the benefit of statin therapy during major noncardiac surgery, including both vascular and nonvascular, or in intermediate risk surgery, excluding vascular surgery entirely.[36-37] From the four reports on statin use during the perioperative period in vascular surgery patients, three were based on retrospective data acquisition, and only one was a prospectively randomized trial.

Poldermans et al. performed a retrospective case-controlled study in 2816 major vascular surgery patients.[32] Cases were all 160 (5.8%) patients who died during perioperative hospital stay, and for each case 2 controls were selected (n=320) from all survivors, stratified according to year and type of surgery. Information on perioperative statin use, cardiac risk factors and other medication was gathered for all subjects. There were significantly fewer statin users among those that died (8%) than among those that survived (25%, P<.001).[32]

Kertai et al. studied 570 patients who underwent abdominal aortic aneurysm (AAA) surgery.[34] Information on advanced age, medical history, results of dobutamine stress echography, and statin and beta-blocker therapy were collected. They found the incidence of the combined endpoint of perioperative mortality and MI to be significantly lower in statin users than nonusers (3.7% vs. 11.0%). Even after correcting for other covariates, statin therapy was beneficial to AAA-surgery patients (OR: 0.24, 95% confidence interval: 0.10-0.70, P=.01). Beta-blockers were also beneficial to the study population, and a combination of both drugs resulted in the highest relative reduction of the composite endpoint.[34]

The StaRRS study was a retrospective study in 1163 hospitalizations for noncardiac vascular surgery.[35] Patient characteristics, medical history and medication were collected, and perioperative complications, including death, MI, ischemia, congestive heart failure and ventricular tachyarrhythmias were scored.

Complications occurred in 9.9% of statin users, and in 16.5% in nonusers (P=.001). Even after correcting for other covariates, statins were associated with a reduction in risk of complications (OR=0.52, P=.001).[35]

Durazzo et al. conducted the first prospective, placebo-controlled, double-blind randomized trial on the effect of statins on cardiovascular event following vascular surgery.[33] They randomized one hundred patients for atorvastatin (n=50) or placebo, on average 30 days before vascular surgery. The combined primary endpoint consisted of death from cardiac cause, nonfatal myocardial infarction, unstable angina and stroke. During the 6-month follow-up period, the incidence of the primary endpoint was 26.0% in the placebo group, compared to 8.0% with atorvastatin. The event-free survival, as a function of time, was significantly higher in patients treated with atorvastatin (P=.018, according to Kaplan-Meier method).[33]

Comparison of the efficacy and safety of fluvastatin and other statins in high-risk surgery patients is inhibited by the lack of comparative studies, and the limited amount of prospectively randomized placebo-controlled trials. Only one prospective RCT was performed using fluvastatin, and one atorvastatin. The rate of cardiac death or non-fatal MI in DECREASE-III was 4.8%, whereas in the data from Durazzo et al. atorvastatin seemed to be associated with an incidence of 8.0% for the same endpoints.[15, 33] However, there were differences in patients characteristics, distribution of target vessels and years of surgery. More importantly, the incidence of cardiac events in the placebo group was substantially higher in the study by Durazzo et al. (20.0%), compared to DECREASE-III (10.1%).

USE OF FLUVASTATIN IN INTERMEDIATE-RISK SURGERY

Besides high-risk surgery, the treatment effect of fluvastatin has also been investigated in intermediate-risk surgery patients. DECREASE-IV was a multicenter RCT among 1066 patients undergoing elective nonvascular surgery.[37] In this trial, a trend was observed of reduced cardiac events with fluvastatin therapy in the perioperative period in nonvascular surgery patients. This was not statistically significant, possibly due to under enrollment of the study.[37] However, a recent follow-up study of DECREASE-IV showed that perioperative fluvastatin therapy had a so-called "legacy-effect" in these nonvascular surgery patients. Perioperative fluvastatin, discontinued 30 days after surgery, was associated with a reduction in the incidence of myocardial infarction at long-term follow-up of over thirty months. This suggests that the pleiotropic effects of statin therapy can help reduce the atherosclerotic burden at postoperative hospital discharge. Untreated patients are predisposed to a higher risk of cardiovascular events, compared to patients who did receive perioperative fluvastatin as cardioprotection.

CONCLUSIONS

The use of fluvastatin as lipid-lowering therapy in the general, non-surgical population has been reported in several studies, and was evaluated in a previous Expert Opinion by Lawrence and Reckless.[11] Fluvastatin was described to be effective at lowering total

and LDL cholesterol, and to be safe and well-tolerated. Furthermore, drug interactions are low in comparison with other hepatically metabolized statins.

The use of statins in patients undergoing high-risk surgery is aimed at reducing the incidence of perioperative cardiovascular adverse events. Of the few reports on this subject, fluvastatin was the agent of choice in the largest, prospectively randomized, placebo-controlled trial to date. Compared to placebo, fluvastatin is associated with a significant reduction of myocardial ischemia, myocardial infarction and cardiac death. In terms of safety, no significant adverse reaction was seen on CK-levels or liver enzymes levels, compared to placebo. Fluvastatin seems to be a safe and effective agent, suitable for perioperative treatment in patients undergoing vascular surgery.

EXPERT OPINION

Although currently no statin has a particular approved indication for perioperative cardiac prevention, statin are often considered in vascular surgery patients. Not only because of the high incidence of dyslipidemia in this population, but also because of the pleiotropic effects of statin therapy.

Fluvastatin is effective in reducing cardiovascular adverse events in the perioperative period in high-risk surgery patients. Considering the low proportion of patients with side-effects and the extended release formulation, no other available statin is preferential to fluvastatin XL. Unfortunately, no large RCTs have been performed with other statins to provide us with comparable data. For now, the DECREASE-III trial advocates fluvastatin XL administration prior to high-risk surgery. For adequate implementation of fluvastatin prescription in patients scheduled for surgery, some questions remain.

Is development of a non-oral form of administration warranted?

Surgery patients often have a limited intake immediately after surgery. To bridge these first days, a non-oral form of fluvastatin might be helpful. However, since the acting site of fluvastatin is primarily the liver, where it is metabolized, an intravenous solution may not be optimal. The passage of enteral fluvastatin through the liver is essential to its therapeutic effect. Suppositories may provide a solution, although uptake in the rectum is still not equal to the proximal intestinal tract.

Are there racial differences in the efficacy of fluvastatin?

Although no specific information on fluvastatin is available, racial differences have been reported in the efficacy of lipid-lowering therapy, with evidence that African-Americans are less likely to reach LDL-level goals with statin therapy.[38-39] However, for cardiac events in the perioperative period, no reports on race as a factor have been published. The racial differences in pleiotropic effects of perioperative fluvastatin use for the prevention of perioperative cardiac events are therefore unknown. Since the lipid profile plays an important role in atherosclerotic advancement, only large future studies may reveal that statin therapy is less effective in the prevention of perioperative cardiac events.
What is the best timing for perioperative statin therapy?

There are some studies of cell cultures and the influence of statins on inflammatory response markers. These show that only 4 hours after simultaneous administration with interleukin (IL)-6, a reduction of C-reactive protein (CRP) release in human hepatic cells was established by statins.[5] An in vitro study found that fluvastatin inhibited IL-6 expression in human vascular smooth muscle cells, even after only 4 hours.40 The effect substantiated after 8 hours, leading to nearly 50% reduction of IL-6 after 24 hours. In other words, in vitro it seems that fluvastatin has pleiotropic effects after 4-24 hours.

What can be expected for the future?

In DECREASE-III, fluvastatin has proven to reduce cardiovascular events immediately after vascular surgery.[15] The mentioned follow-up study from DECREASE-IV showed additional effects of perioperative fluvastatin at long-term follow-up in intermediate-risk surgery patients. Future studies may reveal a similar legacy effect of perioperative fluvastatin therapy in high-risk surgery patients.

REFERENCES

- [1] Gordon DJ, Rifkind BM. 3-Hydroxy-3-methylglutaryl coenzyme A (HMG-CoA) reductase inhibitors: a new class of cholesterol-lowering agents. Ann Intern Med. Nov 1987;107(5):759-761.
- [2] Pasternak RC, Smith SC, Jr., Bairey-Merz CN, et al. ACC/AHA/NHLBI Clinical Advisory on the Use and Safety of Statins. Circulation. Aug 20 2002;106(8):1024-1028.
- [3] Poldermans D, Bax JJ, Boersma E, et al. Guidelines for pre-operative cardiac risk assessment and perioperative cardiac management in non-cardiac surgery: the Task Force for Preoperative Cardiac Risk Assessment and Perioperative Cardiac Management in Non-cardiac Surgery of the European Society of Cardiology (ESC) and endorsed by the European Society of Anaesthesiology (ESA). Eur Heart J. Nov 2009;30(22):2769-2812.
- [4] Davignon J, Laaksonen R. Low-density lipoprotein-independent effects of statins. Curr Opin Lipidol. Dec 1999;10(6):543-559.
- [5] Arnaud C, Burger F, Steffens S, et al. Statins reduce interleukin-6-induced C-reactive protein in human hepatocytes: new evidence for direct antiinflammatory effects of statins. Arterioscler Thromb Vasc Biol. Jun 2005;25(6):1231-1236.
- [6] Van Kuijk JP, Flu WJ, Witteveen OP, Voute M, Bax JJ, Poldermans D. The influence of statins on the expansion rate and rupture risk of abdominal aortic aneurysms. J Cardiovasc Surg (Torino). Oct 2009;50(5):599-609.
- [7] Yuan JN, Tsai MY, Hegland J, Hunninghake DB. Effects of fluvastatin (XU 62-320), an HMG-CoA reductase inhibitor, on the distribution and composition of low density lipoprotein subspecies in humans. Atherosclerosis. Apr 1991;87(2-3):147-157.
- [8] Tse FL, Jaffe JM, Troendle A. Pharmacokinetics of fluvastatin after single and multiple doses in normal volunteers. J Clin Pharmacol. Jul 1992;32(7):630-638.
- [9] Corsini A. Fluvastatin: effects beyond cholesterol lowering. J Cardiovasc Pharmacol Ther. Jul 2000;5(3):161-175.
- [10] Scripture CD, Pieper JA. Clinical pharmacokinetics of fluvastatin. Clin Pharmacokinet. 2001;40(4):263-281.
- [11] Lawrence JM, Reckless JP. Fluvastatin. Expert Opin Pharmacother. Nov 2002;3(11):1631-1641.
- [12] From: http://www.pharma.us.novartis.com/product/pi/pdf/Lescol.pdf.
- [13] Sabia H, Prasad P, Smith HT, Stoltz RR, Rothenberg P. Safety, tolerability, and pharmacokinetics of an extended-release formulation of fluvastatin administered once daily to patients with primary hypercholesterolemia. J Cardiovasc Pharmacol. May 2001;37(5):502-511.
- [14] Boersma E, Kertai MD, Schouten O, et al. Perioperative cardiovascular mortality in noncardiac surgery: validation of the Lee cardiac risk index. Am J Med. Oct 2005;118(10):1134-1141.
- [15] Schouten O, Boersma E, Hoeks SE, et al. Fluvastatin and perioperative events in patients undergoing vascular surgery. N Engl J Med. Sep 3 2009;361(10):980-989.
- [16] Poldermans D. Statins and noncardiac surgery: current evidence and practical considerations. Cleve Clin J Med. Nov 2009;76 Suppl 4:S79-83.
- [17] Benghozi R, Bortolini M, Jia Y, Isaacsohn JL, Troendle AJ, Gonasun L. Frequency of creatine kinase elevation during treatment with fluvastatin. Am J Cardiol. Jan 15 2002;89(2):231-233.
- [18] Bruckert E, Hayem G, Dejager S, Yau C, Begaud B. Mild to moderate muscular symptoms with highdosage statin therapy in hyperlipidemic patients--the PRIMO study. Cardiovasc Drugs Ther. Dec 2005;19(6):403-414.
- [19] van der Graaf A, Nierman MC, Firth JC, Wolmarans KH, Marais AD, de Groot E. Efficacy and safety of fluvastatin in children and adolescents with heterozygous familial hypercholesterolaemia. Acta Paediatr. Nov 2006;95(11):1461-1466.
- [20] Hrab RV, Hartman HA, Cox RH, Jr. Prevention of fluvastatin-induced toxicity, mortality, and cardiac myopathy in pregnant rats by mevalonic acid supplementation. Teratology. Jul 1994;50(1):19-26.
- [21] Briggs GG FR, Yaffe SJ. Drugs in pregnancy and lactation. 5th ed. Baltimore, Maryland; 1998:630-632.
- [22] Lye M, Valacio R, Reckless JP, et al. Elderly patients with hypercholesterolaemia: a double-blind study of the efficacy, safety and tolerability of fluvastatin. Coron Artery Dis. 1998;9(9):583-590.
- [23] Chong PH, Seeger JD, Franklin C. Clinically relevant differences between the statins: implications for therapeutic selection. Am J Med. Oct 1 2001;111(5):390-400.

- [24] Kline SS, Harrell CC. Potential warfarin-fluvastatin interaction. Ann Pharmacother. Jun 1997;31(6):790.
- [25] Andrus MR. Oral anticoagulant drug interactions with statins: case report of fluvastatin and review of the literature. Pharmacotherapy. Feb 2004;24(2):285-290.
- [26] Kokudai M, Inui N, Takeuchi K, Sakaeda T, Kagawa Y, Watanabe H. Effects of statins on the pharmacokinetics of midazolam in healthy volunteers. J Clin Pharmacol. May 2009;49(5):568-573.
- [27] Yang LQ, Yu WF, Cao YF, Gong B, Chang Q, Yang GS. Potential inhibition of cytochrome P450 3A4 by propofol in human primary hepatocytes. World J Gastroenterol. Sep 2003;9(9):1959-1962.
- [28] Feierman DE, Lasker JM. Metabolism of fentanyl, a synthetic opioid analgesic, by human liver microsomes. Role of CYP3A4. Drug Metab Dispos. Sep 1996;24(9):932-939.
- [29] McDonnell CG, Malkan D, Van Pelt FD, Shorten GD. Elimination of alfentanil delivered by infusion is not altered by the chronic administration of atorvastatin. Eur J Anaesthesiol. Aug 2003;20(8):662-667.
- [30] Heeschen C, Hamm CW, Laufs U, et al. Withdrawal of statins increases event rates in patients with acute coronary syndromes. Circulation. Mar 26 2002;105(12):1446-1452.
- [31] Schouten O, Hoeks SE, Welten GM, et al. Effect of statin withdrawal on frequency of cardiac events after vascular surgery. Am J Cardiol. Jul 15 2007;100(2):316-320.
- [32] Poldermans D, Bax JJ, Kertai MD, et al. Statins are associated with a reduced incidence of perioperative mortality in patients undergoing major noncardiac vascular surgery. Circulation. Apr 15 2003;107(14):1848-1851.
- [33] Durazzo AE, Machado FS, Ikeoka DT, et al. Reduction in cardiovascular events after vascular surgery with atorvastatin: a randomized trial. J Vasc Surg. May 2004;39(5):967-975; discussion 975-966.
- [34] Kertai MD, Boersma E, Westerhout CM, et al. A combination of statins and beta-blockers is independently associated with a reduction in the incidence of perioperative mortality and nonfatal myocardial infarction in patients undergoing abdominal aortic aneurysm surgery. Eur J Vasc Endovasc Surg. Oct 2004;28(4):343-352.
- [35] O'Neil-Callahan K, Katsimaglis G, Tepper MR, et al. Statins decrease perioperative cardiac complications in patients undergoing noncardiac vascular surgery: the Statins for Risk Reduction in Surgery (StaRRS) study. J Am Coll Cardiol. Feb 1 2005;45(3):336-342.
- [36] Lindenauer PK, Pekow P, Wang K, Gutierrez B, Benjamin EM. Lipid-lowering therapy and inhospital mortality following major noncardiac surgery. JAMA. May 5 2004;291(17):2092-2099.
- [37] Dunkelgrun M, Boersma E, Schouten O, et al. Bisoprolol and fluvastatin for the reduction of perioperative cardiac mortality and myocardial infarction in intermediate-risk patients undergoing noncardiovascular surgery: a randomized controlled trial (DECREASE-IV). Ann Surg. Jun 2009;249(6):921-926.
- [38] Yood MU, McCarthy BD, Kempf J, et al. Racial differences in reaching target low-density lipoprotein goal among individuals treated with prescription statin therapy. Am Heart J. Oct 2006;152(4):777-784.
- [39] Krauss RM, Mangravite LM, Smith JD, et al. Variation in the 3-hydroxyl-3-methylglutaryl coenzyme a reductase gene is associated with racial differences in low-density lipoprotein cholesterol response to simvastatin treatment. Circulation. Mar 25 2008;117(12):1537-1544.
- [40] Ito T, Ikeda U, Shimpo M, et al. HMG-CoA reductase inhibitors reduce interleukin-6 synthesis in human vascular smooth muscle cells. Cardiovasc Drugs Ther. Mar 2002;16(2):121-126.

Blood thinners Statins

Beta-blockers

3 Vitamin D status **New-onset arrhythmias Carotid stent characteristics** Aortic stentgraft composition Laparoscopic sac fenestration after EVAR Need we intervene for type II endoleaks?

Published as:

"Safety of perioperative beta-blocker use: how do β -blockers compare in terms of side effects?"

Expert Opinion on Drug Safety 2011;10:545-558

Niels J.C. Ravensbergen Michiel T. Voûte Don Poldermans

ABSTRACT

Introduction

In the perioperative setting, there is still a high incidence of adverse cardiac events, due to sudden coronary plaque rupture or oxygen supply-demand imbalance. β -blockers play an important role in preventing these cardiac events. Discussion however remains on side-effects accompanying this therapy.

Areas covered

The evidence for perioperative use of β -blockers is summarized in this review, in terms of risk reduction, perioperative safety and current clinical use. Furthermore, data on pharmacokinetics, -dynamics and -genetics is presented.

Expert Opinion

In perioperative care, β -blockers are recommended and can be given safely when started early in a low dose, titrated to heart rate. In the future, there could be a place for added perioperative short-acting β -blockers to further optimize heart rate control.

MANUSCRIPT

INTRODUCTION

β-blockers are well established drugs for the treatment and prevention of heart failure (HF), cardiac arrhythmias, hypertension (HT) and coronary artery disease (CAD) [1-5]. In recent years β -blocker use in the perioperative setting has become common and has been advocated by both European and US guidelines [6,7]. Their implementation is based upon several RCT's demonstrating a positive effect on reducing cardiac death and non-fatal myocardial infarction (MI) [e.g. 8-15]. This is very recently underscribed by a retrospective study from Wallace et al., in which patients were divided into four groups, based on the pattern of perioperative β -blocker use (None, Addition, Withdrawal and Continuous). The Addition group and the Continuous group were both associated with a reduction in 30 day and 1 year mortality. In contrast withdrawal was associated with an increase in 30 day and 1 year mortality [16]. Perioperative cardiac events can be caused by multiple mechanisms. Due to the stress of surgery, a catecholamine surge takes place, which will increase heart rate and myocardial contractility. In patients with fixed coronary plaques, this can lead to an oxygen supplydemand imbalance, leading to infarction. In patients with unstable plaques the perioperative inflammatory response, as well as sheer stress caused by an increased heart rate, can cause rupture of these plagues, followed by thrombus formation and subsequent acute coronary thrombosis, leading to ischemia and infarction [17]. Perioperative ischemia and infarction have shown to be risk factors for increased cardiac mortality in the perioperative period as well as in long term follow-up. This has also been described for heart failure and arrhythmias [18,19]. Based on the pharmacological properties of β -blockers, they could play a prominent role in the treatment and prevention of these risk factors.

Discussion still remains on the safety of the perioperative use of β -blockers, especially regarding stroke, hypotension and bradycardia, particularly triggered by the results of the large POISE study [14]. In this review, after shortly summarizing the pharmacology of β -blockers, the efficacy and safety of β -blockers in patients undergoing non-cardiac surgery will be evaluated and a treatment recommendation based on current literature and our own experience will be provided.

PHARMACOLOGY

β-Adrenergic antagonists (β-blockers) block β-adrenergic receptors, of which 3 types have been differentiated. $β_1$ -receptors are predominantly located in cardiac tissue, in addition to cells of the juxtaglomerular apparatus (renin release). $β_2$ -receptors are also present in the heart but can mainly be found in the smooth muscle of the peripheral blood vessels and bronchi and are involved in metabolic effects such as lipolysis, glycogenolysis, gluconeogenesis, insulin release. Both receptors will lead to increased heart rate and contractility of the myocardium when they are stimulated [20]. $β_3$ adrenoceptor mRNAs have been detected in the ventricles and atria of the heart, in the

Box 1. Guidelines on perioperative patient management regarding β -blocker use.

ACCF/AHA guideline November 2009 [7]

Class I

 β -blockers should be continued in patients undergoing surgery who are receiving β -blockers for treatment of conditions with ACCF/AHA Class I guideline indications for the drugs. (Level of Evidence: C)

Class IIa

 β -blockers titrated to heart rate and blood pressure are probably recommended for patients undergoing vascular surgery who are at high cardiac risk owing to coronary artery disease or the finding of cardiac ischemia on preoperative testing. (Level of Evidence:B)

 β -blockers titrated to heart rate and blood pressure are reasonable for patients in whom preoperative assessment for vascular surgery identifies high cardiac risk, as defined by the presence of more than 1 clinical risk factor* (Level of Evidence: C)

 β -blockers titrated to heart rate and blood pressure are reasonable for patients in whom preoperative assessment identifies coronary artery disease or high cardiac risk, as defined by the presence of more than 1 clinical risk factor*, who are undergoing intermediate-risk surgery. (Level of Evidence: B)

Class IIb

The usefulness of β -blockers is uncertain for patients who are undergoing either intermediate-risk procedures or vascular surgery in whom preoperative assessment identifies a single clinical risk factor in the absence of coronary artery disease* (Level of Evidence:C)

The usefulness of β -blockers is uncertain in patients undergoing vascular surgery with no clinical risk factors* who are not currently taking β -blockers. (Level of Evidence: B)

Class III

 β -blockers should not be given to patients undergoing surgery who have absolute contraindications to β -blockade. (Level of Evidence: C)

Routine administration of high-dose β -blockers in the absence of dose titration is not useful and may be harmful to patients not currently taking β -blockers who are undergoing noncardiac surgery. (Level of Evidence: B)

*Clinical risk factors include history of ischemic heart disease, history of compensated or prior heart failure, history of cerebrovascular disease, diabetes mellitus, and renal insufficiency (defined in the Revised Cardiac Risk Index as a preoperative serum creatinine of greater than 2 mg/dL). Data from [7]

ESC guideline August 2009 [6]

Class I

$$\begin{split} \beta \text{-blockers are recommended in patients who have known} \\ \text{IHD or myocardial ischemia according to pre-operative} \\ \text{stress testing* (Level of Evidence: B)} \\ \beta \text{-blockers are recommended in patients scheduled for} \\ \text{high-risk surgery* (Level of Evidence: B)} \\ \text{Continuation of } \beta \text{-blockers is recommended in patients} \\ \text{previously treated with } \beta \text{-blockers because of IHD,} \\ \text{arrhythmias, or hypertension (Level of Evidence: C)} \\ \\ \textit{Class IIa} \end{split}$$

 β -blockers should be considered for patients scheduled for intermediate-risk surgery* (Level of Evidence: B) Continuation in patients previously treated with β -blockers because of chronic heart failure with systolic dysfunction should be considered (Level of Evidence: C)

Class IIb

 β -blockers may be considered in patients scheduled for low-risk surgery with risk factor(s) (Level of Evidence: B)

Class III

Perioperative high-dose β-blockers without titration are not recommended (Level of Evidence: A) β-blockers are not recommended in patients scheduled for low-risk surgery without risk factors (Level of Evidence: B)

* Treatment should be initiated optimally between 30 days and at least 1 week before surgery. Target: heart rate 60–70 beats/min, systolic blood pressure >100 mmHg. IHD, ischemic heart disease. Data from [6] brain, stomach, small and large intestines, gall and urinary bladders, prostate and myometrium [21]. Stimulation of these receptors in the ventricles decrease cardiac contractility, as has been shown by giving isoprenaline in the presence of nadolol, a β_1 and β_2 -receptor antagonist [22]. This indicates that β_3 -receptors might act as a safety valve during intense adrenergic stimulation, as further demonstrated by their overexpression in hypertension and heart failure. [23]. Results concerning atrial β_3 -receptor stimulation are still conflicting [21].

Table 1. Receptor involvement and effect by organ system					
Organ	Receptor-subtype	Physiological effect			
Heart	β1 >> β2	Heart rate, contractility, conduction velocity and			
	β3	automacity 个			
		contractility↓			
Gastrointestinal tract	β1, β2	Smooth muscle tone \downarrow			
Bronchial tract	β2	Smooth muscle tone \downarrow (bronchodialtion)			
Uterus	β2	Smooth muscle tone \downarrow (relaxation)			
Blood vessel	β2	Smooth muscle tone \downarrow (vasodilation)			
	β3	vasodilation			
Kidney	β1	Renin release 个			
Fat tissue	β2 > β1 (2:1), (β3)	Lipolysis 个			
Pancreas (β-cells)	β2, (β3)	Insulin release 个			
Liver	β2	Glycogenolysis 个, Gluconeogenesis 个			
Skeletal muscle	β2	Glycogenolysis 个, K+ uptake 个, Tremor 个			
Thyroid gland	β2	T4 \rightarrow T3 \uparrow (conversion)			

β-blockers prevent catecholamines from binding to the β-adrenergic receptor, hereby blocking their positive chronotropic and inotropic actions, which gives a decrease in heart rate (HR) and myocardial contractility, resulting in a lower cardiac output [20]. The effect on HR is especially visible during periods of dominant sympathetic control, such as exercise and stress. Further aspects of β-blockers consist of anti-ischemic, antiarrhythmic and anti-renin/angiotension properties. They prolong coronary diastolic filling time, inhibit catecholamine induced cardiac necrosis [24] and suppress the induction of stress induced proinflammatory cytokines [25]. Some β-antagonists partially activate the β-adrenergic receptor. This phenomenon is known as Intrinsic Sympathetic Activity (ISA). It is thought that β-blockers with intrinsic activity have less influence on resting HR, therefore giving fewer bradycardias [26]. However, the clinical relevance has shown to be rather controversial. Because of this, β-blockers with ISA do not play a role any longer in clinical use [27].

 β -antagonists can be classified in multiple ways, by generation, selectivity, or by elimination. There are 3 generations of β -blockers. First generation agents are non-selective (eg propranolol). Second generation agents are β_1 -selective (atenolol, metoprolol, betaxolol, bisoprolol) and third generation β -blockers have additional properties, such as vasodilation (carvedilol, nebivolol). β_1 -selectivity depends on the β -

blocker concentration at the receptor and is therefore relative. β_2 -blocking will give rise to peripheral vasoconstriction, by unopposed α -stimulation and to bronchoconstriction. Elimination can be either by hepatic metabolism, renal excretion (unchanged substance) or both [20]. Those drugs that are eliminated via the hepatic route are mostly lipid soluble, have a low bioavailability due to a high first pass effect and require more frequent administration, due to a short half life. In contrast, drugs that undergo renal metabolism are water soluble, also have a low variable bioavailability but have longer half lives [26]. Like in all orally administered drugs, the effective dose is dependent on the extent of absorption, half life and bioavailability. Dosing interval depends mostly on half life and absorption through the gastrointestinal tract, especially in extended release agents.

The four most described β -blockers in perioperative care are 1) atenolol, 2) metoprolol, 3) bisoprolol and 4) esmolol. All four agents are β_1 -selective. However, the degree of β_1 selectivity is different between these agents [20]. While the β_1 -selectivity of bisoprolol extends beyond the therapeutic dose range, this is not the case with atenolol and metoprolol which may also block some β_2 -receptors. At a dose of 100mg or more, metoprolol will also have an effect on β_2 -receptors [26]. For bisoprolol this happens at 20mg, where the range of used perioperative dose is 2.5 – 10mg [26]. Schnabel et al. concluded that atenolol, metoprolol and bisoprolol also bind to the β_3 -receptors [28]. Taken this into account, bisoprolol remains the most selective agent. The effect of these β -blockers on β_3 -receptors is still unclear. It has been demonstrated that Nebivolol (a β_1 -selective agent) has an agonistic effect on β_3 -receptors [29]. For the β blockers used in perioperative care, this still has to be assessed.

Metoprolol comes in a short- and long-acting oral form (tartrate and succinate respectively) and is available for intravenous (iv) injection. The tartrate form has a half life of 3-4 hours. For succinate, this is a factor 2-4 longer. As a consequence, dosage should be adjusted accordingly. Bisoprolol comes in an oral form only and has a half life of 10-12 hours, which is comparable to metoprolol succinate. Atenolol is available for iv and oral administration, has a half life of 5-8 hours and should therefore be dosed twice daily. Esmolol can only be administered intravenously and has a very short half life of 9 minutes.

A different aspect of β -blockers that has been given more and more attention lies in the field of pharmacogenetics. The hepatic elimination of some β -blockers depends on the catalytic activity of the hepatic cytochrome P450 isozyme CYP2D6 (carvedilol, nebivolol and metoprolol) [30]. A poor metabolizer (PM) phenotype results if all inherited alleles are non-functional (or null- *0) alleles. It has been demonstrated that in patients with these phenotypes metoprolol concentrations are 3-10 times higher. This phenomenon is seen shortly after drug administration, as well as during chronic use [30-32]. Bisoprolol has a relatively constant β -adrenergic inhibition independent of CYP2D6 genotype. Because of this difference in CYP2D6 dependency, there may be a greater variety in β -blocker plasma concentrations in the trials using metoprolol, compared to those which use bisoprolol.

RISK REDUCTION

Among patients undergoing noncardiac surgery, complications of underlying (a)symptomatic coronary artery disease are the major cause of perioperative mortality [33]. Mangano described that a myocardial infarction or cardiac death was seen in 2.5% of all patients over the age of 40, undergoing noncardiac surgery. Within vascular surgery patients this was 6.2% [34]. He furthermore estimated that in 2001 a hundred million patients would undergo surgery around the world, of which one third would be over the age of 65 or have more than 2 cardiac risk factors. An estimated 10% of these patients would suffer a perioperative MI. Because of these findings there have been many trials investigating the effectiveness of perioperative β -blocker use to reduce the oxygen demand of-, and the stress on the myocardium, and thus prevent perioperative myocardial ischemia. Of the randomized controlled trials (RCTs) that we reviewed, 7 show a reduction in perioperative myocardial ischemia and/or MI after β -blocker use [8,9,11-15], and 4 do not [35-38]. These RCT's differ substantially in their methods, as described in Table 2.

In 1988 Stone et al. randomized 128 hypertensive patients (blood pressure 160/90 mmHg – 200/100 mmHg) to either a single small oral dose of a beta-blocker (labetolol 100mg, atenolol 50mg, or oxprenolol 20mg) or standard care [8]. 11 of the 39 untreated patients had intraoperative ischemia, where in the patients who were treated with a β -blocker only 2 out of 89 suffered ischemia (p<0.001). All episodes of ischemia were related to a preceding episode of tachycardia, but not to an increase in blood pressure.

In 1996 Mangano et al. performed a randomized, double-blind, placebo-controlled trial to compare the effect of atenolol with that of a placebo on overall survival and cardiovascular morbidity in patients with or at risk for coronary artery disease (CAD) who were undergoing noncardiac surgery [9]. Patients were considered at risk for CAD if they had at least two of the following cardiac risk factors: age ≥ 65 years, hypertension, current smoking, a serum cholesterol concentration ≥240 mg per deciliter (6.2 mmol per liter), and diabetes mellitus. 200 patients were randomly assigned to receive either atenolol or placebo. In the placebo group 12 patients had a cardiac event (defined as a combined variable of myocardial infarction, unstable angina or congestive heart failure requiring hospital admission and clinical diagnosis and treatment, myocardial revascularization [coronary-artery bypass graft surgery or percutaneous transluminal coronary angioplasty], and death), six months after surgery. Within the group of patients who were treated with atenolol, no events were observed (p<0.001). Holter registration was performed from the day of surgery to 7 days postoperatively. These recordings showed no difference in ischemia pre- and during surgery. In the postoperative period however, there was a 50% lower incidence of ischemia in the atenolol group in the first 48 hours post surgery (p=0.03). For the first 7 days this was 40% (p=0.008) [10].

Zaugg et al. randomly assigned 63 patients to 1 of 3 groups: group I, no atenolol; group II, pre- and postoperative atenolol; group III, intraoperative atenolol [11]. The study

period ranged from directly preoperative to 72 hours postoperative. 17 out of 59 (29%) patients had a troponin I (cTnI) release. Perioperative release of cTnI was detected in 8 of 19, 4 of 20, and 5 of 20 patients in groups I, II, and III, respectively. Peak levels of cTnI occurred intraoperatively in 6 patients (two from each group) and postoperatively in 11 patients (group I: 6; group II: 2; group III: 3). Postoperative cTnI release was significantly related to increased postoperative HR (68 \pm 14 vs. 78 \pm 15 bpm, p=0.002). The percentage of patients with a postoperative tachycardia was higher in patients in group I, compared to those in groups II and III. These findings suggest a protective effect of β -blockers on myocardial damage.

In 1999 Raby et al. included 26 patients with preoperative ischemia at 24 hour Holter monitoring, scheduled for aortic aneurysm repair, infrainguinal arterial bypass, or carotid endarterectomy [12]. They identified the lowest HR at which ischemia occurred (ischemic threshold) and set the target heart rate at a heart rate 20% below the ischemic threshold or an absolute minimum of 60 bpm. Patients were then randomized to receive either IV esmolol, starting at 100 μ g · kg⁻¹ · min⁻¹, or to placebo. The study period included the first 48 postoperative hours. During this period patients were monitored by a Holter recording. Patients receiving esmolol had 33% postoperative ischemia, whereas in the placebo group this was 73% (p<0.05). Of the 15 patients receiving esmolol, 9 patients had a mean HR below the ischemic threshold, and all these subjects were without postoperative ischemia. In both groups one cardiac event occurred. Both had a mean HR well above their ischemic threshold. Heart rate control was the only multivariate predictor of postoperative ischemia (p < 0.01). The authors concluded that postoperative ischemia depended more on a tight heart rate control, rather than on β -blocker therapy. Furthermore, they concluded that the type of β blocker was also less important than heart rate control.

Also in 1999, Poldermans et al. performed a randomized controlled trial (Dutch Echocardiographic Cardiac Risk Evaluation Applying Stress Echocardiography [DECREASE] -I) [13]. 112 high risk patients, selected by stress echocardiography, were randomized to receive bisoprolol and standard care (n=59) or standard care alone (n=53). In the standard care group 9 patients (17%) had a nonfatal MI within 30 days postoperative, as compared to none in the bisoprolol group.

The PeriOperative Ischemic Evaluation trial (POISE), which was published in 2008, reported a protective effect of perioperative metoprolol succinate with regard to myocardial infarctions 30 days after surgery (HR 0.73, 95%Cl 0.60 – 0.89; p=0.0017) [14]. In this study noncardiac surgery patients were randomly assigned to extended release metoprolol succinate or placebo. 152 of 4174 (3.6%) patients receiving metoprolol had a non-fatal myocardial infarction as compared to 215 of 4177 (5.1%) in the placebo group.

Dunkelgrun et al. showed in 2009 that for intermediate risk patients undergoing noncardiac and nonvascular surgery, bisoprolol was associated with a significant reduction in 30-day non-fatal MI [15]. They randomized a total of 1066 patients to 4 groups (bisoprolol, fluvastatine, both or neither). In the bisoprolol group (n=264) there was an incidence of 1.9% for perioperative MI. In the double control group (n=268) this was 6.7% (p = 0.01).

Study	Beta- blocker	Starting Dose	Continuous Dose	Treatment start	Treatment end	Minimum HR *	Minimum RR *	Cardiac risk groups†	Surgery type risk groups‡
Stone [8]	Labe- tolol Atenolol Ox- prenolol	L 100mg A 50mg O 20mg	NA	Single oral dose 2hrs before induction				Inter & High	Inter & High
Mangan o [9]	Atenolol	10mg iv	10mg iv or 1dd 100mg oral	30 min preop	7 days postop	< 55 bpm	SBP <100 mmHg	Inter & High	All
Zaugg [11]	Atenolol	5-10mg iv or 5mg/5 min	5-10mg iv or NA	30 min preop or intraop only	72 hrs postop or directly postop	< 55 bpm	SBP <100 mmHg	Inter & High	Inter
Raby [12]	Esmolol	100-300 microgr/k g/min	100-300 microgr/ kg/min	directly postop	48 hours postop			Inter & High	Inter & High
DECREAS E I [13]	Biso- prolol	5-10mg	5-10mg	1 week preop	30 days postop	< 50 bpm	SBP <100 mmHg	High	High
POISE [14]	Meto- prolol Succ.	100mg	200mg	2-4 hours preop	30 days postop	< 50 bpm	SBP <100 mmHg	Inter & High	Inter & High
DECREAS E IV [15]	Biso- prolol	2.5mg – 5mg	2.5mg – 5mg	1 month preop	1 month postop	< 50 bpm	SBP <100 mmHg	Inter	Low & Inter
POBBLE [33]	Meto- prolol Tart.	4mg iv	2dd 25- 50mg	5-10 min preop	7 days postop	< 50 bpm	SBP <100 mmHg	Inter	High
MaVS [34]	Meto- prolol Tart.	25- 100mg	15mg iv/6 hours or 2dd 25- 100mg	2 hours preop	5 days postop	< 50 bpm (awake), <45 bpm (asleep)	SBP <100 mmHg	All	High
BBSA [35]	Biso- prolol	5-10mg	5-10mg	3 hours preop	10 days postop	< 50 bpm	SBP <100 mmHg	Low & Inter (3,6% High)	Inter
DIPOM [36]	Meto- prolol Succ.	100mg	100mg	2 hours preop	8 days postop	< 55 bpm	SBP <100 mmHg	Inter & High	All

‡ Surgical risk group as described by Boersma et al. [6]

The 4 RCT's that do not report a beneficial effect of perioperative β -blocker therapy are the PeriOperative β -Blocker trial (POBBLE) [35] the Metoprolol after Vascular Surgery (MaVS) study [36], the Beta Blocker in Spinal Anaesthesia (BBSA) study [37] and the Dlabetes POstoperative Mortality and morbidity (DIPOM) trial [38].

The POBBLE trial showed that 18 (42%) of 43 patients in the placebo group and 16 (32%) of 50 patients in the metoprolol group, had significant myocardial ischemia in the first month after surgery (P = 0.36) [35]. In the MaVS trial 496 patients were randomized to receive metoprolol or placebo [36]. 19 out of 246 patients in the metoprolol group had a perioperative MI within 30 days after surgery, as compared to 21 out of 250 patients in the control group (p=0.87).

Similar findings were reported from the BBSA study. This trial randomized between bisoprolol and placebo [37]. In the treatment group the incidence of perioperative MI was 1.8% (2 out of 110). In the placebo group the incidence was 0.9% (1 out of 109) (p=0.90).

Juul et al. reported the results of 921 diabetic patients from the DIPOM trial [38]. Patients were randomized to either metoprolol or placebo. 30 days after surgery, 27 patients in the metoprolol group and 21 patients in the placebo group (6% and 5% respectively), reached the primary endpoint (a combined endpoint of all cause mortality, acute MI, unstable angina, or congestive heart failure discovered or aggravated during admission to the hospital). 6 months after surgery this was 21% in the metoprolol and 20% in the placebo group (p=0.66).

The limitations of these trials might partially explain why these differed in their findings as compared to the previously mentioned studies. POBBLE included 103 patients over a period of nearly 3 years and was discontinued because of poor recruitment and lack of funding. The BBSA trial was underpowered and included patients had varying cardiac risk profiles. Furthermore, the authors from the DIPOM trial concluded that they included only patients with diabetes and might not have included enough patients. However, an absolute risk reduction of 7% with a power of 80% was possible to detect. Bangalore et al. performed a meta-analysis [39] in which they included 12306 patients from 33 randomized trials. They found that β -blocker therapy was associated with a 35% decreased risk of non-fatal myocardial infarction (numbers needed to treat (NNT) 63) and a 64% decreased risk of myocardial ischemia (NNT 16) at 30 days postoperative. Trials were classified on bias-risk. Trials were analyzed for sequence generation of allocation, allocation concealment and masking of participants, personnel, and outcome assessors. When a trial was considered to have a low quality on any of these 3 components, it was classified as high or unclear risk for bias. When analyzing only the low-risk bias trials, there still was a 28% decreased risk of non-fatal MI (NNT 80) and a 59% decreased risk of myocardial ischemia (NNT 23). Figure 1 describes the OR's for a composite endpoint of ischemia, non-fatal MI and cardiac death for the previous mentioned studies.

Decreasing myocardial oxygen consumption is one way of preventing perioperative cardiac death by using β -blockers. Additional cardioprotective factors are redistribution of coronary blood flow to the subendocardium and increasing the threshold for ventricular fibrillation. Analyzing the previously mentioned RCT's for perioperative death, either all cause or cardiac origin, the results are mainly in line with what was found for ischemia. Except in the POISE trial, where an increased total mortality was found in the treatment group (HR 1.33, 95%CI 1.03 – 1.74; p=0.0317) [14]. The amount of cardiovascular death did not differ significantly (HR 1.30, 95%CI 0.92 – 1.83; p=0.14). Only sepsis or infection was significantly different between groups as a cause of death. Both were more common among the patients allocated to metoprolol. Clinically significant hypotension had the largest population attributable risk (PAR) for death (37.3%, 29 – 45) and had an adjusted odds ratio (OR) of 4.97 (95% CI 3.62 – 6.81). Furthermore, suffering an intra- or postoperative stroke had an OR of 18.97 (95% CI

9.93 – 36.3). Patients in the metoprolol group more often suffered a stroke than did patients in the placebo group. This was also related to clinically significant hypotension ([OR 2. 14, 95%CI 1.15 – 3.96] [PAR 14.7%, 95%CI 5.2 – 35.4]).



Mangano et al. reported a significant reduction in all cause mortality during 2 year follow-up [9]. In the placebo group 21 patients out of a 101 died (20%), of which 12 were of cardiac causes. For the atenolol group this was 9 out of 99 (9%) (4 of cardiac causes). Thus, overall mortality was 55 percent lower in the atenolol group (p=0.019) and mortality from cardiac causes was 65 percent lower (p=0.033). The first six to eight months were mostly responsible for the effect of atenolol therapy on cardiac outcomes (1 death from noncardiac causes in the atenolol group vs. 10 in the placebo group, 7 of which were from cardiac causes; p=0.001). It has to be noted that in a multivariate analysis, diabetes was the only statistical significant predictor of death. Furthermore, results might have been influenced by preoperative β -blocker withdrawal in the control group. However the 2 year results did not indicate this. There were 12 cardiac deaths in the 2 years after surgery in the control group. From the 8 patients using β -blockers preoperatively, 1 patient died (12%), as compared to 11 deaths out of 91 patients not using β -blockers (12%).

In the DECREASE-I trial 9 patients in the standard-care group (17 percent) died of cardiac causes during the perioperative period, as compared to two patients (3.4 percent) in the bisoprolol group (p=0.02) [13]. This resulted in 34% of patients receiving standard care alone and 3.4% of patients receiving standard care and bisoprolol reaching the combined endpoint of cardiac death and nonfatal MI (p<0.001).

Dunkelgrun et al. showed a similar result in the DECREASE IV study [15]. 5 patients died in the control group, where no patients from the bisoprolol group died. In the study of Raby et al. no patients died [12]. In the 4 trials that did not show a difference with regard to ischemia (POBBLE, MaVS, BBSA and DIPOM) [35-38], there also was no difference in mortality. Bangalore reported a 28% increased risk of all-cause mortality (number needed to harm (NNH) 164), while there was no difference in cardiovascular mortality between the treatment and the placebo group. This result was mainly driven by the results from the POISE trial [39].

SAFETY EVALUATION

Historically β -blockers were thought to worsen symptoms of respiratory dysfunction, heart failure, impotence and intermittent claudication. In this section, these aspects and other side effects of β -blockade will be discussed, and summarized in Table 3.

Respiratory effects

Chronic obstructive pulmonary disease (COPD) has shown to be an independent cardiovascular risk factor. In 1998 Gottlieb et al. retrospectively investigated the records of 201.752 patients with MI [40]. 22.1% of COPD patients and 17.7% of asthma patients received a β -blocker. These patients had a 40% reduction in mortality compared to patients with pulmonary disease who did not use a β -blocker. Side effects however, were not reported.

Van Gestel et al. evaluated 3371 patients undergoing major vascular surgery, of which 1205 patients had a history of COPD. 462 (37%) received β -blocker therapy. β -Blocker use was associated independently with lower 30-day (OR 0.37; 95% CI 0.19–0.72) and long-term mortality in patients with COPD (HR 0.73; 95% CI 0.60–0.88) [41]. It was concluded that in carefully selected patients with COPD, the use of cardioselective β -blockers appears to be safe and associated with reduced mortality.

Salpeter et al. performed a meta-analysis to asses the effect of β -blockers on respiratory function in patients with reactive airway disease [42]. Nineteen single dose treatment studies (with a total of 204 patients) and 10 continued dose studies (with a total of 141 patients) were included. The β -blockers used were atenolol, metoprolol, bisoprolol and practolol. FEV-1 was reduced with 7.46% (CI 5.59 - 9.32%) after a single dose and was not reduced after continuous treatment (which ranged from 3 days to 4 weeks). For both single dose and continuous treatment, no increase in respiratory symptoms was seen. The authors concluded that, based on these trials, cardioselective β -blockers should not be withheld from patients with mild to moderate airway disease in conditions for which a clear benefit has been demonstrated. Salpeter et al. found that the same held true for patients with COPD [43]. No significant change in FEV-1 or patient symptoms was seen after β -blocker use, be it either single dose or continuous therapy, even in patients with severe COPD.

Table 3. Take home messages concerning side effects.					
Side effect	Take home message				
Respiratory	- β -blockers minimally reduce FEV-1, but don't increase respiratory symptoms in asthma and COPD patients [40-43].				
Bradycardia	- β -blockers commonly cause bradycardia. However, only a small part requires treatment [37,44].				
Hypotension and stroke	- In a high dose, given shortly preoperative, β -blockers significantly increase the incidence of hypotension and stroke. This effect disappears when administration is started early, in a low dose and titrated to heart rate [14,47,48].				
Heart failure	- (A)symptomatic heart failure is a predictor for worse perioperative cardiovascular outcome. β -blockers have become a cornerstone of heart failure therapy [18,49-52].				
Intermittent claudication	- β -blockers do not increase symptoms of intermittent claudication [53].				
Impotence	 Erectile dysfunction seems to be caused by the underlying disease and psychological factors [54-56]. 				
Withdrawal	- β -blocker withdrawal shortly preoperative has shown to cause a hypersensitivity reaction, and is related to worse perioperative cardiovascular outcome. Therefore, β -blockers should be continued in patients already using them [16,57-64].				

Bradycardia

Since β -blockers suppress sympathetic activity and thereby cause a decrease in heart rate, bradycardia is a common side-effect [44]. Bangalore et al. described a high risk of perioperative bradycardia (OR 3.13, 95% Cl 2.51–3.92, p<0.0001; l^2 =29.5; NNH 8), but the risk of bradycardia requiring treatment was markedly lower (NNH 22) [37].

Hypotension and Stroke

Controversy about perioperative β -blocker use is mainly based on a reported increase in the amount of strokes in the POISE trial [14]. It reported an incidence of stroke of 1.0% in the metoprolol group and of 0.5% in the placebo group (HR 2.17, 95%Cl 1.26 – 3.74, p=0.0053) and that clinically significant hypotension had the largest intraoperative or postoperative risk for stroke (adjusted OR 2.14, 95%Cl 1.15 – 3.96. Population attributable risk 14.7%, 95%Cl 5.2 – 35.4). This high incidence of stroke was not found in the other previously described RCT's. The difference might be explained by important differences in the treatment protocols, i.e. differences in the use of the BB, such as different dosage, titration of the dose according to HR or not, route of administration and time of onset and duration of β -blocker therapy (Table 3). In POISE, the following treatment protocol was used: 100mg metoprolol succinate was given two to 4 hours preoperatively and 0 to 6 hours postoperatively. 12 hours after the first postoperative dose, patients received a dose of 200mg metroprolol succinate, if permitted by heart rate and blood pressure, which was then continued daily for 30 days post surgery. This protocol could have led to a maximum dose of 400mg in the first 24 hours, which is 100% of the maximum recommended daily dose as stated by the Food and Drug Administration's Center for Drug Evaluation and Research database (6,670 mg/kg/day for an average adult of 60 kg) [45].

At high metoprolol doses, additional β_2 -receptor blockade will ensue, due to the decreased β_1 -selectivity, which might cause cerebral ischemia by blocking β_2 -receptor mediated cerebral vasodilation, as described by Badget et al. [46]. As a comparison, for heart failure the recommended daily starting dose of metoprolol succinate is 12.5-25mg, usually increased at 2-4 weekly intervals [3] and for hypertension the starting dose is 25-50mg [1]. Van Lier et al. described the total incidence of postoperative stroke in the DECREASE I, II and IV trials, which was 0.46% (18 of 3,884) [47]. For patients on perioperative β -blocker therapy, the incidence was 0.5% (OR 1.16, 95% CI 0.4 to 3.4). They concluded that a low-dose bisoprolol regimen, started at least 30 days before surgery, had no association between β -blocker use and postoperative stroke. Figure 2 shows the OR's for perioperative stroke in the previously mentioned studies.



Risk Evaluation Applying Stress Echocardiography; DIPOM = Diabetes Postoperative Mortality and morbidity; MaVS = Metoprolol after Vascular Surgery; POBBLE = Perioperative Beta-BLockadE; POISE = PeriOperative Ischemic Evaluation trial. Also provided are the weighted means for all studies using bisoprolol and for all studies using metoprolol.

A case control study among 186,779 patients who underwent noncardiac surgery, carotid artery and intracerebral surgery excluded, showed an incidence of perioperative stroke of 0.02% [48]. All of the 34 cases of strokes were matched with 2 controls, which were stratified according to calendar year, type of surgery, and age. A similar use of β -blockers was found in both groups (29% vs. 29% p=1.0). In the subgroups of patients

who used cardiovascular therapy (blockers, statins, angiotensin-converting enzyme inhibitors, calcium channel blockers, angiotensin-II antagonists, nitrates, aspirin, dipyridamole, and clopidogrel) or had a presence of cardiac risk factors, comparable results were found. None of the cases had used β -blockers shorter than one month preoperatively. The percentage of the maximum recommended therapeutic dose was also similar in the cases and controls (median 25% vs. 25%, p=0.239).

These findings implicate that an early started, low dose long acting β -blocker regimen, titrated according to heart rate will have no increased risk for perioperative stroke, where a regimen started shortly preoperatively in a high dose, does have an increased risk.

Heart Failure

Flu et al. demonstrated that asymptomatic heart failure is predictive for 30-day and long-term cardiovascular outcome in open vascular surgery. Therefore, they advocated preoperative echocardiographic screening, next to inquiring for heart failure symptoms, as part of the standard preoperative work-up [18]. A recently published review concludes that adequate treatment of CHF is of great importance in the reduction of perioperative morbidity and mortality. Furthermore, it states that bisoprolol reduces all-cause mortality in patients with CHF undergoing major noncardiac surgery [49]. It was once thought that β -blockers would worsen outcome in patients with heart failure, because of the negative inotropic effect, which could cause decompensation in these patients. However, multiple studies have shown that in this population β -blockers give a reduction of all-cause mortality and significantly reduce HF hospitalizations and worsening of HF [50]. The benefit of β -blockers on mortality in heart failure patients is for ischemic as well as for non-ischemic heart failure through all stages of the disease [50-52].

Worsening of Intermittent Claudication

It was thought β -blockade would worsen complaints of patients with peripheral arterial disease, because of a blockade of peripheral β_2 -receptors by non-selective or less beta₁-selective β -blockers thus inhibiting the vasodilation these receptors normally are mediating. Blocking of these peripheral β_2 -receptors leads to peripheral vasoconstriction, due to unopposed vasoconstriction mediated by α_1 -receptors and therefore less perfusion of the lower limbs. A recent Cochrane Review demonstrated that there is currently no evidence supporting this hypothesis. It reviewed 6 RCT's with a total of 119 patients and concluded that none of the trials showed a statistically significant worsening effect of beta blockers [53].

Impotence

Animal studies suggested that β -blockers, especially lipophilic, may increase the latency to initial erection and ejaculation, and may reduce the number of erectile reflexes [54]. Recently Cocco performed a randomized controlled trial in which hypertensive patients received Metoprolol Succinate, and were randomized into 3 groups [55]. Group 1 was informed that the drug was Metoprolol and that it might induce Erectile Dysfunction (ED). Group 2 was told that they received Metoprolol, but the possible ED was not

mentioned. Group 3 was told neither. After 60 days the incidence of ED was 32%, 13% and 8% in Group 1, 2 and 3 respectively (p<0.01). The second phase of the study was that Metoprolol was continued and patients would be randomized to Tadalafil or placebo to treat ED. The results were similar. Therefore, they concluded that ED was mostly psychological and not so much due to β -blockade.

ED is common in patients with vascular disease. Both diseases share the same risk factors (age, hypertension, hypercholesterolemia, diabetes, and smoking) [56]. Patients tend to blame their ED on drug use, but a large part of this condition seems to be caused by the underlying disease and psychological factors.

Withdrawal

Next to side effects, withdrawal of currently used medication shortly before surgery or in the immediate postoperative period, also influences the perioperative safety of β blocker use. It might contribute to adverse myocardial outcome, because of a rebound effect, resulting in increased heart rate, blood pressure and plasma noradrenalin concentrations [57-61]. Redelmeier et al. demonstrated that longer acting β -blockers (such as atenolol) have a better perioperative cardiac protective action than short acting drugs (metoprolol), probably as a result of acute withdrawal effects from missed doses of short-acting β -blockers [62]. In concordance with these findings, Shammash et al. showed that postoperative β -blocker withdrawal was associated with an increase in total mortality (50% after withdrawal vs. 1.5% with continuation, OR 65.0, p<0.001), cardiovascular mortality (29% after withdrawal vs. 0% with continuation, p=0.005) and postoperative myocardial infarction (OR 17.7, p=0.003) [63]. Furthermore, the study of Hoeks et al. also demonstrated an increase in 1-year mortality after β -blocker withdrawal compared with non-users (HR 2.7, 95%CI 1.2 to 5.9) [64]. This was recently underscribed by Wallace et al. as discussed previously [16].

CLINICAL APPLICATIONS

Current recommendations on the perioperative use of BBs from the ACCF/AHA [7] and from the ESC [6] are described in Box 1. Both advocate a target heart rate of 60-70 bpm or 60-80 bpm for the ESC and ACCF/AHA respectively. To achieve this, initiation of β -blocker therapy is recommended between 30 days and 1 week before surgery at a daily dose of 2.5mg of bisoprolol or 50mg of metoprolol succinate. Furthermore, it is recommended that the dose be adjusted before surgery to achieve the target heart rate. The ESC advocates a systolic blood pressure >100mmHg, where the ACCF/AHA only mentions "no hypertension". The optimal heart rate range remains the same throughout the entire perioperative period, using iv administration when oral administration is not possible. Post-operative tachycardia should result in the first instance in treating the underlying cause, for example hypovolaemia, pain, blood loss, or infection, rather than increasing the β -blocker dose [6,7]. This regimen will have the best chance of attaining an optimal perioperative heart rate with a low risk of perioperative complications.

There are some differences between both guidelines [65]. These are mainly on the number of risk-factors needed, before β -blockers are mandatory. The ACCF/AHA seems

to be somewhat more conservative. However, both guidelines recommend perioperative β -blocker use in high- and intermediate risk surgery.

Although both guidelines recommend the perioperative use of β -blockers, guideline adherence is far from complete. Sidiqui et al. described 336 cases of cholecystectomy, for which criteria for β -blocker use were met in 146 patients [66]. Of these patients 70% were not receiving a β -blocker and of that group, β -blocker therapy was only started in 8% preoperatively [66]. In accordance to these findings, Hoeks et al. showed that the use of recommended medication in 711 patients with peripheral arterial disease (PAD) was lower than expected based on the current guidelines [67]. At baseline 48% of patients were on β -blocker therapy. After 3 year follow-up this increased slightly to 54%. In patients with PAD and ischemic heart disease (IHD), β blocker use was higher (68% at baseline and 68% 3 years after surgery).

CONCLUSIONS

Historically there has been a reluctance to use β -blockers in a perioperative setting, especially in patients with peripheral arterial disease, because of the hypothetical side effects and worsening of outcome. In the last decade it has become apparent that β -blockers reduce the risk of perioperative myocardial ischemia, myocardial infarction and cardiac mortality. When started at least one week prior to surgery, given in a low dose and titrated to heart rate, the benefits of β -blocker therapy still remain and side effects can be kept to a minimum. Both US and European guidelines have therefore incorporated perioperative β -blocker use into their recommendations.

EXPERT OPINION

 β -blockers have earned their place in the perioperative treatment of intermediate and high risk patients. An optimal balance between the reduction of cardiac events and the frequency of major side effects still has to be found. Cucherat performed a metaregression of 17 randomized trials to asses the relationship between HR reduction and mortality after MI. He concluded that a drop of 10 bpm in HR reduces the relative risk for cardiac death by 30% [68]. Feringa et al. showed that a higher dosed β -blocker leads to a lower heart rate and thereby to a lower incidence of myocardial ischemia, Troponin T release and mortality [69]. On the other hand, the results from POISE and the ClOpidogrel and Metoprolol in Myocardial Infarction Trial (COMMIT) demonstrated that a high dose β -blocker regime can also lead to more adverse events such as bradycardia, hypotension, cardiogenic shock and stroke [14,70]. In COMMIT 45852 patients with an acute MI, existing shorter than 24 hours, were randomly assigned to metoprolol or placebo. Metoprolol was administered intravenously up to 15mg directly after inclusion. 15 min after these intravenous doses, a 50 mg metoprolol or placebo tablet was to be given, and repeated every 6 h during days 0-1. From day 2 onwards, a 200 mg controlled-release metoprolol or placebo tablet was to be given once daily for up to 4 weeks. They observed a decrease of 5 patients with a reinfarction and 5 patients having ventricular fibrillation. However they also observed a rise of 11 patients developing cardiogenic shock. This was mostly seen in hemodynamically unstable patients. Both this dosing scheme as the one used in POISE are markedly higher than the ones used in those trials where no increase in major adverse events was observed.

The perioperative period is known to be a period of fluctuating heart rate, due to stress and a catecholamine surge, but also due to other factors such as hypovolemia. These different causes of tachycardia need a different treatment approach. Where β -blocker therapy is well suited to treat the extra catecholamine release, it is not indicated in case of a hypovolemia. Therefore, it is preferred not to have a β -receptor blocking effect which is too strong during surgery.

As shown by Raby et al., Esmolol (a short-acting β -blocker with a half-life of 9 minutes) can be well titrated and thereby achieve a heart rate within the preferred range [12]. A combination of a timely started, low dose long-acting β_1 -selective β -blocker and a perioperative short-acting, easily titratable β -blocker might lead to a better heart rate control, without the extra risk of adverse events as seen in a high dose long-acting β -blocker regimen. With such a regimen, treating physicians will be better able to diagnose and treat the different causes of tachycardia, wile the chance of hypotension and bradycardia is reduced. A perioperative target heart rate of 60-70 bpm is advocated. To achieve this, initiation of β -blocker therapy is recommended between 30 days and 1 week before surgery. The starting dose should be low (2.5 mg bisoprolol or 50mg metoprolol succinate) and titrated to heart rate. Especially shortly postoperative, there is a rise in heart rate as shown by Raby et al. [12]. Using a short-acting β -blocker in this period, titrated to heart rate, might improve outcome with respect to cardiac events.

REFERENCES

- [1] Mancia G, De Backer G, Dominiczak A, et al. 2007 Guidelines for the management of arterial hypertension. The Task Force for the Management of Arterial Hypertension of the European Society of Hypertension (ESH) and of the European Society of Cardiology (ESC). Eur Heart J 2007;28:1462-1536
- [2] Camm AJ, Kirchhof P, Lip GY, et al. Guidelines for the management of atrial fibrillation: the Task Force for the Management of Atrial Fibrillation of the European Society of Cardiology (ESC). Eur Heart J 2010;31(19):2369-429
- [3] Dickstein K, Cohen-Solal A, Fillippatos G, et al. ESC Guidelines for the diagnosis and treatment of acute and chronic heart failure 2008. The Task Force for the Diagnosis and Treatment of Acute and Chronic Heart Failure 2008 of the European Society of Cardiology. Developed in collaboration with the Heart Failure Association of the ESC (HFA) and endorsed by the European Society of Intensive Care Medicine (ESICM). Eur Heart J 2008;29(19):2388-2442
- [4] Kushner FG, Hand M, Smith SC Jr, et al. 2009 focused updates: ACC/AHA guidelines for the management of patients with ST-elevation myocardial infarction (updating the 2004 guideline and 2007 focused update) and ACC/AHA/SCAI guidelines on percutaneous coronary intervention (updating the 2005 guideline and 2007 focused update) a report of the American College of Cardiology Foundation/American Heart Association Task Force on Practice Guidelines. J Am Coll Cardiol 2009;54(23):2205-41
- [5] Fox K, Garcia MA, Ardissino D, et al. Guidelines on the management of stable angina pectoris: executive summary: The Task Force on the Management of Stable Angina Pectoris of the European Society of Cardiololy. Eur Heart J 2006;27(11):1341-81
- [6] Poldermans D, Bax JJ, Boersma E, et al. Guidelines for pre-operative cardiac risk assessment and perioperative cardiac management in non-cardiac surgery. The Task Force for Preoperative Cardiac Risk Assessment and Perioperative Cardiac Management in Non-cardiac Surgery of the European Society of Cardiology (ESC) and endorsed by the European Society of Anaesthesiology (ESA). Eur Heart J 2009;22:2769–2812
- [7] Fleisher LA, Beckman JA, Brown KA, et al. 2009 ACCF/AHA Focused Update on Perioperative Beta Blockade Incorporated Into the ACC/AHA 2007 Guidelines on Perioperative Cardiovascular Evaluation and Care for Noncardiac Surgery. A Report of the American College of Cardiology Foundation/American Heart Association Task Force on Practice Guidelines. Circulation 2009;120:e169-e276
- [8] Stone JG, Foëx P, Sear JW, et al. Myocardial ischemia in untreated hypertensive patients: effects of a single small dose of a beta-adrenergic blocking agent. Anesthesiology 1988;68(4):495-500
- [9] Mangano DT, Layug EL, Wallace A, Tateo I. For The Multicenter Study of Perioperative Ischemia Research Group. Effect of Atenolol on mortality and cardiovascular morbidity after noncardiac surgery. N Engl J Med 1996;335(23):1713-20
- [10] Wallace A, Layug B, Tateo I, et al. Prophylactic atenolol reduces postoperative myocardial ischemia. McSPI Research Group. Anesthesiology 1998;88(1):7-17
- [11] Zaugg M, Tagliente T, Lucchinetti E, et al. Beneficial effects from beta-adrenergic blockade in elderly patients undergoing noncardiac surgery. Anesthesiology 1999;91(6):1674-86
- [12] Raby KE, Brull SJ, Timimi F, et al. The effect of heart rate control on myocardial ischemia among high-risk patients after vascular surgery. Anesth Analg 1999;88(3):477-82
- [13] Poldermans D, Boersma E, Bax JJ, et al. Dutch Echocardiographic Cardiac Risk Evaluation Applying Stress Echocardiography Study Group. The effect of bisoprolol on perioperative mortality and myocardial infarction in high-risk patients undergoing vascular surgery. N Engl J Med 1999;341(24):1789-94
- [14] Devereaux PJ, Yang H, Yusuf S, et al. Effects of extended-release metoprolol succinate in patients undergoing non-cardiac surgery (POISE trial): a randomised controlled trial. Lancet 2008; 371(9627):1839-47
- [15] Dunkelgrun M, Boersma E, Schouten O, et al. Bisoprolol and fluvastatin for the reduction of perioperative cardiac mortality and myocardial infarction in intermediate-risk patients undergoing noncardiovascular surgery: a randomized controlled trial (DECREASE-IV). Ann Surg 2009;249(6):921-6

- [16] Wallace AW, Au S, Cason BA. Association of the pattern of use of perioperative β-blockade and postoperative mortality. Anesthesiology 2010;113(4):794-805
- [17] Thygesen K, Alpert JS, White HD. Universal definition of myocardial infarction. Joint ESC/ACCF/AHA/WHF Task Force for the Redefinition of Myocardial Infarction. J Am Coll Cardiol 2007;50(22):2173-95
- [18] Flu WJ, van Kuijk JP, Hoeks SE, et al. Prognostic implications of asymptomatic left ventricular dysfunction in patients undergoing vascular surgery. Anesthesiology 2010;112(6):1316-24
- [19] Winkel TA, Schouten O, Hoeks SE, et al. Risk factors and outcome of new-onset cardiac arrhythmias in vascular surgery patients. Am Heart J 2010;159(6):1108-15
- [20] Cruickshank JM. The modern role of beta-blockers in cardiovascular medicine. Shelton, Connecticut: People's medical publishing house USA, 2010
- [21] Ursino MG, Vasina V, Raschi E, et al. The β3-adrenoceptor as a therapeutic target: Current perspectives. Pharmacol Res 2009;59(4):221-34
- [22] Gauthier C, Tavernier G, Charpentier F, et al. Functional β3-adrenoceptor in the human heart. J Clin Invest 1996;98(2):556–62
- [23] Gauthier C, Sèze-Goismier C, Rozec B. Beta 3-adrenoceptors in the cardiovascular system. Clin Hemorheol Microcirc 2007;37(1-2):193-204
- [24] Cruickshank JM. Beta-blockers continue to surprise us. Eur Heart J 2000;21(5):354-64
- [25] Johnson JD, Campisi J, Sharkey SL, et al. Catecholamines mediate stress-induced increases in peripheral and central inflammatory cytokines. Neuroscience 2005;35:1295-1307
- [26] Reiter MJ. Cardiovascular drug class specificity: beta-blockers. Prog Cardiovasc Dis 2004;47(1):11-33
- [27] Cruickshank JM. Are we misunderstanding beta-blockers. Int J Cardiol 2007;120(1):10-27
- [28] Schnabel P, Maack C, Mies F, et al. Binding properties of beta-blockers at recombinant beta1-, beta2-, and beta3-adrenoceptors. J Cardiovasc Pharmacol 2000;36(4):466-71
- [29] Rozec B, Erfanian M, Laurent K, et al. Nebivolol, a vasodilating selective beta(1)-blocker, is a beta(3)-adrenoceptor agonist in the nonfailing transplanted human heart. J Am Coll Cardiol 2009;53:1532-8
- [30] Nozawa T, Taguchi M, Tahara K, et al. Influence of CYP2D6 genotype on metoprolol plasma concentration and beta-adrenergic inhibition during long-term treatment: a comparison with bisoprolol. J Cardiovascular Pharmacol 2005;46(5):713-20
- [31] Deroubaix X, Lins RL, Lens S, et al. Comparative bioavailability of a metoprolol controlled release formulation and a bisoprolol normal release tablet after single oral dose administration in healthy volunteers. Int J Clin Pharmacol Ther 1996;34(2):61-70
- [32] Rau T, Heide R, Bergmann K, et al. Effect of the CYP2D6 genotype on metoprolol metabolism persists during long-term treatment. Pharmacogenetics 2002;12(6):465-72
- [33] Mangano DT, Browner WS, Hollenberg M, et al. Association of perioperative myocardial ischemia with cardiac morbidity and mortality in men undergoing noncardiac surgery. The Study of Perioperative Ischemia Research Group. New Eng J Med 1990;323(26):1781-8
- [34] Mangano DT. Adverse outcomes after surgery in the year 2001—a continuing odyssey. Anesthesiology 1998;88(3):561-4
- [35] Brady AR, Gibbs JS, Greenhalgh RM, et al. Perioperative beta-blockade (POBBLE) for patients undergoing infrarenal vascular surgery: results of a randomized double-blind controlled trial. J Vasc Surg 2005;41(4):602-9
- [36] Yang H, Raymer K, Butler R, et al. The effects of perioperative beta-blockade: results of the Metoprolol after Vascular Surgery (MaVS) study, a randomized controlled trial. 2006;152(5):983-90
- [37] Zaugg M, Bestmann L, Wacker J, et al. Adrenergic receptor genotype but not perioperative bisoprolol therapy may determine cardiovascular outcome in at-risk patients undergoing surgery with spinal block: the Swiss Beta Blocker in Spinal Anesthesia (BBSA) study: a double-blinded, placebo-controlled, multicenter trial with 1-year follow-up. Anesthesiology 2007;107(1):33-44
- [38] Juul AB, Wetterslev J, Gluud C, et al. Effect of perioperative beta blockade in patients with diabetes undergoing major non-cardiac surgery: randomised placebo controlled, blinded multicentre trial. BMJ 2006;332(7556):1482-8
- [39] Bangalore S, Wetterslev J, Pranesh S, et al. Perioperative beta blockers in patients having noncardiac surgery: a meta-analysis. Lancet 2008;372(9654):1962-76

- [40] Gottlieb SS, McCarter RJ, Vogel RA. Effect of beta-blockade on mortality among high-risk and lowrisk patients after myocardial infarction. New Eng J Med 1998;339(8):489-97
- [41] Van Gestel YR, Hoeks SE, Sin DD, et al. Impact of cardioselective beta-blockers on mortality in patients with chronic obstructive pulmonary disease and atherosclerosis. Am J Respir Crit Care Med 2008;178(7):695-700
- [42] Salpeter SR, Orminston TM, Salpeter EE. Cardioselective beta-blockers in patients with reactive airway disease: a meta-analysis. Ann Intern Med 2002;137:715-25
- [43] Salpeter SR, Orminston TM, Salpeter EE. Cardioselective beta-blockers for chronic obstructive pulmonary disease: a meta-analysis. Respir Med 2003;97(10):1094-101
- [44] Ko DT, Hebert PR, Coffey CS, et al. Adverse effects of beta-blocker therapy for patients with heart failure: a quantitative overview of randomized trials. Arch Intern Med 2004;164(13):1389-94
- [45] U.S. Food and Drug Administration. Maximum Recommended Therapeutic Dose Database. Available from: http://www.fda.gov/AboutFDA/CentersOffices/CDER/ucm092199.htm
- [46] Badgett RG, Lawrence VA, Cohn SL. Variations in pharmacology of beta-blockers may contribute to heterogeneous results in trials of perioperative beta-blockade. Anesthesiology 2010;113(3):585-92
- [47] Van Lier F, Schouten O, Hoeks SE, et al. Impact of prophylactic beta-blocker therapy to prevent stroke after noncardiac surgery. Am J Cardiol 2010;105(1):43-7
- [48] Van Lier F, Schouten O, van Domburg RT, et al. Effect of chronic beta-blocker use on stroke after noncardiac surgery. 2009;104(3):429-33
- [49] Flu WJ, Winkel TA, Bax JJ, Poldermans D. Bisoprolol in patients with chronic Heart failure undergoing noncardiac surgery. Aging Health 2009;5(1):19-27
- [50] Bristow MR. Beta-adrenergic receptor blockade in chronic heart failure. Circulation.2000;101:558-569
- [51] Fauchier L, Pierre B, de Labriolle A, et al. Comparison of the beneficial effect of beta-blockers on mortality in patients with ischaemic or non-ischaemic systolic heart failure: A meta-analysis of randomised controlled trials. Eur J Heart Fail 2007;9(11):1136-9
- [52] Klapholz M. Beta-blocker use for the stages of heart failure. Mayo Clin Proc 2009;84(8):718-29
- [53] Paravastu SC, Mendonca DA, Da Silva A. Beta blockers for peripheral arterial disease. Eur J Vasc Endovasc Surg 2009;38(1):66-70
- [54] Srilatha B, Adaikan PG, Arlkumaran S, et al. Sexual dysfunction related to antihypertensive agents: results from the animal model. Int J Impot Res 1999;11(2):107-13
- [55] Cocco G. Erectile dysfunction after therapy with metoprolol: the Hawthorne effect. 2009;112(3):174-7
- [56] Erdmann E. Safety and tolerability of beta-blockers: predjudices. EHJ Supp 2009;11(Supp A):A21-5
- [57] Boudoulas H, Lewis RP, Kates RE, et al. Hypersensitivity to adrenergic stimulation after propranolol withdrawal in normal subjects. Ann Intern Med 1977;87:433-6.
- [58] Maling TJ, Dollery CT. Changes in blood pressure, heart rate, and plasma noradrenalin concentration after sudden withdrawal of propranolol. BMJ 1979;2:366-7.
- [59] Rangno RE, Langlois S, Lutterodt A. Metoprolol withdrawal phenomena: mechanism and prevention. Clin Pharmacol Ther 1982; 31:8-15.
- [60] Pontén J, Biber B, Bjurö T, et al. Beta-receptor blocker withdrawal. A preoperative problem in general surgery? Acta Anaesthesiol Scand Suppl 1982;76:32-7
- [61] Houston MC, Hodge R. Beta-adrenergic blocker withdrawal syndromes in hypertension and other cardiovascular diseases. Am Heart J 1988; 116:515-23.
- [62] Redelmeier D, Scales D, Kopp A. Beta blockers for elective surgery in elderly patients: population based, retrospective cohort study. BMJ 2005;331(7522):932-8
- [63] Shammash JB, Trost JC, Gold JM, et al. Perioperative beta-blocker withdrawal and mortality in vascular surgical patients. Am Heart J 2001;141(1):148-53
- [64] Hoeks SE, Scholte Op Reimer WJ, van Urk H, et al. Increase of 1-year mortality after perioperative beta-blocker withdrawal in endovascular and vascular surgery patients. Eur J Vasc Endovasc Surg 2007;33(1):13-9
- [65] Sears JW, Foex P. Recommendations on perioperative beta-blockers: differing guidelines: so what should the clinician do? Br J Anaesth 2010;104(3):273-5
- [66] Siddiqui AK, Ahmed S, Delbeau H, et al. Lack of physician concordance with guidelines on the perioperative use of beta-blockers. Arch Intern Med 2004;164(6):664-7

- [67] Hoeks SE, Scholte Op Reimer WJ, van Gestel YR, et al. Medication underuse during long-term follow-up in patients with peripheral arterial disease. Circ Cardiovasc Qual Outcomes 2009;2(4):338-43.
- [68] Cucherat M. Quantitative relationship between resting heart rate reduction and magnitude of clinical benefits in post-myocardial infarction: a meta-regression of randomized clinical trials. Eur Heart J 2007;28(24):3012-9
- [69] Feringa HH, Bax JJ, Boersma E, et al. High-dose beta-blockers and tight Heart rate control reduce myocardial ischemia and Troponin T release in vascular surgery patients. Circulation 2006;114(1 Suppl):1344-9
- [70] Chen ZM, Pan HC, Chen YP, et al. Early intravenous then oral Metoprolol in 45.852 patients with acute myocardial infarction: randomised placebo-controlled trial. Lancet 2005;366(9497):1622-32

Blood thinners Statins Beta-blockers

Vitamin D status

4

New-onset arrhythmias Carotid stent characteristics Aortic stentgraft composition Laparoscopic sac fenestration after EVAR Need we intervene for type II endoleaks?

Published as:

"Vitamin D deficiency may be an independent risk factor for arterial disease"

European Journal of Vascular and Endovascular Surgery 2012;44:301-306

Koen M. van de Luijtgaarden Michiel T. Voûte Sanne E. Hoeks Erik J. Bakker Michel Chonchol Robert Jan Stolker Ellen V. Rouwet Hence J.M.Verhagen

ABSTRACT

Objectives

The aim of this study was to assess the vitamin D status in patients with occlusive or aneurysmatic arterial disease in relation to clinical cardiovascular risk profiles and markers of atherosclerotic disease.

Methods

We included 490 patients with symptomatic peripheral arterial disease (PAD, n=254) or aortic aneurysm (n=236). Cardiovascular risk factors and comorbidities, carotid intimamedia thickness (CIMT), ankle-brachial index (ABI), serum high-sensitive C-reactive protein (hs-CRP), and vitamin D were assessed. Patients were categorized into severely (<25nmol/L) or moderately vitamin D deficient (26-50nmol/L), vitamin D insufficient (51-75nmol/L), or vitamin D sufficient (>75nmol/L).

Results

Overall, 45% of patients suffered from moderate or severe vitamin D deficiency. The prevalence of vitamin D deficiency was similar in patients with PAD and those with an aortic aneurysm. Low levels of vitamin D were associated with comorbid congestive heart failure and cerebrovascular disease. Adjusting for clinical cardiovascular risk factors, multivariable regression analyses showed that vitamin D deficiency was associated with high CIMT (P=0.001), low ABI (P<0.001) and elevated hs-CRP (P=0.025).

Conclusions

The current study shows a strong association between low vitamin D status and arterial disease, independent of traditional cardiovascular risk factors and irrespective of the type of vascular disease, i.e. occlusive or aneurysmatic disease.

MANUSCRIPT

INTRODUCTION

Several large epidemiological studies have concluded that vitamin D deficiency is associated with excess mortality.[1,2] It is becoming increasingly clear that vitamin D has a much broader range of actions in the human body in addition to its well-known effects on calcium homeostasis and bone metabolism. There is accumulating evidence that vitamin D deficiency has important extraskeletal effects, including the cardiovascular system.[3,4] Several clinical studies have reported a high prevalence of vitamin D deficiency in patients with peripheral arterial disease[5], coronary artery disease[6], and stroke[7], as well as the association of vitamin D deficiency with cardiovascular mortality.[2,8,9] Furthermore, low vitamin D status is related to major cardiovascular risk factors, such as hypertension, obesity, and diabetes mellitus.[4,10,11]

The aforementioned studies suggest that vitamin D deficiency promotes atherosclerosis.[4,12] However, it is not known whether this is a direct effect of vitamin D on the arterial wall, and/or the result of a vitamin D deficiency-associated increase in established cardiovascular risk factors. It is also unclear whether the severity of arterial disease is related to the severity of vitamin D deficiency. Furthermore, it is not known whether patients with aneurysmatic arterial disease also display vitamin D deficiency.

To answer these questions, we assessed the vitamin D status in a large population of patients with occlusive or aneurysmatic arterial disease, and related this to clinical cardiovascular risk profiles as well as to markers for the severity of arterial disease.

MATERIALS AND METHODS

Study population

The study population consisted of patients with peripheral arterial disease (PAD) or aortic aneurysmatic disease treated between 2004 and 2011 in the Erasmus University Medical Center in Rotterdam, the Netherlands. Patients with PAD were defined as having symptomatic atherosclerotic lower extremity arterial disease with an anklebrachial index (ABI) of \leq 0.9. Patients with aortic aneurysms were defined as having an aortic diameter >30 mm. Common carotid artery intima-media thickness (CIMT), anklebrachial index, and high-sensitive C-reactive protein (hs-CRP) were routinely measured in all vascular surgery patients. Patients with routinely measured serum vitamin D levels at the vascular outpatient clinic were included, whereas patients using vitamin D supplementation were excluded from this study. The study complies with the Declaration of Helsinki and was approved by the Institutional Review Board.

Baseline characteristics

A detailed history was obtained from every patient, including traditional risk factors; age, sex, hypertension (defined as a blood pressure \geq 140/90 mmHg in non-diabetics, \geq 130/80 mmHg in diabetics or use of antihypertensive medication),

hypercholesterolemia (defined as a low-density lipoprotein [LDL] cholesterol \geq 3.5 mmol/L or use of lipid lowering medication), chronic obstructive pulmonary disease (COPD; defined as a history of COPD or stage \geq 1 according to the GOLD classification), diabetes mellitus (defined as a fasting plasma glucose \geq 7.0 mmol/L, non-fasting glucose \geq 11.1 mmol/L or use of anti-diabetic medication) and smoking status. Furthermore, the atherosclerotic and cardiac risk factors as embedded in the Revised Cardiac Risk (RCR) index were obtained.¹³ The RCR index includes congestive heart failure (defined as a history of congestive heart failure), ischaemic heart disease (defined as a history of myocardial infarction, coronary revascularisation or the presence pathologic Q-waves on the electrocardiogram), cerebrovascular disease (defined as a history of ischaemic/haemorrhagic stroke or transient ischaemic attack), renal failure (defined as a serum creatinine \geq 2.0 mg/dL) and insulin dependent diabetes mellitus. The use of prescription medications was recorded and included statins, beta-blockers, reninangiotensin system (RAAS) inhibitors and diuretics.

Atherosclerotic markers

The severity of atherosclerotic disease was assessed by measurements of the CIMT, ABI and hs-CRP. The CIMT was measured using the guidelines from the 'Mannheim Carotid Intima-Media Thickness Consensus'.[14,15] Several measurements from the left and the right common carotid artery were made. The highest CIMT value was used for analysis, while measurements of plaques (defined as a focal structure encroaching into the arterial lumen of at least 0.5mm)[14] were excluded from analysis. The ABI was measured at rest using a portable counter-top Doppler 8-MHz vascular probe (Imexdop CT+ Vascular Doppler; Nicolet Vascular, Madison, WI, USA). The ABI was calculated by dividing the higher of the right and left systolic ankle pressures (posterior tibial or dorsal pedal artery) by the highest systolic brachial blood pressure according to the TASC guidelines.¹⁶ Serum hs-CRP was measured using immunochemistry (Beckman Coulter, Woerden, the Netherlands).

Vitamin D measurements

Serum vitamin D was measured in fresh blood samples using a 25-hydroxyvitamin D radioimmunoassay (Diasorin Inc, Stillwater, MN, USA). Within-run coefficient of variation (CV) was 8.6-12.5% and total imprecision CV was 8.2–11.0%. Patients were categorized into 4 groups based on commonly used cut-off values:¹⁷⁻¹⁹ severely (\leq 25 nmol/L) or moderately (26-50 nmol/L) vitamin D deficient, vitamin D insufficient (51-75 nmol/L), or vitamin D sufficient (>75 nmol/L). To convert nanomolar to nanogram per millilitre one should divide by 2.496.

Statistical analysis

Dichotomous data are described as counts and percentages. Continuous variables are described as mean±standard deviation (SD), or median and interquartile ranges [IQR] in case of non-Gaussian distribution. Categorical data were compared using chi-square tests. Continuous variables were compared using ANOVA, or using Kruskal-Wallis tests as appropriate. Linear univariable and multivariable regression analyses were performed in separate models using CIMT, ABI, or the natural logarithm of hs-CRP as dependent variable.

Table 1. Baseline characteristics according to vitamin D status						
			P for trend			
	Total population	Severely deficient	Moderately deficient	Insufficient	Sufficient	
		≤25 nmol/L	26-50 nmol/L	51-75 nmol/L	>75 nmol/L	
	n=490	n=62	n=160	n=138	n=130	
Vitamin D level (nmol/L), mean(±SD) Baseline characteristics	57±93	17±6	39±7	62±7	96±19	-
Male gender (%)	355(72.4)	42(67.7)	114(71.3)	111(80.4)	88(67.7)	0.083
Age (years±SD)	66.8±10.7	64.±11.6	66.9±11.2	67.8±9.6	66.7±10.6	0.212
Body mass index (kg/m2), mean(±SD)	26.4±4.4	26.1±5.3	26.4±4.6	26.8±4.2	26.0±4.0	0.495
eGFR (ml/min/1,73m2), mean(±SD)	78.32±26.29	86.07±30.75	75.18±28.74	78.46±23.48	78.35±22.97	0.053
Diagnosis						
Peripheral arterial disease (%)	254(51.8)	39(62.9)	81(50.6)	72(52.1)	62(47.6)	0.259
Thoracic and/or abdominal aneurysm (%)	236(48.2)	23(37.1)	79(49.4)	66(47.8)	68(52.3)	0.256
Cardiovascular diseases		. ,	. ,	. ,		
Congestive heart failure (%)	40(8.1)	12(19.3)	16(10.0)	6(4.3)	6(4.6)	0.001
Ischemic heart disease (%)	185(37.7)	27(43.5)	69(43.1)	50(36.2)	39(30.0)	0.112
Cerebrovascular disease (%)	85(17.3)	13(20.9)	35(21.8)	27(19.5)	10(7.6)	0.009
Cardiovascular risk factors						
Kidney disease (≥2.0mg/dl)	46(9.1)	4(6.4)	22(13.7)	11(7.9)	9(6.9)	0.103
Diabetes mellitus (%)	100(20.4)	20(32.2)	32(20.0)	25(18.1)	23(17.6)	0.103
Hypertension (%)	329(67.1)	41(66.1)	105(65.6)	103(74.6)	80(61.5)	0.152
Hypercholesterolemia (%)	455(92.8)	58(93.5)	152(95.0)	126(91.3)	119(91.5)	0.573
Smoking – current (%)	209(42.6)	36(58.0)	71(44.3)	62(44.9)	40(30.7)	0.014
Smoking – ever (%)	379(77.3)	54(87.0)	120(75.0)	110(79.7)	95(73.0)	0.129
Chronic obstructive pulmonary disease (%)	171(34.8)	22(35.4)	59(36.8)	43(31.1)	47(36.1)	0.691
Revised Cardiac Risk (RCR) index						
RCR score, mean(±SD)	1.16±1.01	1.45±1.14	1.27±1.10	1.11±0.84	0.93±0.97	0.004
0-1 risk factors (%)	333(67.9)	36(58.0)	100(62.5)	98(71.1)	99(76.1)	
2 risk factors (%)	105(21.4)	14(22.5)	38(23.7)	33(23.9)	20(15.3)	0.001
≥3 risk factors (%)	52(10.6)	12(19.3)	22(13.7)	7(5.0)	11(8.4)	
Medication						
Statins (%)	411(83.8)	54(87.0)	139(81.2)	112(81.1)	106(81.5)	0.548
Beta-blockers (%)	383(78.1)	50(80.6)	124(77.5)	110(79.7)	99(76.1)	0.903
Renin-angiotensin system inhibitors (%)	235(47.9)	32(51.6)	78(48.7)	71(51.4)	54(41.5)	0.384
Diuretics (%)	122(24.8)	14(22.5)	44(27.5)	37(26.8)	27(20.7)	0.536
Antiplatelets (%)	327(66.7)	49(79.0)	99(61.8)	92(66.6)	87(66.9)	0.124

25-hydroxyvitamin D per 10 nmol/L was used as independent variable and adjustments congestive were made for age. gender. heart failure. ischemic heart disease, cerebrovascular disease, renal function by estimated glomerular filtration rate (eGFR), diabetes mellitus, chronic obstructive pulmonary disease, hypertension, and smoking. To address the seasonal fluctuation of vitamin D levels, further adjustments were made for calendar season of vitamin D measurement. For all tests, a P-value < 0.05 (two-sided) was considered significant. All analyses were performed using PASW version 17.0 statistical software (SPSS Inc., Chicago, IL, USA).

RESULTS

A total of 490 patients were included in the study. 254 patients (51.8%) were diagnosed with peripheral arterial disease (PAD) of the lower extremities and 236 patients (48.2%) were diagnosed with a thoracic and/or abdominal aortic aneurysm. The mean age of the population was 67±11 years and the average value of vitamin D concentration was 57±93 nmol/L, as presented in Table 1. A total of 62 patients (12.7%) were severely vitamin D deficient, 160 patients (32.7%) were moderately deficient, 138 patients (28.2%) were vitamin D insufficient, and 130 patients (26.5%) had sufficient vitamin D levels. There were no differences between patients with PAD and those with an aortic aneurysm with regard to the frequencies of vitamin D deficiency (P=0.258, Figure 1), or the mean vitamin D

concentration (57±31 and 59.2±27 nmol/L, P=0.390). Mean ABI in the patients with aneurysmatic disease was 0.88 and 47% of these patients had an ABI \leq 0.9. No significant differences in vitamin D concentration were found between AAA patients with normal ABI or low ABI (mean 63 nmol/L vs. 55 nmol/L, P=0.066).





Also, although seasonal variation in vitamin D deficiency was observed in the overall population, no differences between patients with PAD and aneurysms were observed, as presented in Figure 2.

Cardiovascular risk factors

Patient groups with decreasing vitamin D levels had an increasing prevalence of congestive heart failure (P=0.001), cerebrovascular disease (P=0.009) and were more frequent current smokers (P=0.014), as presented in Table 1. Overall high risk cardiovascular profiles were significantly associated with lower vitamin D levels, as illustrated by a stepwise increase in RCR scores for groups with increasing vitamin D deficiency (P=0.004).

Atherosclerotic markers

The mean (±SD) CIMT in all patients was 0.97 ± 0.31 mm and a stepwise decrease was observed from 1.06 ± 0.37 mm in patients with severe vitamin D deficiency to 0.90 ± 0.27 in patients with sufficient vitamin D levels (P=0.007), as presented in Table 2. The mean ABI was 0.70 ± 0.26 and increased stepwise in each group from 0.56 ± 0.28 in patients with severe vitamin D deficiency to 0.77 ± 0.24 in patients with sufficient vitamin D levels (P<0.001). Furthermore, median hs-CRP in all groups was 4.3 mg/L [IQR: 2.2-7.8 mg/L]. High concentrations of hs-CRP were especially observed in patients with severe vitamin D deficiency with a median of 7.5 mg/L [2.5-12.7 mg/L] (P=0.040).

Table 2. Atherosclerotic markers according to vitamin D status							
		Vitamin D status					
Athero- sclerotic	Total population	Severely deficient	Moderately deficient	Insufficient	Sufficient	P for	
markers		≤25 nmol/L	26-50 nmol/L	51-75 nmol/L	>75 nmol/L	trend	
CIMT (mm)	0.97±0.31	1.06±0.37	1.01±0.34	0.94±0.27	0.90±0.27	0.007	
ABI	0.70±0.26	0.56±0.28	0.68±0.25	0.72±0.26	0.77±0.24	<0.001	
hs-CRP (mg/L)	4.3[2.2-7.8]	7.5[2.5-12.7]	4.0[2.3-7.9]	3.8[1.9-6.8]	4.8[2.2-7.8]	0.040	
CIMT and ABI are presented as mean±SD, hs-CRP as median and interquartile range.							
Abbreviations: CIMT; common carotid intima-media thickness, ABI; ankle-brachial index, hs-CRP; high-							
sensitive C-rea	ctive protein.						

Multivariable linear regression analyses were performed to determine the association between vitamin D concentration and CIMT, ABI and hs-CRP independently of clinical risk factors. Significant associations for vitamin D concentration per 10 nmol/L were observed for CIMT (beta -0.017 mm, 95%CI: -0.027:-0.007, P=0.001), ABI (beta 0.017, 95%CI: 0.008:0.026, P<0.001) and hs-CRP (beta -0.046 mg/L, 95%CI: -0.085:-0.006, P=0.025) (Table 3).

Table 3. Multivariable linear regression models for associations between vitamin D and atherosclerotic markers						
Markers	n		Beta for vitamin D [#]	95% CI for Beta	P-value	
СІМТ	420	Unadjusted	-0.019	-0.029 : -0.009	<0.001	
	459	Adjusted [*]	-0.017	-0.027 : -0.007	0.001	
ABI	265	Unadjusted	0.017	0.008 : 0.026	<0.001	
	303	Adjusted [*]	0.017	0.008 : 0.026	<0.001	
hs-CRP	201	Unadjusted	-0.044	-0.082 : -0.005	0.027	
	391	Adjusted [*]	-0.046	-0.085 : -0.006	0.025	
[*] adjusted for: age, gender, congestive heart failure, ischaemic heart disease, cerebrovascular disease, renal function using eGFR, diabetes mellitus, chronic obstructive pulmonary disease, hypertension,						

smoking and calendar season of 25-hydroxyvitamin D measurement.

[#]Vitamin D per 10 nmol/L.

DISCUSSION

The current study shows a strong association between low vitamin D status and the severity of arterial disease, independent of traditional cardiovascular risk factors and irrespective of the type of vascular disease, i.e. occlusive or aneurysmatic disease.

Vitamin D₃ is synthesized in the skin from cholesterol under the action of ultraviolet B light.[3] Furthermore, vitamin D can be ingested as cholecalciferol (vitamin D₃) or ergocalciferol (vitamin D₂). Vitamin D is subsequently converted to 25-hydroxyvitamin D (calcidiol) in the liver or stored in adipose tissue. In the kidneys, 25-hydroxyvitamin D is converted to 1,25-dihydroxyvitamin D (calcitriol), which is the biologically active form of vitamin D.[3] The blood concentration of 25-hydroxyvitamin D reflects the dietary intake of vitamin D₂ or D₃ and the amount of vitamin D₃ produced in the skin, and is considered the best indicator of vitamin D storage.[17] Since there is still some debate on the best classification of vitamin D status,[17-20] we used a currently proposed vitamin D classification including clinical relevant cut-off values to describe vitamin D status in our patient cohort.

The observed prevalence of vitamin D deficiency (i.e. ≤50 nmol/L) of 45% in patients with arterial disease is comparable to previous reports on vitamin D levels in patients with peripheral arterial disease.[21-24] Since vitamin D deficiency has been identified as an independent risk factor for mortality,[1,2] the question arises if and how vitamin D deficiency is related to the occurrence of cardiovascular events. In line with previous reports,[7,25] we found that vitamin D deficiency is associated with the occurrence of congestive heart failure and cerebrovascular disease in univariable analyses. In addition, as compared to patients with sufficient vitamin D levels, patients with severe vitamin D deficiency had a significantly higher RCR index, a well known predictor of postoperative cardiovascular events in patients undergoing non-cardiac surgery.[13]

Next, we attempted to identify how vitamin D deficiency is related to the severity of arterial disease. We observed a strong association between vitamin D deficiency and the atherosclerotic markers of CIMT and ABI. The CIMT and ABI provide information about the progression of atherosclerosis. In previous reports, Flu et al. showed the prognostic value of CIMT and ABI, independent of the RCR index.[26,27] Targher et al. observed a similar association between vitamin D deficiency and CIMT in patients with diabetes mellitus,[28] and Reis et al. reported a significant association between vitamin D deficiency and the internal, rather than the common, carotid intima-media thickness.[29] To our knowledge, only two other studies reported ABI measurements in patients with vitamin D deficiency.[5,30] Although both studies reported mild associations, our study clearly shows the stepwise decrease in ABI per vitamin D deficiency category, and a significant correlation in multivariable linear regression models. Additionally, whereas other studies reported varying results regarding CRP and vitamin D deficiency,[30-32] the current study shows that serum hs-CRP levels are elevated in patients with severe vitamin D deficiency.

In contrast to previous studies, we found that vitamin D deficiency was not related to the classic clinical risk factors for cardiovascular disease, including hypertension, obesity, diabetes, and dyslipidemia. Furthermore, the correlation between low vitamin D status and markers of atherosclerotic severity was independent of these cardiovascular risk factors.

Interestingly, a similar association between vitamin D deficiency and occlusive arterial disease was also observed in patients with aneurysmatic disease. To our knowledge, this relationship between vitamin D status and aneurysm formation has thus far not been reported in humans. Although aortic aneurysms have traditionally been attributed to atherosclerosis, there is increasing epidemiological, biochemical and genetic evidence that aneurysmal arterial disease is different from occlusive atherosclerotic disease, a common denominator being aging of the arterial wall.

Taken together, the data in the current study suggest that the relationship between vitamin D deficiency and arterial disease is mediated by an independent effect of vitamin D deficiency on the arterial wall. Vitamin D receptors are not exclusively detected in the bone and mineral pathway, but have a wide tissue distribution, including vascular smooth muscle cells and vascular endothelial cells.[17] The diverse physiologic actions of vitamin D on the vascular wall include reduction of smooth muscle cell proliferation,[33] reduction of macrophage secretion of pro-inflammatory cytokines IL-6 and TNF- α , and increased secretion of the anti-inflammatory cytokine IL-10, leading to a state of vascular inflammation.[34-36] In an atherosclerotic mouse model it has been demonstrated that oral vitamin D₃ reduces the formation of atherosclerotic plaques by the suppression of proatherogenic T lymphocytes.[37] In addition, low circulating levels of vitamin D have been associated with endothelial dysfunction in humans.[38,39] Furthermore, it has previously been reported that people with vitamin D deficiency have increased vascular calcification, a sign of advanced atherosclerosis,[40,41] as well as increased aortic stiffness.[42] These vitamin

D related effects all promote arterial disease.[4,12] Experimental studies provide increasing evidence that factors regulating mineral ion homeostasis, such as vitamin D, affect the aging process, including vascular aging.[43]

There are several limitations that need to be considered. Due to the nature of this study it remains uncertain whether the association between vitamin D deficiency and arterial disease is causal, or whether vitamin D deficiency is just a bystander. Furthermore, several potentially confounding factors could have influenced our analyses, the most important ones being race, diet, and sunlight exposure. Since our study population consisted mostly of Caucasians, race was not a factor in our analyses. Moreover, as lower vitamin D levels are observed in non-Caucasian populations, the true prevalence of vitamin D deficiency in PAD patients may actually have been underestimated. The influence of low dietary intake, thereby not only reducing vitamin D but also other nutrients, was not taken into account in this study. However, low vitamin D in the European population is mainly caused by low sunlight exposure rather than diet.[17,44] Therefore, in the multivariable models we corrected for the season of vitamin D measurement to minimize confounding by seasonal variations in sunlight exposure.

CONCLUSION

In conclusion, this study demonstrates that low vitamin D status is an indicator for the severity of arterial disease, independent of traditional cardiovascular risk factors and irrespective of the type of vascular disease, i.e. occlusive or aneurysmatic disease. It might be hypothesized that primary and secondary preventive strategies to reduce vascular disease should focus on vitamin D status, in addition to blood pressure reduction, lipid and glucose control, weight loss, and lifestyle changes. A beneficial effect of vitamin D supplementation on blood pressure reduction has been demonstrated in several clinical trials.[45,46] Although improving vitamin D status might be a promising public health strategy to reduce cardiovascular disease and improve survival,[47,48] there is still much debate about the requirement levels of vitamin D in relation to extra-skeletal outcomes.[20] Further large-scale, randomized clinical trials are needed to test the effects of vitamin D on cardiovascular disease and to further elucidate the biology of vitamin D on the arterial wall.
REFERENCES

- [1] Melamed ML, Michos ED, Post W, Astor B. 25-hydroxyvitamin D levels and the risk of mortality in the general population. *Arch Intern Med* 2008;168:1629-37.
- [2] Dobnig H, Pilz S, Scharnagl H, Renner W, Seelhorst U, Wellnitz B, et al. Independent association of low serum 25-hydroxyvitamin d and 1,25-dihydroxyvitamin d levels with all-cause and cardiovascular mortality. Arch Intern Med 2008;168:1340-9.
- [3] Rosen CJ. Clinical practice. Vitamin D insufficiency. N Engl J Med 2011;364:248-54.
- [4] Wang TJ, Pencina MJ, Booth SL, Jacques PF, Ingelsson E, Lanier K, et al. Vitamin D deficiency and risk of cardiovascular disease. *Circulation* 2008;117:503-11.
- [5] Melamed ML, Muntner P, Michos ED, Uribarri J, Weber C, Sharma J, et al. Serum 25hydroxyvitamin D levels and the prevalence of peripheral arterial disease: results from NHANES 2001 to 2004. Arterioscler Thromb Vasc Biol 2008;28:1179-85.
- [6] Zittermann A, Koerfer R. Vitamin D in the prevention and treatment of coronary heart disease. *Curr Opin Clin Nutr Metab Care* 2008;11:752-7.
- [7] Pilz S, Dobnig H, Fischer JE, Wellnitz B, Seelhorst U, Boehm BO, et al. Low vitamin d levels predict stroke in patients referred to coronary angiography. *Stroke* 2008;39:2611-3.
- [8] Pilz S, Dobnig H, Nijpels G, Heine RJ, Stehouwer CD, Snijder MB, et al. Vitamin D and mortality in older men and women. *Clin Endocrinol (Oxf)* 2009;71:666-72.
- [9] Ginde AA, Scragg R, Schwartz RS, Camargo CA, Jr. Prospective study of serum 25-hydroxyvitamin D level, cardiovascular disease mortality, and all-cause mortality in older U.S. adults. J Am Geriatr Soc 2009;57:1595-603.
- [10] Witham MD, Nadir MA, Struthers AD. Effect of vitamin D on blood pressure: a systematic review and meta-analysis. *J Hypertens* 2009;27:1948-54.
- [11] Pittas AG, Lau J, Hu FB, Dawson-Hughes B. The role of vitamin D and calcium in type 2 diabetes. A systematic review and meta-analysis. J Clin Endocrinol Metab 2007;92:2017-29.
- [12] Norman PE, Powell JT. Vitamin D, shedding light on the development of disease in peripheral arteries. *Arterioscler Thromb Vasc Biol* 2005;25:39-46.
- [13] Lee TH, Marcantonio ER, Mangione CM, Thomas EJ, Polanczyk CA, Cook EF, et al. Derivation and prospective validation of a simple index for prediction of cardiac risk of major noncardiac surgery. *Circulation* 1999;100:1043-9.
- [14] Touboul PJ, Hennerici MG, Meairs S, Adams H, Amarenco P, Desvarieux M, et al. Mannheim intima-media thickness consensus. *Cerebrovasc Dis* 2004;18:346-9.
- [15] Wendelhag I, Gustavsson T, Suurkula M, Berglund G, Wikstrand J. Ultrasound measurement of wall thickness in the carotid artery: fundamental principles and description of a computerized analysing system. *Clin Physiol* 1991;11:565-77.
- [16]
 Management of peripheral arterial disease (PAD). TransAtlantic Inter-Society Consensus (TASC).

 Section D: chronic critical limb ischaemia. Eur J Vasc Endovasc Surg 2000;19 Suppl A:S144-243.
- [17] Holick MF. Vitamin D deficiency. *N Engl J Med* 2007;357:266-81.
- [18] Pilz S, Marz W, Wellnitz B, Seelhorst U, Fahrleitner-Pammer A, Dimai HP, et al. Association of vitamin D deficiency with heart failure and sudden cardiac death in a large cross-sectional study of patients referred for coronary angiography. J Clin Endocrinol Metab 2008;93:3927-35.
- [19] Pilz S, Tomaschitz A, Marz W, Drechsler C, Ritz E, Zittermann A, et al. Vitamin D, cardiovascular disease and mortality. *Clin Endocrinol (Oxf)* 2011;75:575-84.
- [20] Ross AC, Manson JE, Abrams SA, Aloia JF, Brannon PM, Clinton SK, et al. The 2011 report on dietary reference intakes for calcium and vitamin D from the Institute of Medicine: what clinicians need to know. J Clin Endocrinol Metab 2011;96:53-8.
- [21] Forrest KY, Stuhldreher WL. Prevalence and correlates of vitamin D deficiency in US adults. *Nutr Res* 2011;31:48-54.
- [22] Gaddipati VC, Bailey BA, Kuriacose R, Copeland RJ, Manning T, Peiris AN. The relationship of vitamin D status to cardiovascular risk factors and amputation risk in veterans with peripheral arterial disease. J Am Med Dir Assoc 2011;12:58-61.
- [23] Fahrleitner A, Dobnig H, Obernosterer A, Pilger E, Leb G, Weber K, et al. Vitamin D deficiency and secondary hyperparathyroidism are common complications in patients with peripheral arterial disease. J Gen Intern Med 2002;17:663-9.

- [24] Fahrleitner-Pammer A, Obernosterer A, Pilger E, Dobnig H, Dimai HP, Leb G, et al. Hypovitaminosis D, impaired bone turnover and low bone mass are common in patients with peripheral arterial disease. Osteoporos Int 2005;16:319-24.
- [25] Zittermann A. Vitamin D and disease prevention with special reference to cardiovascular disease. *Prog Biophys Mol Biol* 2006;92:39-48.
- [26] Flu WJ, van Kuijk JP, Hoeks SE, Kuiper R, Schouten O, Goei D, et al. Intima media thickness of the common carotid artery in vascular surgery patients: a predictor of postoperative cardiovascular events. Am Heart J 2009;158:202-8.
- [27] Flu WJ, van Kuijk JP, Voute MT, Kuiper R, Verhagen HJ, Bax JJ, et al. Asymptomatic low anklebrachial index in vascular surgery patients: a predictor of perioperative myocardial damage. *Eur J Vasc Endovasc Surg* 2010;39:62-9.
- [28] Targher G, Bertolini L, Padovani R, Zenari L, Scala L, Cigolini M, et al. Serum 25-hydroxyvitamin D3 concentrations and carotid artery intima-media thickness among type 2 diabetic patients. *Clin Endocrinol (Oxf)* 2006;65:593-7.
- [29] Reis JP, von Muhlen D, Michos ED, Miller ER, 3rd, Appel LJ, Araneta MR, et al. Serum vitamin D, parathyroid hormone levels, and carotid atherosclerosis. *Atherosclerosis* 2009;207:585-90.
- [30] Reis JP, Michos ED, von Muhlen D, Miller ER, 3rd. Differences in vitamin D status as a possible contributor to the racial disparity in peripheral arterial disease. *Am J Clin Nutr* 2008;88:1469-77.
- [31] Drechsler C, Pilz S, Obermayer-Pietsch B, Verduijn M, Tomaschitz A, Krane V, et al. Vitamin D deficiency is associated with sudden cardiac death, combined cardiovascular events, and mortality in haemodialysis patients. *Eur Heart J* 2010;31:2253-61.
- [32] Kim DH, Sabour S, Sagar UN, Adams S, Whellan DJ. Prevalence of hypovitaminosis D in cardiovascular diseases (from the National Health and Nutrition Examination Survey 2001 to 2004). *Am J Cardiol* 2008;102:1540-4.
- [33] Davies MR, Hruska KA. Pathophysiological mechanisms of vascular calcification in end-stage renal disease. *Kidney Int* 2001;60:472-9.
- [34] Guillot X, Semerano L, Saidenberg-Kermanac'h N, Falgarone G, Boissier MC. Vitamin D and inflammation. *Joint Bone Spine* 2010;77:552-7.
- [35] Muller K, Haahr PM, Diamant M, Rieneck K, Kharazmi A, Bendtzen K. 1,25-Dihydroxyvitamin D3 inhibits cytokine production by human blood monocytes at the post-transcriptional level. *Cytokine* 1992;4:506-12.
- [36] Canning MO, Grotenhuis K, de Wit H, Ruwhof C, Drexhage HA. 1-alpha,25-Dihydroxyvitamin D3 (1,25(OH)(2)D(3)) hampers the maturation of fully active immature dendritic cells from monocytes. *Eur J Endocrinol* 2001;145:351-7.
- [37] Takeda M, Yamashita T, Sasaki N, Nakajima K, Kita T, Shinohara M, et al. Oral administration of an active form of vitamin D3 (calcitriol) decreases atherosclerosis in mice by inducing regulatory T cells and immature dendritic cells with tolerogenic functions. *Arterioscler Thromb Vasc Biol* 2010;30:2495-503.
- [38] Tarcin O, Yavuz DG, Ozben B, Telli A, Ogunc AV, Yuksel M, et al. Effect of vitamin D deficiency and replacement on endothelial function in asymptomatic subjects. *J Clin Endocrinol Metab* 2009;94:4023-30.
- [39] Jablonski KL, Chonchol M, Pierce GL, Walker AE, Seals DR. 25-Hydroxyvitamin D deficiency is associated with inflammation-linked vascular endothelial dysfunction in middle-aged and older adults. *Hypertension* 2011;57:63-9.
- [40] Zagura M, Serg M, Kampus P, Zilmer M, Eha J, Unt E, et al. Aortic stiffness and vitamin D are independent markers of aortic calcification in patients with peripheral arterial disease and in healthy subjects. *Eur J Vasc Endovasc Surg* 2011;42:689-95.
- [41] Zittermann A, Koerfer R. Protective and toxic effects of vitamin D on vascular calcification: clinical implications. *Mol Aspects Med* 2008;29:423-32.
- [42] Reynolds JA, Haque S, Berry JL, Pemberton P, Teh LS, Ho P, et al. 25-Hydroxyvitamin D deficiency is associated with increased aortic stiffness in patients with systemic lupus erythematosus. *Rheumatology (Oxford)* 2012;51:544-51.
- [43] Lanske B, Razzaque MS. Mineral metabolism and aging: the fibroblast growth factor 23 enigma. *Curr Opin Nephrol Hypertens* 2007;16:311-8.
- [44] Zittermann A, Schleithoff SS, Koerfer R. Putting cardiovascular disease and vitamin D insufficiency into perspective. *Br J Nutr* 2005;94:483-92.

- [45] Kooienga L, Fried L, Scragg R, Kendrick J, Smits G, Chonchol M. The effect of combined calcium and vitamin D3 supplementation on serum intact parathyroid hormone in moderate CKD. *Am J Kidney Dis* 2009;53:408-16.
- [46] Pfeifer M, Begerow B, Minne HW, Nachtigall D, Hansen C. Effects of a short-term vitamin D(3) and calcium supplementation on blood pressure and parathyroid hormone levels in elderly women. J Clin Endocrinol Metab 2001;86:1633-7.
- [47] Autier P, Gandini S. Vitamin D supplementation and total mortality: a meta-analysis of randomized controlled trials. *Arch Intern Med* 2007;167:1730-7.
- [48] Zittermann A, von Helden R, Grant W, Kipshoven C, Ringe JD. An estimate of the survival benefit of improving vitamin D status in the adult german population. *Dermatoendocrinol* 2009;1:300-6.

Blood thinners Statins Beta-blockers Vitamin D status

New-onset arrhythmias

5

Carotid stent characteristics Aortic stentgraft composition Laparoscopic sac fenestration after EVAR Need we intervene for type II endoleaks?

Submitted as:

"The incidence of new-onset perioperative arrhythmias in major vascular surgery may be much higher than expected"

Michiel T. Voûte Tamara A. Winkel Mirko de Melis Sanne E. Hoeks H.W.M. (Thijs) Plokker Robert Jan Stolker Hence J.M.Verhagen

ABSTRACT

Introduction

Cardiovascular complications, such as ischemic and/or arrhythmic events, are a major cause for morbidity and mortality after vascular surgery. The most frequent perioperative arrhythmia is new-onset atrial fibrillation (AF), with a reported incidence of 4-13%. AF is a known major risk factor for postoperative stroke, myocardial infarction and pulmonary embolism. Some of these complications may be preventable such as AF, and can be treated if diagnosed. Current standard Holter recording or periodical electrocardiography (ECG) cannot always identify these paroxysmal arrhythmias. An insertable cardiac monitor (ICM; Reveal XT, Medtronic) continuously monitors for a prolonged period of time, and has clinically proven its use in the detection of paroxysmal arrhythmias. The primary objective of the current study was to detect the true incidence of new-onset cardiac arrhythmias in patients undergoing major vascular surgery using an ICM, and to compare the results with a 72-hour Holter monitoring.

Material/Methods

After informed consent, patients undergoing major vascular surgery received a 72-hour Holter and an ICM prior to elective surgery. After 72 hours the Holter was removed and the data collected, while the ICM data was retrieved electronically from the implanted device. One month after surgery the ICM was removed after a final read-out of the data. Collected data was reviewed independently by a senior cardiologist and an independent core lab, scoring AF, sustained ventricular tachycardia, or ventricular flutter. Fisher's exact test was used to compare Reveal and Holter outcomes. Inter-rater agreement was reviewed by Cohen's kappa.

Results

A total of 43 patients were monitored successfully with both devices. Implantation of the Reveal took approximately 20 minutes under local anesthesia. One patient was diagnosed chronic AF, despite normal ECG recording at the outpatient clinic. Of the remaining 42 patients, 5 patients (11.9%) developed arrhythmias according to Reveal, and only 1 (2.4%) on Holter monitoring in the first 72-hours after surgery. Up to one month postoperatively, new-onset arrhythmias were detected in 11 patients (26.2%) with ICM, comprising paroxysmal AF in 10 cases (90.9%). Sustained VT was detected in one case (2.4%). Inter-rater agreement was 94.9%, with a 0.88 kappa.

Conclusions

The true incidence of paroxysmal AF after major vascular surgery seems to be much higher than estimated with standard practice Holter monitoring, leaving most cases currently undetected and untreated. An insertable cardiac monitor, evaluating rhythm disturbances continuously for weeks or months before and after surgery, detects patients at risk for thromboembolic complications reliably, opening a new treatment window for these patients. Future research should be aimed at the optimal treatment strategy for this patient category.

MANUSCRIPT

INTRODUCTION

Vascular surgery is associated with a high risk of cardiovascular complications, such as cardiac ischemia, stroke and cardiac arrhythmias.[1] Cardiac arrhythmias most often occur during surgery or within the first few postoperative days.[2, 3] Fluid challenges, vagal triggering, cardiac stress and medication are all among the possible contributing factors.[4, 5]

The most frequent perioperative arrhythmia is atrial fibrillation (AF).[6] New-onset AF in vascular surgery has a reported incidence of 4-13%[7-10], and is a known major risk factor for postoperative stroke, myocardial infarction, congestive heart failure and pulmonary embolism.[11, 12] Furthermore, perioperative arrhythmias are directly associated with morbidity, long-term mortality and increased length of hospital stay and health costs.[13-15]

The majority of perioperative arrhythmias is asymptomatic, transient and unpredictable.[16, 17] This causes arrhythmias to be frequently missed and, hence, undertreated. Using traditional detection strategies - such as serial ECG, in-hospital telemetry or even Holter monitoring - the true incidence of perioperative arrhythmias are possibly underestimated.[18, 19] Longer continuous cardiac monitoring could have an advantage over Holter monitoring and significantly increase the number of detected arrhythmias.[20] The use of an insertable cardiac monitor (ICM) has been validated to detect paroxysmal AF, and proven effective in out-patients.[21] Revealing the true incidence of perioperative arrhythmias and raising awareness may open a new treatment window for patients undergoing major vascular surgery, potentially improving outcome.

The primary objective of the current study was to estimate the true incidence of newonset cardiac arrhythmias in major vascular surgery patients by making use of an ICM, and to compare it to the present gold standard, the 72-hour Holter device.

Table 1. Inclusion and exclusion criteria						
Inclusion criteria	Exclusion criteria					
 Elective AAA repair or open surgical lower extremity revascularization Willingness to wear a 72-hour Holter and receive an ICM perioperatively Written informed consent Aged 18 years and over 	 An implanted pacemaker or implantable cardioverter defibrillator Physiological of ECG abnormalities that preclude assessment of cardiac arrhythmias A history of cardiac arrhythmia Legal incompetence Pregnancy 					

METHODS

This study was a non-randomized, open-label, single-center prospective pilot study. The study was conducted from 2008 to 2012 at the department of Vascular Surgery, Erasmus Medical Center, Rotterdam, the Netherlands.

Subject enrollment

All patients without a history of arrhythmias scheduled for elective aortic abdominal aneurysm repair or open surgical lower extremity revascularization at the Erasmus Medical Centre were approached to participate in this study. Inclusion and exclusion criteria are presented in Table 1. The study was compliant with the international standard for clinical investigation of medical devices in human subjects, ISO 14155, and the Declaration of Helsinki. The study protocol was approved by the institutional Medical Ethics Committee (MEC-2008-130).

Baseline characteristics of the patients, including medical history and possible risk factors for the development of new-onset arrhythmias were prospectively registered. These included advanced age, a history of congestive heart failure (CHF), hypertension (defined as systolic blood pressure over 140mmHg or diastolic blood pressure over 90mmHg or the use of antihypertensive drugs), renal insufficiency (serum creatinin >2 mg/dl) and preoperative low serum potassium levels.[3] Furthermore, the CHA₂DS₂-VASc score was documented. The CHA₂DS₂-VASc score is the main prediction model for stroke in patients with (persistent of paroxysmal) AF, and includes hypertension, age over 65, diabetes and vascular disease, among other risk factors.[22]

Cardiac rhythm monitoring

All patients were monitored in the perioperative period using two devices, a 72-hour surface ECG recording system (12-lead DR180 Digital Holter Recorder; NorthEast Monitoring Inc., Maynard, MA, USA), and an ICM (Reveal[®] XT: Medtronic, Minneapolis, MN, USA), as illustrated in Figure 1. Reveal® is a singleuse device containing two electrodes on the body of the device for continuous (i.e., looping) recording of the patient's subcutaneous ECG. The Holter was applied the day before surgery and ran for up to 72 hours. The ICM was inserted approximately one month prior to surgery and removed one month after surgery.



Intermediate reports from the ICM were collected one day prior to surgery, 2 to 3 days after surgery and after one month. All Holter and ICM application and data collection was subject to the elective planning of the surgical procedures.

Study endpoints

The primary endpoint was any episode of new-onset arrhythmia, comprising AF or sustained ventricular tachycardia (VT), detected either by Holter or ICM. Definitions of the reported episodes of arrhythmia for this study are defined in Table 2. Secondary endpoints for the study were long-term survival and length of hospital stay. Possible adverse events in this study included ICM device rejection phenomena that participants could encounter, including local tissue reaction, device migration, infection and erosion through the skin.

Data analysis of all ICM and Holter recordings was performed by an independent senior cardiologist. An independent core lab (Cardialysis, Rotterdam, The Netherlands), blinded for the results, analyzed a sample of twenty ICM and twenty Holter recordings to verify the outcome.

Statistics

Baseline characteristics of participants are presented as medians with the interquartile range for continuous variables and categorical data are presented as counts and percentages. Use of monitoring, arrhythmia detection, risk factors, length of hospital stay and survival were compared between patients with and without new-onset arrhythmias. Considering the small number of patients in this pilot study, non-parametric tests were used to compare these two groups, e.g. Kruskal-Wallis tests for continuous data and Fisher's exact test tests for categorical data. Differences in mortality between patients with and without a detected event were calculated by a log-rank test, taking into account the length of follow-up after surgery. Agreement between the independent cardiologist and the core lab was verified using Cohen's Kappa. All statistical analyses were performed on a windows-based computer, using SPSS version 21 (IBM Corporation, Armonk, NY, USA).

Table 2. Definitions of reported arrhythmias				
Type of arrhythmia	Definition			
Atrial fibrillation (AF)	An episode of at least 30 seconds in which the RR intervals follow no repetitive pattern, but can be labeled as 'irregularly irregular'. Also no distinct P waves can be detected. Defined as paroxysmal AF (PAF) when episodes recur at least once, and last for no more than 7 days.			
Sustained ventricular tachycardia (VT)	A sequence of ventricular beats, with a frequency higher than 100 beats per minute (bpm). Sustained implies a duration of more than 30 seconds.			

RESULTS

Study population

A total of forty-nine patients had an ICM device implanted, of which one subsequently did not undergo surgery due to health issues. One patient suffered from surgical site infection, therefore the ICM was removed prior to surgery with no further clinical consequences. This resulted in an ICM implantation adverse event rate of 2.0% (1/49). Of the 47 remaining cases, Holter lead wires were disconnected in four patients upon arrival at the OR or at the ICU postoperatively, as they were judged to interfere with clinical ECG monitoring by tending physicians. No postoperative Holter data was recovered for these patients. All patients without postoperative data from either Holter or ICM monitoring were excluded from the analysis. Finally, one case was diagnosed with chronic AF on both ICM and Holter monitoring, despite normal ECG recordings at the outpatient clinic. Exclusion of this case resulted in 42 patients in the final analysis (Figure 2).



The majority of subjects was male (85.7) and median age was 69 (interquartile range 61.7-74.8). A total of 14 patients (33.3%) underwent open aortic surgery, while 18 (42.9%) underwent endovascular aneurysm repair (EVAR) of an infrarenal aneurysm, and 10 patients (23.8%) underwent lower extremity revascularization. Medical history of the study population, as illustrated in Table 3, was comparable to any general vascular surgery population.

Table 3. Baseline characteristics				
Variable	Total (n=42)			
General statistics				
Male gender, n (%)	36 (85.7)			
Age, years	69.0 (61.7 – 74.8)			
Open abdominal aortic surgery, n (%)	14 (33.3)			
Endovascular aneurysm repair, n (%)	18 (42.9)			
Lower extremity revascularization, n (%)	10 (23.8)			
Medical history				
Ischemic heart disease, n (%)	18 (42.9)			
CHF, n (%)	3 (7.1)			
Stroke, n (%)	4 (9.5)			
Renal insufficiency, n (%)	8 (19.0)			
Hypertension, n (%)	33 (18.6)			
High cholesterol, n (%)	36 (85.7)			
Diabetes, n (%)	8 (19.0)			
COPD Gold 3-4, n (%)	3 (7.1)			
BMI	26.3 (23.0 – 29.4)			
Serum potassium (mmol/l)	4.5 (4.3 – 4.8)			
Rhythm registration				
Holter recording, days	3 (3-3)			
ICM recording, days	30 (30-30)			
Note: CHF denotes congestive heart failure, COPD chronic obstructive pulmonary				
disease, ICM insertable cardiac monitor, BMI body n	nass index. Continuous values			
are presented as median and interquartile range.				

New-onset arrhythmia detection

The total reported incidence of new-onset arrhythmias in the study was 11 (26.2%) with ICM monitoring, yet only 1 (2.4%) with Holter monitoring (Table 4). The median time of arrhythmia detection was 5 days (IQR 1-12) after surgery. The one arrhythmia both devices identified was an episode of sustained VT just hours after surgery, which lasted for well over a minute. This was accompanied by a lack of cardiac output, for which CPR was administered briefly, before restoration of sinus rhythm and cardiac output.

Table 4. New-onset arrhythmia detection of Holter and ICM over time										
Device	Timeline of arrhythmia detection					Events, n (%)				
Holter	V									1 (2.4)
ICM	VA	AA	А	А	А	А	А	А	А	11 (26.2)
	0	1	2	5	6	10	12	20	30	
Time (days after surgery)										
A denotes atrial fibrillation, V sustained VT. Grey boxes display the intended length of recording by each device.										

An additional 10 cases, all with paroxysmal AF, were identified only by the ICM, of which 6 occurred outside the 72-hour window of Holter monitoring.

There was an inter-observer agreement between the core-lab and the independent cardiologist regarding arrhythmia detection in nearly ninety-five percent of observations (Kappa: 0.88, p<0.001). The median CHA_2DS_2 -VASc score of 10 patients with new-onset AF was 4.0 (IQR 2.5-4.25), with 8 out of 2 cases (80.0%) gathering a score of ≥ 2 on this risk index. Of these 8, three cases died shortly after surgery, respectively due to graft infection and shock (n=1), respiratory failure (n=1) and a myocardial infarction (n=1). Upon discharge of the remaining five cases, 3 received oral anticoagulants and 2 were only treated with aspirin.

Secondary endpoints

Possible risk factors associated with new-onset arrhythmias were not found to be significantly increased in the patients that suffered an event, compared to the control group (Table 5). A significant increase in length of hospital stay was associated with perioperative arrhythmias (median 18.0 days) in comparison with patients that had an event-free discharge (median 4.0 days; P=0.002). Also, a higher 3-year mortality rate was associated with perioperative arrhythmias during long-term follow-up (54.5% versus 6.5%, P=0.01).

Variable	New-onset arrhythmia (n=11)	No arrhythmia (n=31)	P value
General statistics			
Age, years	65.5 (62.6 – 74.7)	69.3 (61.7 – 74.9)	0.81
CHF, n (%)	1 (9.1)	2 (6.5)	1.0
Renal insufficiency, n (%)	4 (36.4)	4 (12.9)	0.17
Hypertension, n (%)	9 (81.8)	24 (77.4)	1.0
Serum potassium (mmol/l)	4.5 (4.4 – 4.7)	4.5 (4.2 – 4.8)	0.92
Secondary outcome			
Length of hospital stay, days	18.0 (6.0-37.0)	4.0 (2.0-7.0)	0.002
Survival, years	2.4 (0.1-4.3)	3.2 (2.3-4.0)	0.32
3-year mortality, n (%)	6 (54.5)	2 (6.5)	0.01

DISCUSSION

In this pilot study, the percentages of AF and sustained VT registered with Holter (2.4%) are similar to the historical estimate. However, the reported incidence of new-onset AF was over twenty-five percent using an ICM. The possibilities of an ICM to monitor continuously for a longer period was of direct influence on the number of arrhythmias detected in the perioperative period, as suggested in other studies.[24, 25] This could be of advantage to current methods, e.g. routine intermittent 12-lead ECGs or continuous surface electrocardiography during no more than a few days of ICU stay.[26-29]

Outside the field of vascular surgery, the increased diagnostic yield of the ICM for arrhythmia detection was already embraced earlier; for instance in neurology, among patients suffering from cryptogenic stroke.[30] In over two hundred patients randomized to receive an ICM, the detected incidence of AF was more than six times greater than in an equal sized group that was randomized to receive standard follow-up after cryptogenic stroke.[31] In cardiology, ICM implantation helped to identify almost twice as many patients with recurrent AF after trans-catheter ablation, when compared to regular follow-up and periodical 12-lead ECG screening.[32].

The large difference in arrhythmias detection in the current study is explained by the prolonged duration of ICM monitoring versus Holter monitoring, in part at least. However, even in the first 48-hours after surgery – the time-frame of postoperative Holter monitoring – the ICM registered three times more arrhythmias than the Holter. A possible explanation could be that body sweat, external manipulation and battery failure cause a Holter to detach, record interference, or terminate, impairing proper rhythm analysis.[33] Although this may be a weakness of the study, it is even more so an indication of the weakness of 12-lead surface cardiac monitoring. There are other, wireless alternatives that are not invasive, which also proved useful in arrhythmia detection in the first 14 days after implantation.[34]

Apart from a much higher reported incidence, the current study also suggest that patients suffering from paroxysmal AF not only have a longer hospital stay, but also have a shorter life expectancy. After identifying patients with new-onset AF, it is key to imbed a proper treatment strategy in clinical practice. This should include referral to a cardiologist to consider medical treatment of AF, as well as risk management for late thromboembolic adverse events.[35]

According to the 2010 Guidelines from the European Society of Cardiology on the treatment of (persistent or paroxysmal) AF, harboring one of these risk factors already merits treatment with aspirin, but preferably oral anticoagulant therapy (OAC).[35] When harboring two risk factors, e.g. a vascular surgery patient with hypertension or advanced age, it is strictly advised to start OAC upon detection of AF. The patients with new-onset arrhythmia in our study had a score of ≥ 2 in the large majority of cases, but not all received anticoagulant therapy at the time. Currently, no specific clinical data is

available on the health benefits of starting OAC in vascular surgery patients with at least one more CHA₂DS₂-VASc risk factor, suffering from a brief episode of paroxysmal AF. Future research is needed to develop a comprehensive treatment guideline for this particular patient population.

This is the first study to report the use of the ICM in vascular surgery patients without a history of cardiac arrhythmias. Results show that the incidence of new-onset paroxysmal AF is higher than generally believed. However, being able to identify these cases depends on their willingness to undergo ICM implantation prior to a vascular procedure. In the process of enrollment for this study, only a small proportion of patients were willing to undergo implantation of the ICM. This was the main factor in the slow enrollment of this pilot study, and could reduce future perspectives for this type of monitoring. With the recent introduction of an even smaller ICM, the Reveal LINQ[™], the threshold for patients to undergo implantation of such a device will be lower, since implantation is easier and causes less morbidity.[36]

Eventhough this is a pilot study in which only a "proof of concept" was studied, it still has limitations such as the relatively small number of patients included. Second, slow enrollment could have introduced a selection bias. However, gold-standard Holter monitoring led to a similar incidence of new-onset arrhythmia as current literature, and baseline characteristics showed our population was exemplary for the general surgery population. Third, it remains unclear what the clinical significance is for paroxysmal AF, detected by an ICM. Although this pilot was not powered for it, detected arrhythmias (in post-analysis) were related to a longer hospital stay and a higher long-term mortality.

CONCLUSION

The true incidence of paroxysmal AF after major vascular surgery seems to be much higher than estimated with standard practice Holter monitoring, leaving most cases currently undetected. An insertable cardiac monitor, evaluating rhythm disturbances continuously for weeks or months after surgery, detects patients at risk for thromboembolic complications reliably, opening a new treatment window for these patients. Future research should be aimed at the optimal treatment strategy for this patient category.

REFERENCES

- Gerson, M.C., et al., Prediction of cardiac and pulmonary complications related to elective abdominal and noncardiac thoracic surgery in geriatric patients. Am J Med, 1990. 88(2): p. 101-7.
- 2. Polanczyk, C.A., et al., Supraventricular arrhythmia in patients having noncardiac surgery: clinical correlates and effect on length of stay. Ann Intern Med, 1998. **129**(4): p. 279-85.
- Mathew, J.P., et al., A multicenter risk index for atrial fibrillation after cardiac surgery. JAMA, 2004. 291(14): p. 1720-9.
- Rogers, W.K. and K.M. Schroeder, Perioperative atrial fibrillation and epidural anesthesia: case report and review of the literature. J Clin Anesth, 2012. 24(4): p. 329-33.
- Ommen, S.R., J.A. Odell, and M.S. Stanton, Atrial arrhythmias after cardiothoracic surgery. N Engl J Med, 1997. 336(20): p. 1429-34.
- Walsh, S.R., et al., Postoperative arrhythmias in general surgical patients. Ann R Coll Surg Engl, 2007. 89(2): p. 91-5.
- Sposato, L.A., et al., Intraoperative hypotension, new onset atrial fibrillation, and adverse outcome after carotid endarterectomy. J Neurol Sci, 2011. 309(1-2): p. 5-8.
- Valentine, R.J., et al., The clinical course of new-onset atrial fibrillation after elective aortic operations. J Am Coll Surg, 2001. 193(5): p. 499-504.
- Noorani, A., et al., Atrial fibrillation following elective open abdominal aortic aneurysm repair. Int J Surg, 2009. 7(1): p. 24-7.
- 10. Hafez, H., et al., *Transverse minilaparotomy for open abdominal aortic aneurysm repair*. J Vasc Surg, 2011. **53**(6): p. 1514-9.
- 11. Flaker, G.C., et al., Asymptomatic atrial fibrillation: demographic features and prognostic information from the Atrial Fibrillation Follow-up Investigation of Rhythm Management (AFFIRM) study. Am Heart J, 2005. **149**(4): p. 657-63.
- 12. Creswell, L.L., et al., *Hazards of postoperative atrial arrhythmias*. Ann Thorac Surg, 1993. **56**(3): p. 539-49.
- 13. Aranki, S.F., et al., *Predictors of atrial fibrillation after coronary artery surgery. Current trends and impact on hospital resources.* Circulation, 1996. **94**(3): p. 390-7.
- Kinjo, K., et al., Prognostic significance of atrial fibrillation/atrial flutter in patients with acute myocardial infarction treated with percutaneous coronary intervention. Am J Cardiol, 2003. 92(10): p. 1150-4.
- 15. Melduni, R.M., Y. Koshino, and W.K. Shen, *Management of arrhythmias in the perioperative setting*. Clin Geriatr Med, 2012. **28**(4): p. 729-43.
- 16. Healey, J.S., et al., *Subclinical atrial fibrillation and the risk of stroke*. N Engl J Med, 2012. **366**(2): p. 120-9.
- Seet, R.C., P.A. Friedman, and A.A. Rabinstein, Prolonged rhythm monitoring for the detection of occult paroxysmal atrial fibrillation in ischemic stroke of unknown cause. Circulation, 2011. 124(4): p. 477-86.
- 18. Hendrikx, T., et al., *Intermittent short ECG recording is more effective than 24-hour Holter ECG in detection of arrhythmias.* BMC Cardiovasc Disord, 2014. **14**: p. 41.
- 19. Doliwa, P.S., M. Rosenqvist, and V. Frykman, *Paroxysmal atrial fibrillation with silent episodes: intermittent versus continuous monitoring.* Scand Cardiovasc J, 2012. **46**(3): p. 144-8.
- 20. Ziegler, P.D., J.L. Koehler, and R. Mehra, *Comparison of continuous versus intermittent monitoring of atrial arrhythmias*. Heart Rhythm, 2006. **3**(12): p. 1445-52.
- 21. Hindricks, G., et al., *Performance of a new leadless implantable cardiac monitor in detecting and quantifying atrial fibrillation: Results of the XPECT trial.* Circ Arrhythm Electrophysiol, 2010. **3**(2): p. 141-7.
- 22. Lip, G.Y., et al., Refining clinical risk stratification for predicting stroke and thromboembolism in atrial fibrillation using a novel risk factor-based approach: the euro heart survey on atrial fibrillation. Chest, 2010. **137**(2): p. 263-72.
- 23. Cyr, L. and K. Francis, Measures of clinical agreement for nominal and categorical data: the kappa coefficient. Comput Biol Med, 1992. **22**(4): p. 239-46.
- 24. Charitos, E.I., et al., A comprehensive evaluation of rhythm monitoring strategies for the detection of atrial fibrillation recurrence: insights from 647 continuously monitored patients and implications for monitoring after therapeutic interventions. Circulation, 2012. **126**(7): p. 806-14.

- Hanke, T., et al., Twenty-four-hour holter monitor follow-up does not provide accurate heart rhythm status after surgical atrial fibrillation ablation therapy: up to 12 months experience with a novel permanently implantable heart rhythm monitor device. Circulation, 2009. 120(11 Suppl): p. S177-84.
- 26. Batra, G.S., J. Molyneux, and N.A. Scott, *Colorectal patients and cardiac arrhythmias detected on the surgical high dependency unit.* Ann R Coll Surg Engl, 2001. **83**(3): p. 174-6.
- 27. Brathwaite, D. and C. Weissman, *The new onset of atrial arrhythmias following major noncardiothoracic surgery is associated with increased mortality.* Chest, 1998. **114**(2): p. 462-8.
- Goldman, L., Supraventricular tachyarrhythmias in hospitalized adults after surgery. Clinical correlates in patients over 40 years of age after major noncardiac surgery. Chest, 1978. 73(4): p. 450-4.
- 29. Walsh, S.R., et al., *Postoperative arrhythmias in colorectal surgical patients: incidence and clinical correlates.* Colorectal Dis, 2006. **8**(3): p. 212-6.
- 30. Sinha, A.M., et al., *Cryptogenic Stroke and underlying Atrial Fibrillation (CRYSTAL AF): design and rationale*. Am Heart J, 2010. **160**(1): p. 36-41 e1.
- Sanna, T., et al., Cryptogenic stroke and underlying atrial fibrillation. N Engl J Med, 2014. 370(26): p. 2478-86.
- 32. Manganiello, S., et al., *Symptomatic and asymptomatic long-term recurrences following transcatheter atrial fibrillation ablation*. Pacing Clin Electrophysiol, 2014. **37**(6): p. 697-702.
- 33. Ritter, M.A., et al., Occult atrial fibrillation in cryptogenic stroke: detection by 7-day electrocardiogram versus implantable cardiac monitors. Stroke, 2013. 44(5): p. 1449-52.
- Turakhia, M.P., et al., Diagnostic utility of a novel leadless arrhythmia monitoring device. Am J Cardiol, 2013. 112(4): p. 520-4.
- 35. European Heart Rhythm, A., et al., Guidelines for the management of atrial fibrillation: the Task Force for the Management of Atrial Fibrillation of the European Society of Cardiology (ESC). Eur Heart J, 2010. 31(19): p. 2369-429.
- 36. Tomson, T.T. and R. Passman, *The Reveal LINQ insertable cardiac monitor*. Expert Rev Med Devices, 2014: p. 1-12.

Technical Considerations

90 |

Blood thinners Statins Beta-blockers Vitamin D status New-onset arrhythmias



Carotid stent characteristics

Aortic stentgraft composition Laparoscopic sac fenestration after EVAR Need we intervene for type II endoleaks?

Published as:

"Radial force measurements in carotid stents: influence of stent design and length of the lesion"

Journal of Vascular and Interventional Radiology 2011;22:661-666

Michiel T. Voûte Joke M. Hendriks Jorinde H.H. van Laanen Peter M.T. Pattynama Bart E. Muhs Don Poldermans Hence J.M.Verhagen

ABSTRACT

Purpose

To assess the differences in radial force of carotid stents, and to evaluate if the length of the lesion is influential to the measurements.

Materials and methods

Similar sized, tapered stents (length 30mm) of different make and models were used. The tapered nitinol Acculink[®], Protégé[®] and Cristallo Ideale[®] carotid artery stents, and the straight, braided elgiloy carotid Wallstent[®] were compared. A measurement device was developed, consisting of three film loops along the stent body, connected to aluminium rods armed with copper strain gauges. Five stents of each type were deployed within a 3mm stenosis, in a long (26mm) and in a short (8mm) stenosis simulation.

Results

In the short stenosis simulation, the greatest radial force was seen in the Protégé, reaching 3.14 ± 0.45 Newton, followed by the Cristallo Ideale (1.73 ± 0.51 N), the Acculink (1.16 ± 0.21 N) and the Wallstent (0.84 ± 0.10 N)(P<0.001). In the long stenosis simulation, peak radial force again was highest in the Protégé (1.67 ± 0.37 N), but second was the Acculink (0.95 ± 0.12 N) and third, the Wallstent (0.80 ± 0.06 N). The Cristallo Ideale, in large contrast with a short stenosis, produced the least radial force (0.44 ± 0.13 N) in a long stenosis simulation (P= 0.001).

Conclusions

Radial forces exerted by carotid stents vary significantly between various stent designs. Differences between stent types are dependent of the length of the stenosis. An understanding of radial force is necessary for a well-considered choice of stent type in the individual patient.

MANUSCRIPT

INTRODUCTION

Carotid angioplasty and stenting (CAS) is a frequent treatment modality for carotid artery stenosis.[1,2] The most important complication of CAS is stroke, due to emboli or hemodynamic depression (HD).[3-9] According to literature, several properties of the carotid stent and the delivery system play an important role in the final outcome.[10] The main concerns for procedural efficacy are trackability in straight or tortuous arteries, scaffolding of hard or soft plaques and achieving optimal patency. These individual stent properties have been investigated on many occasions, for better understanding and possible improvement of the technique.

Trackability was evaluated by comparing strut behaviour of open versus closed cell designs in a curved vessel model.[11] Additionally, the force needed for bending and the torsion angle were reported in another effort to compare trackability between open and closed stents.[12] Scaffolding properties and free cell area have been assumed to play a role in the embolic complications rate.[12] While open cell designs were observed to have a higher rate of emboli in one study, another study reported conflicting results, leaving room for debate.[13,14] Regarding achieved patency rates after CAS, there are numerous studies on the safety and efficacy with different types of carotid stents.[15,16] The difficulty in analyzing results from all these reports, is that plaque constitution is often not taken into account. Therefore, the results may be influenced by patient-related factors. In theory, patency rates are primarily ascribed to the radial force of carotid stents.

In this study, we aim to measure the radial force in three frequently used nitinol selfexpanding stents and in one type of braided elgiloy stents, in both a short and a long stenosis. This can provide information for comparison, and support future decisionmaking in the choice of stent type in the individual patient.

METHODS

Stent types

In this comparative study we tested the nitinol Acculink[®] (Guidant Corporation, Santa Clara, CA, USA), Protégé[®] (EV3, Plymouth, MN, USA) and Cristallo Ideale[®] (Invatec, Frauenfeld, Switzerland) carotid artery stents, and the braided elgiloy carotid Wallstent[®] (Boston Scientific, Natick, MA, USA). Stents of all four types were distributed freely and without terms of agreement. The stent designs are shown in figure 1. The Acculink stent has an open cell design, with connecting struts aligned throughout the length of the stent. The Protégé stent has a cell design with many connecting struts at alternating locations on the stent, but is still regarded an open cell design. The Cristallo Ideale has few connected. This combination of open and closed cell design is called hybrid. The Wallstent has a closed cell design over the entire length of the stent. Another difference in design is that the Acculink and Cristallo Ideale have a gradual tapered shape over the length of the stent, while the Protégé has a bottleneck shape in



the middle section. The Wallstent is not tapered, but has a straight tube-like design, as can be seen on the left of figure 1.

For comparison purposes we chose to use stents of similar dimensions for all four types. The stent size according to manufacturer's data was 6-8x30mm for Protégé and Acculink stents, 6-9x30mm for Cristallo Ideale stents and 7x30mm for the Wallstent.

Measuring device

A testing method was used, similar to one previously reported by Duda et al.[17] The device is shown in figure 2. Basic elements of the device are three film sheets that are fixed on one end, looped around the distal, middle or proximal section of

the stent and connected to aluminium rods on the other end. By changing the distance to the aluminium rods from the set-up, the diameter of the film loops could be adjusted. The films are made of biaxially-oriented polyethylene terephthalate, or BOPET, with a width of each film loop of 8mm. The radial (or expansive) force of the stent is translated to pulling strength of the film on the aluminium rods. Copper strain gauges, forming a half Wheatstone bridge, on either side of the rods measure the strain in the aluminium. Pre-tests with calibrated weights were executed to allow for conversion from measured currents to radial forces in Newtons.

Radial force measurement

Testing was performed in a heated booth, at a constant temperature of 36°C. We tested five stents of every type, from different batches. Due to the obvious differences in stent design, blinding was not possible. Each stent was deployed inside the loops at a diameter of 3mm, the most common diameter of endovascular balloons used in our clinic for predilatation. This is a short procedure prior to carotid stenting, where a high-grade stenosis is widened slightly by balloon-dilatation, so that the delivery device of the carotid stent can safely pass the lesion. Generally, 3mm is a standard diameter through which most delivery devices can pass. The diameter of the loops was enlarged to full expansion on all three sections, gradually deploying the stent inside the device to



prevent displacement of the stent due to shortening. Once satisfied with stent placement inside the three loops, with the middle section of the stent located inside the middle film loop, simulations started.

For a long (26mm) stenosis, just shorter than the manufacturer's data on stent length, the diameter of all three loops was adjusted to 3mm and then gradually and simultaneously increased with increments of a 1mm, up to full expansion. The sum of radial force within all three loops was measured and tabulated. For a short (8mm) stenosis, located in the center of the stent position, only the middle loop was narrowed. Subsequently, radial force was only measured in the middle loop during the simulation of a short stenosis. For steady fixation, both outer loops were adjusted to fit the fully expanded ends of the stent, tapered in all but the Wallstent. A short stenosis was then simulated by adjusting the middle loop to a diameter of 3mm, as illustrated in figure 3. The diameter of the loop was then increased with increments of 1mm, up to full deployment. Measurements of the radial force were performed on each interval. We repeated this procedure 3 times for each stent, treating all measurements as individual results.

Statistics

Results were analysed using non-parametric tests. To compare all stent types at different diameters, the Kruskal-Wallis test was used. Peak radial force differences between stent types were then separately analyzed, using the Mann-Whitney test. Statistical differences with a p-value below 0.05 were considered significant. All analyses were performed using the Statistical Package for Social Sciences (version 15; SPSS Inc., Chicago, IL, USA).



RESULTS

Stent placement within the device was successful in all cases. No stents showed failure or manufacturing flaws. The radial force of the stents was measured at diameters ranging from 3-6 mm and compared for all four stent types (Table 1). In the short stenosis simulation, the greatest radial force was seen in the Protégé, reaching 3.14±0.45 Newton, followed by the Cristallo Ideale (1.73±0.51 N), the Acculink (1.16±0.21 N) and the Wallstent (0.84±0.10 N)(P<0.001). At increasing stent diameters the radial force stepwise decreased in all four stent types. Although the radial force in the Protégé decreased most rapidly, its residual radial force remained higher than all other stent types up to a diameter of 5mm. With all stent diameters at 6mm, the residual radial force in the acculink was the highest of the four types.

Table 1. Radial forces in a short and long stenosis at different diameters.							
Radial force, N	Protégé	Cristallo	Acculink	Wallstent	p-value		
Short stenosis (8mm)							
Ø 3.0 mm	3.10±0.45	1.73±0.51	1.16±0.21	0.84±0.10	< 0.001		
Ø 4.0 mm	0.95±0.13	0.50±0.08	0.40±0.05	0.29±0.03	< 0.001		
Ø 5.0 mm	0.34±0.10	0.28±0.10	0.26±0.04	0.15±0.02	< 0.001		
Ø 6.0 mm	0.06±0.04	0.12±0.04	0.16±0.04	-	< 0.001		
Long stenosis (26mm)							
Ø 3.0 mm	1.67±0.37	0.44±0.13	0.95±0.12	0.80±0.06	0.001		
Ø 4.0 mm	0.82±0.36	0.26±0.11	0.74±0.13	0.55±0.09	0.013		
Ø 5.0 mm	0.39±0.25	0.18±0.08	0.63±0.11	0.28±0.12	0.019		
Ø 6.0 mm	0.10±0.11	0.10±0.07	0.51±0.10	-	0.011		
Continuous data are represented as means (in Newtons) ± standard deviation P-values are calculated with the Kruskal-Wallis test Ø: outer diameter							

In the long stenosis simulation, peak radial force again was highest in the Protégé (1.67 \pm 0.37 N), but second was the Acculink (0.95 \pm 0.12 N) and third, the Wallstent (0.80 \pm 0.06 N). The Cristallo Ideale, in large contrast with a short stenosis, produced the least radial force (0.44 \pm 0.13 N) in the long stenosis simulation (P = 0.001). Differences between all four stent types were highly significant at all diameters, both in a short (8mm) and a long (26mm) stenosis, according to Kruskal-Wallis statistics.

Separate analyses of the radial force differences between each two stent types were then performed. Differences between any two stents were found to be significant, according to Mann-Whitney U statistics (Figure 4).



DISCUSSION

In the current study, a method was presented for comparative radial force measurements in different types of carotid stents, in both a short and a long stenosis. Significant differences were observed between stent types, and also a clear influence of lesion length was noticeable on the total radial force, exerted by the stents.

Radial force measurements have been previously reported using a dual plate compression test.[18] However, this is more a measurement of force during deformity rather than reduced lumen diameters. A stenosis results in a restriction of the luminal volume. However, during plate compression, the diameter is reduced in the vertical direction, but expands in the horizontal direction, much like a balloon that is

compressed. The volume is not altered, and the reactive force of the stent during plate compression is quite different from a test where the actual luminal volume is reduced. A circular compression, loop-strap or hoop strength test was used in previous reports on carotid stents, biliary stents and stents used in stenosed native aortic valves.[12,19,20]

In the current report, we set out to compare the radial force of three different nitinol stents and the elgiloy Wallstent. Our results clearly show that all tested nitinol stents produce a higher radial force than the elgiloy Wallstent. The Protégé generated a radial force far greater than all others, in both a short and a long stenosis simulation. The hybrid design Cristallo Ideale produces a large radial force in a short stenosis, compared to the Acculink and the Wallstent, but loses the majority of its expansive capacity in a longer stenosis. It produces less radial force than any other stent type measured in a long stenosis in the current study. This dependency of lesion length in the Cristallo Ideale is probably related to the hybrid design of the stent. The middle section (closed cell) and the outer sections (open cell) are of different composition, which may cause a discrepancy in the performance in a stenosis that is limited to the middle section and a full length stenosis. Another interesting finding was the great difference between two open cell design stents. Of the three nitinol stents we tested, the open cell Protégé exerted the highest radial force, and the open cell Acculink the lowest. In clinical literature, results with open or closed cell designs are often clustered as a group. Our results show that this is erroneous, and clinical results with one stent should never be generalized for all others from the same – open or closed cell – group.

In the individual stent selection, radial force should be taken into account. Reading into Bosiers et al., one could argue that a high degree of scaffolding - e.g. in a closed cell design - is most important when the plaque is soft, or atheromatous, and a stent with a large free cell area can not sufficiently support this tissue to keep the lumen patent.[13] We propose that in a more calcified plaque support at only a few sites - e.g. an open cell design - may possibly suffice in scaffolding the plaque, but a larger radial force would be necessary to achieve patency in such a hardened plaque.

There are certain limitations to this study. First, we only included five stents of each type in our tests; second, stent placement inside the device was performed manually, leaving room for human error; third, friction between the film loops and the stent might be of influence on the results. However, friction between the vessel wall, plaque and different stent designs will also occur in vivo. Given the comparative purpose of this study, not correcting the results for possible friction was therefore considered acceptable. Considering the use of high quality materials, such as the non-expansive BOPET film and validated copper strain gauges, this study is regarded as easily reproducible, in spite of the above mentioned limitations. Finally, the stenosis model in this study was straight and static. In vivo, carotid stents would experience vessel curvature, vascular smooth muscle actions and a much more dynamic situation. Although this makes extrapolation to a clinical setting difficult, the reported results are clear and undiluted by any of these confounders.

CONCLUSIONS

Radial forces exerted by carotid stents vary significantly between various stent designs. Clinical results of CAS may be dependent of a number of stent specific factors, including the radial force exerted upon the arterial lesion and the carotid wall. Future studies should investigate the relationship between plaque constitution, radial force and patency rates.

REFERENCES

- [1] International Carotid Stenting Study Investigatoris. Carotid artery stenting compared with endarterectomy in patients with symptomatic carotid stenosis (International Carotid Stenting Study): an interim analysis of a randomised controlled trial. Lancet. 2010;375:985-997.
- [2] Brott TG, Hobson RW, Howard G, et al. Stenting versus Endarterectomy for treatment of carotidartery stenosis. N Eng J Med. 2010;363:11-23
- [3] Roubin GS, New G, Iyer SS, et al. Immediate and late clinical outcomes of carotid artery stenting in patients with symptomatic and asymptomatic carotid artery stenosis: a 5-year prospective analysis. Circulation. 2001;103:532-537.
- [4] Hwang KW, Suh SI, Seo WK, et al. Hemodynamic depression during carotid angioplasty with stenting: potential risk factors determined by multidetector computed tomography angiography and related clinical factors. J Comp Assist Tomogr. 2008;32:124-129.
- [5] Taha MM, Toma N, Sakaida H, et al. Periprocedural hemodynamic instability with carotid angioplasty and stenting. Surg Neurol. 2008;70:279-285; discussion 85-86.
- [6] Nano G, Dalainas I, Bianchi P, et al. Ballooning-induced bradycardia during carotid stenting in primary stenosis and restenosis. Neuroradiology. 2006;48:533-536.
- [7] Howell M, Krajcer Z, Dougherty K, et al. Correlation of periprocedural systolic blood pressure changes with neurological events in high-risk carotid stent patients. J Endovasc Ther. 2002;9:810-816.
- [8] Mlekusch W, Schillinger M, Sabeti S, et al. Hypotension and bradycardia after elective carotid stenting: frequency and risk factors. J Endovasc Ther. 2003;10:851-859; discussion 60-61.
- [9] Gupta R, Abou-Chebl A, Bajzer CT, Schumacher HC, Yadav JS. Rate, predictors, and consequences of hemodynamic depression after carotid artery stenting. J Am Coll Cardiol. 2006;47:1538-1543.
- [10] Diehm N, Katzen BT, Dick F, et al. Influence of stent type on hemodynamic depression after carotid artery stent placement. J Vasc Interv Radiol. 2008;19:23-30.
- [11] Kalmar G, Hubner F, Voelker W, et al. Radial force and wall apposition of balloon-expandable vascular stents in eccentric stenoses: an in vitro evaluation in a curved vessel model. J Vasc Interv Radiol. 2002;13:499-508.
- [12] Mueller-Huelsbeck S, Schaefer PJ, Charalambous N, Schaffner SR, Heller M, Jahnke T. Comparison of carotid stents: an in-vitro experiment focusing on stent design. J Endovasc Ther 2009;16:168-177
- [13] Bosiers M, de Donato G, Deloose K, et al. Does free cell area influence the outcome in carotid artery stenting? Eur J Vasc Endovasc Surg. 2007;33:135-141; discussion 42-43.
- [14] Schillinger M, Gschwendtner M, Reimers B, et al. Does carotid stent cell design matter? Stroke. 2008;39:905-909.
- [15] Sugita J, Cremonesi A, Van Elst F, et al. European carotid PROCAR Trial: prospective multicenter trial to evaluate the safety and performance of the ev3 Protege stent in the treatment of carotid artery stenosis--1- and 6-month follow-up. J Interv Cardiol. 2006;19:215-221.
- [16] Cremonesi A, Rubino P, Grattoni C, Scheinert D, Castriota F, Biamino G. Multicenter experience with a new "hybrid" carotid stent. J Endovasc Ther. 2008;15:186-192.
- [17] Duda SH, Wiskirchen J, Tepe G, et al. Physical properties of endovascular stents: an experimental comparison. J Vasc Interv Radiol. 2000;11:645-654.
- [18] Ahlhelm F, Kaufmann R, Ahlhelm D, Ong MF, Roth C, Reith W. Carotid Artery Stenting Using a Novel Self-Expanding Braided Nickel-Titanium Stent: Feasibility and Safety Porcine Trial. Cardiovasc Intervent Radiol. 2009;32:1019-1027
- [19] Isayama H, Nakai Y, Toyokawa Y, et al. Measurement of radial and axial forces of biliary selfexpandable metallic stents. Gastrointest Endosc 2009;70:37-44
- [20] Zegdi R, Lecuyer L, Achouh P, et al. Increased Radial Force Improves Stent Deployment in Tricuspid but Not in Bicuspid Stenotic Native Aortic Valves. Ann Thorac Surg 2010;89:768-772

Blood thinners Statins Beta-blockers Vitamin D status New-onset arrhythmias Carotid stent characteristics

Aortic stentgraft composition

Laparoscopic sac fenestration after EVAR Need we intervene for type II endoleaks?

Published as: **"Stentgraft composition plays a material role in the post-implantation syndrome"** *Journal of Vascular Surgery 2012;56:1503-1509*

Michiel T. Voûte Frederico M. Bastos Gonçalves Koen M. van de Luijtgaarden Casper G.A. Klein Nulent Sanne E. Hoeks Robert Jan Stolker Hence J.M. Verhagen

ABSTRACT

Objective

In patients undergoing EVAR the post-implantation syndrome (PIS), comprising fever and inflammation, occurs frequently. The cause of PIS is unclear, but graft composition and acute thrombus formation may play a role. The objective of this study was to evaluate these possible causes of the inflammatory response after endovascular aneurysm repair (EVAR).

Methods

One-hundred and forty-nine patients undergoing elective EVAR were included. Implanted stentgrafts differed mainly in the type of fabric used; either woven polyester (n=82) or expanded polytetrafluorethylene (ePTFE, n=67). Tympanic temperature and CRP were assessed daily during hospitalization. PIS was defined as the composite of a body temperature of \geq 38 degrees Celcius coinciding with C-reactive protein (CRP) >10 mg/L. Besides graft composition, the size of the grafts and the volume of new-onset thrombus were calculated using dedicated software and results were correlated to PIS.

Results

Implantation of grafts made of polyester was associated with higher postoperative temperature (P<0.001), CRP levels (P<0.001) and incidence of PIS (56.1% versus 17.9%; P<0.001), compared to ePFTE. Following multivariate analysis, woven polyester stentgrafts were independently associated with an increased risk of PIS (Hazard ratio 5.6, 95%CI 1.6-19.4, P=0.007). Demographics, amount of graft material implanted or new-onset thrombus had no association with PIS.

Conclusions

The composition of stentgrafts may play a material role in the incidence of postimplantation syndrome in patients undergoing endovascular aneurysm repair. Implantation of stentgrafts based on woven polyester was independently associated with a stronger inflammatory response.

MANUSCRIPT

INTRODUCTION

In patients undergoing endovascular aneurysm repair (EVAR) for an abdominal aortic aneurysm (AAA), an acute phase inflammatory response may occur shortly after implantation.[1-4] This so-called post-implantation syndrome (PIS) is defined as fever coinciding with a rise in inflammatory markers.[5-9] PIS is thought to be transient and harmless, but its true significance is unknown and no clear guidelines exist for management.]10] Importantly, the cause of the inflammatory response remains unclear. Proposed mechanisms are related to the introduction of the different components of the stentgraft[11,12] or the amount of mural thrombus within the aneurysm.[13]

Initial results of endovascular repair with the woven polyester Talent Abdominal Stent Graft (Medtronic Inc., Minneapolis, MN, USA) showed a high incidence of fever and a systemic inflammatory response as the most common serious complication.[14] Based on clinical experience, PIS seems to be even more frequent since the recent introduction of the woven polyester Endurant Abdominal Stent Graft (Medtronic). It is not uncommon for patients to suffer from fatigue and elevated body temperatures, sometimes for weeks after the procedure.

For optimal management of patients with post-implantation fever and rise in inflammatory markers, and to contribute to future stent-graft design, a better understanding of the cause of PIS is necessary. This retrospective study investigates the role of graft material on the post-implantation syndrome by comparing two types of graft material, woven polyester and ePTFE. Besides the type of material, the quantity of implanted graft and the association of new-onset thrombus with the inflammatory response are investigated.

METHODS

Study population

The study population was derived from a cohort of consecutive patients undergoing an EVAR procedure between 2004 and 2010 at the Erasmus University Medical Center, Rotterdam, The Netherlands. Exclusion criteria were the concurrent use of different graft materials on the same patient, hybrid procedures combining endovascular with open surgical treatment, urgent EVAR and recent surgery or major trauma within 30 days of the procedure. Patients with missing data on temperature or C-reactive protein (CRP) were also excluded, as were patients who suffered a postoperative complication that had an effect on inflammatory markers, including urinary tract infections, pneumonia and haematomas (Figure 1).

Baseline characteristics comprised of gender, age and all traditional cardiac risk factors from the Revised Cardiac Risk (RCR) index[15], as well as the incidence of chronic obstructive pulmonary disease (COPD), smoking and hypercholesterolemia. Additionally, the use of medication with known anti-inflammatory or anti-pyretic effects such as aspirin, statins and beta-blockers, was recorded.



According to hospital protocol, all patients underwent an endovascular procedure with prophylactic antibiotics in the form of 1 gram of cefazolin 30 minutes prior to incision, and 5000 units of heparin prior to introduction of the stent graft deployment system. Additionally, low-molecular weight heparin was administered in all surgical patients during hospital admission (2500 IU dalteparin daily in all patients, 5000 IU daily in those with a body weight over 80 kg). Type of anesthesia was selected at the discretion of the surgical team. The study was conducted according to the guidelines provided by the Institutional Review Board.

Definition of endpoints

The primary endpoint of the study was the occurrence of PIS shortly after EVAR. PIS was defined as fever coinciding with an elevated serum CRP level. Fever was defined as a tympanic temperature of \geq 38.0 degrees Celsius (°C), and the upper level of normal for CRP was 10 mg/L in our institutional laboratory. Tympanic temperature and serum CRP levels were assessed each morning, starting one day prior to EVAR. Subsequent measurements were performed on the day of EVAR, and then daily up to 4 days following implantation. As mentioned, patients suffering from non-graft-related complications associated with inflammation were excluded from the study, including patients with reported postoperative wound infections, pneumonia and infections of the urinary tract.

Endograft composition

To evaluate the role of the graft component of endovascular aortic devices on the occurrence of PIS, enrolled patients were divided into two groups: in the first group the graft composition was expanded polytetrafluoroethylene (ePTFE), in the second group both devices were composed of a woven polyester graft. The first group comprised patients exclusively treated with the low-permeability Excluder AAA Endoprosthesis (W.L. Gore & Associates, Flagstaff, AZ, USA). The second group was composed of patients treated with the Talent Abdominal Stent Graft or the Endurant Abdominal Stent Graft (both Medtronic). All used stentgrafts were bifurcated, modular devices with a nitinol exoskeleton. An important difference between the two groups was that the low-permeability Excluder was available since late 2004, while the majority of woven polyester grafts were Endurant, which was first used in our hospital in 2008.

Other EVAR-related causes of PIS

Additional to the type of graft material, the total quantity of implanted material may be of importance, as it may be possible that a certain type of graft is simply larger or more extensions are used. As a marker for size, the volume inside each graft was measured on contrast-enhanced computed tomographic angiography (CTA) using dedicated postanalysis software with central lumen line reconstructions (3mensio Vascular software, 3mensio Medical Imaging BV, Bilthoven, The Netherlands). Volume measurements were done semi-automatically, according to a standard protocol as described earlier.[16,17]

Finally, the amount of newonset thrombus - filling the excluded aneurysm sac immediately after EVAR - was measured using the same dedicated software by luminal calculating volume prior to EVAR and comparing this with the postoperative volume measurements (Figure 2). The difference between the two measurements represents excluded the sac volume, discarding any chronic mural thrombus already present before the procedure. For both these measurements, CTA prior to and/or after EVAR are a necessity, thus excluding patients without available CTAs from these sub-analyses.



Statistical analysis

All baseline characteristics and medication use were tabulated, as well as temperatures and CRP levels. Continuous variables were presented as means±SD or, in case of a non-Gaussian distribution, as medians and interquartile range [IQR], and compared with Student's T-test or Mann-Whitney U statistics, respectively. Dichotomous variables were presented as counts and percentages, and compared between groups using Pearson's chi-square statistics. The maximum body temperature and CRP level from the first four days after the procedure was compared to the day prior to surgery. The changes in body temperature were compared for the two types of graft material using Student's T-test, and changes in CRP were compared by Mann-Whitney U statistics. To test the association of graft size and new-onset thrombus with the postoperative rise in temperature and CRP, Pearson's and Spearman's correlation coefficients were calculated in the total population and within each group of graft material. Accounting for the historical difference between groups, we calculated the conditional probability of receiving a woven polyester graft based on baseline characteristics using propensity score analysis.

Propensity scores were generated using logistic regression with graft material as the dependent variable. Variables included to generate the propensity scores were all those presented as group descriptive in Table 1, complemented by the type of anaesthesia. To evaluate the association of graft material with PIS, a

propensity adjusted binary logistic regression analysis was performed, further correcting for statin use, type of anesthesia, new-onset thrombus and graft size. Hazard ratios and 95% confidence intervals (CIs) were presented. All statistical tests were 2-sided, and considered statistically significant when the P-value was <.05. All analyses were performed using PASW statistics 17 for Windows (SPSS Inc., Chicago, Illinois).

RESULTS

A total of 8 Talent, 74 Endurant and 67 Excluder stentgrafts were included in the study. This amounts to a study population of 149 patients, divided in woven polyester (n=82) and ePTFE (n=67) grafts. Patients were predominantly male (87.9%) and had a mean age of 72.6 \pm 7.5 years at the time of surgery. In terms of baseline characteristics, traditional cardiac risk factors were equally distributed among groups (Table I). The preoperative AAA diameter was 59.8 \pm 11.5 mm, without significant difference between the two groups.

When compared to the ePTFE group, patients in the woven polyester group had a higher BMI (26.6 ± 4.8 versus 25.1 ± 3.2 , P=0.026) and were more frequently medicated with statins (79.3% versus 64.2%, P=0.040) Also, A higher proportion patients in the woven polyester group received general anesthesia (73.2% versus 35.8%, P<0.001). Besides general anesthesia (n=84), other types of anesthesia used were spinal (n=33), local (n=24) or a combination (n=8).

Table 1. Baseline characteristics, medication use and risk profile by graft material						
Baseline characteristics	Woven polyester (n=82)	ePTFE (n=67)	P-value			
AAA diameter, mean (mm) ± SD	58.4 ± 11.0	61.1 ± 11.9	.22			
Age, mean (years) ± SD	72.8 ± 7.2	72.4 ± 7.9	.75			
Male gender, n (%)	71 (86.6)	60 (89.6)	.58			
Ischemic heart disease, n (%)	32 (39.0)	34 (50.7)	.15			
Diabetes mellitus, n (%)	14 (17.1)	10 (14.9)	.72			
Renal dysfunction, n (%)	15 (18.3)	9 (13.4)	.42			
Cerebrovascular disease, n (%)	7 (8.5)	10 (14.9)	.22			
CHF, n (%)	5 (6.1)	8 (11.9)	.21			
COPD, n (%)	5 (6.1)	19 (28.4)	<.001			
Hypercholesterolemia, n (%)	23 (28.0)	21 (31.3)	.66			
Body mass index, mean $(kg/m^2) \pm SD$	26.6 ± 4.8	25.1 ± 3.2	.026			
Creatinine, mean (umol/L) ± SD	102.1 ± 38.4	97.3 ± 32.2	.42			
Smoking, n (%)	20 (24.4)	26 (38.8)	.058			
Medication use						
Aspirin, n (%)	49 (59.8)	34 (50.7)	.27			
Statin, n (%)	65 (79.3)	43 (64.2)	.040			
Beta-blocker, n (%)	71 (86.6)	59 (88.1)	.79			
ePTFE denotes expanded polytetrafluoroethy	ylene, SD standard deviati	on,				

CHF congestive heart failure and COPD chronic obstructive pulmonary disease

Table 2. Inflammatory markers before and after surgery by graft material						
	Woven polyester (n=82)	ePTFE (n=67)	P-value			
Temperature (°C)						
Prior to surgery, mean ± SD	36.6 ± 0.4	36.7 ± 0.5	.15			
Post-operative, mean ± SD	38.2 ± 0.7	37.6 ± 0.7	<.001			
Difference, mean ± SD	1.6 ± 0.7	0.9 ± 0.8	<.001			
C-reactive protein (mg/L)						
Prior to surgery, median [IQR]	5.0 [2.0 - 8.0]	3.2 [2.0 - 8.0]	.30			
Post-operative, median [IQR]	164.0 [87.0 – 201.0]	49.0 [20.0 – 104.0]	<.001			
Difference, median [IQR]	154.8 [82.6 – 198.5]	38.0 [13.7 – 94.0]	<.001			
Postoperative values presented are the maximum for days 1 to 4 after the procedure						

Graft type and PIS

The mean length of stay was 4.4 days (SD 4.7 days). The incidence of PIS was 46 (56.1%) for the woven polyester group, compared to 12 (17.9%) in the ePTFE group (P<.001). PIS occurred almost exclusively in the first three days after woven polyester implantation and the first two days after ePTFE implantation.

Broken down to individual inflammatory markers, a similar trend was observed. On the morning prior to surgery, temperatures were not significantly different between the woven polyester and ePTFE groups ($36.6\pm0.6^{\circ}$ C versus $36.7\pm0.5^{\circ}$ C, P=.15), nor were CRP levels (5.0 [2.0-8.0] versus 3.2 [2.0-8.0], P=.30) (Table 2).

In the four days after EVAR, both body temperature and CRP levels rose significantly higher in the woven polyester group, compared to the ePTFE group (Figure 3). When calculating the maximum rise in body temperature and CRP compared to baseline for both groups, patients that received polyester woven graft а suffered a higher rise in body temperature (+1.6 versus +0.9 °C, P<.001) and CRP levels (+154.8 versus +38.0 mg/L, P<.001) compared to patients that received an ePTFE graft (Table 2).



Subanalyses of graft size and new-onset thrombus

A total of 72 patients from the woven polyester group (87.8%) and 64 from the ePTFE group (95.5%) had available imaging for in-graft volume measurements. The mean ingraft volume of the implanted grafts was 44.6 ± 13.7 cc. The woven polyester group had a larger in-graft volume than the ePTFE group (50.8 ± 13.3 cc versus 37.7 ± 10.6 cc, P<.001). In general, this marker for graft size showed a statistical correlation to the postoperative rise in temperature (Pearson's rho 0.29, P=.001) and CRP (Spearman's rho 0.26, P=.003). However, after stratifying for type of material, these correlations were no longer significant.

A total of 63 patients from the woven polyester group (76.8%) and 46 from the ePTFE group (68.7%) had available imaging for new-onset thrombus measurements. The mean volume of new-onset thrombus was 51.3 ± 45.8 ml, without significant differences between woven polyester and ePTFE groups (50.8 ± 45.1 ml versus 51.7 ± 46.6 ml, P=.91). Subsequent analyses showed no significant correlation between new-onset thrombus and the rise in temperature (P=.08) or CRP (P=.17).
Multivariable risk model for PIS

As mentioned, several possible confounders of the inflammatory status of the patients were found to be significantly different between groups. Therefore, differences in baseline characteristics were addressed in a propensity score analysis. The association of graft material with PIS was evaluated in a propensity adjusted model, additionally corrected for differences in statin use, graft size and new-onset thrombus. In this analysis the use of woven polyester remained the only significant factor associated with an increased risk of developing post-implantation syndrome (hazard ratio 5.58, 95% confidence interval 1.60-19.42, P=.007)(Figure 4).



DISCUSSION

With the growing application of EVAR and resolution of many significant issues related to this treatment option, particular details and previously overlooked problems are becoming more evident for physicians and warrant further analysis. PIS is a clear entity that affects a significant number of patients, but the mechanisms behind this phenomenon have been scarcely investigated. The current study shows that the graft type plays a primordial role in the development of an acute phase inflammatory response after EVAR. The implantation of stentgrafts that include woven polyester in their composition is associated with significant changes in body temperature and serum CRP, compared to those that are made of ePTFE.

The current results suggest that ePTFE has less pro-inflammatory properties than woven polyester following endovascular implantation in humans. This is supported by an in vitro study by Swartbol et al., comparing the response of human white blood cells in vitro upon incubation with vascular grafts of ePTFE or woven polyester.¹⁸ The authors

found that woven polyester triggered a significantly larger release of pro-inflammatory markers than ePTFE.

The differences in the incidence of fever between patients who receive stentgrafts of woven polyester or ePTFE are previously described in small numbers.[12] Gerasmidis et al. described a total of 22 consecutive EVAR patients that received a woven polyester graft (n=12) or ePTFE (n=10), and compared the incidence of fever and the post-procedural changes in several biomarkers. Although the study lacked the power to observe large difference in laboratory measurements, fever was observed more frequently in the woven polyester group (3/12 patients versus 1/10).

Another suggested origin for PIS is the excluded aneurysm sac filling with new-onset thrombus, a process that may involve various pro-inflammatory cytokines.[13] The current study is the first to address this hypothesis in a quantitative manner, using dedicated software to calculate the excluded volume after EVAR. Data was not available for every patient in this study, and the amount of graft material was estimated using true, post-implantation graft volumes. Although both these limitations may have influenced the results, the measured volumes of excluded sac content in this study showed no correlation to PIS.

There are other differences between stentgrafts, unrelated to graft material, which may also influence the proposed foreign body reaction. All stentgrafts in this study have an exoskeleton made of nitinol, a nickel-titanium alloy. Unlike the others, the Excluder features an additional outer layer of ePTFE covering the alloy, while in the Endurant and Talent the metal and fabric are adjoined by stitches.[19-21] In addition, these two latter feature a bare top stent, further increasing the amount of nitinol directly exposed to the circulation. Apart from quantity, the precise balance between nickel and titanium or even the cutting and polishing may differ between manufacturers, potentially affecting the antigenic properties of the nitinol.

Since the introduction of nitinol for medical application, it has been widely used in coronary and peripheral arterial "bare-metal" stents.[22] No inflammatory response is reported in these applications, despite frequent treatment of multiple and lengthy lesions, requiring large quantities of the material. Furthermore, the chemical production of nitinol prevents breakdown and special coating reduces nickel exposure.[23] It is therefore unlikely that differences in the application of nitinol between stentgrafts may influence the post-implantation syndrome. We cannot, however, completely exclude this factor using our data. The stitching used in Talent and Endurant devices can also not be ruled out as a possible confounder for the inflammatory response after implantation.

The hypothesis that endothelial damage, due to active fixation from the top stent, plays a role may be dismissed by the current study. Evidently, if endothelial aggression due to penetration of foreign material such as hooks or barbs was key, the inflammatory response would be independent of graft type, since both the Excluder and the Edurant have active proximal fixation. A final difference between stentgrafts that could have played a confounding role in the incidence of PIS is the different delivery devices they come with. It is quite conceivable that a different method of graft delivery might affect either the rate of embolism, or the duration of lower extremity ischemia, both of which stimulate an inflammatory response. Although we cannot provide data to refute these arguments, it seems unlikely that the delivery system, with its differences but also with many similarities, could explain the observed difference.

Certain limitations related to our study warrant consideration. Firstly, a proportion of eligible patients had missing information on inflammatory markers, excluding them from this study. Although this is regrettable, such a limitation is inevitable in a retrospective study. To assess if a selection bias occurred, a comparison of baseline characteristics was performed between included patients and excluded patients. Demographics were not significantly different in any parameter, rejecting the possibility of selection bias in that regard. Coincidently, this study was not a randomized trial. The choice of graft was not random, but based on individual parameters. This could have caused a selection bias, although factors that influence the choice of graft generally focus on anatomical suitability, and information on the inflammatory response was not available at the time. Additionally, due to chronological differences between groups, baseline differences were observed in possible co-factors such as smoking, but also the use of statins and general anesthesia. These differences mostly reflect the introduction of guidelines on perioperative treatment and subsequent improvement of risk reduction strategies, since most woven polyester grafts were implanted in recent years.[24] Especially, statin therapy has been known to attenuate perioperative inflammation, if administered prior to surgery. [25-28] However, as statin therapy was more frequent in the woven polyester group it could therefore be expected to have attenuated rather than exaggerated the inflammatory response in this group, compared to ePTFE. To address historical differences and non-randomized nature of the study, we performed a propensity adjusted analysis with the addition of statin use as separate covariate, which identified the use of woven polyester as a independent predictor of PIS, compared to ePTFE. Another limitation is that PIS was only measured during hospital stay. Theoretically, PIS could have occurred in patients that were discharged rapidly (i.e. in the first 2-3 days), but patients are generally discharged when inflammation is decreasing and body temperatures are normal. It is therefore not expected that we missed many PIS cases. Lastly, for the CTA-based subanalyses data was missing, due to patient dependent imaging protocols. However, the available data still represents the largest cohort ever published on the subject, and possible selection bias was not associated with the inflammatory endpoints of the study.

CONCLUSIONS

The type of fabric used in manufacturing endovascular stent grafts may play a material role in the development of post-implantation syndrome, measured by an increase in post-procedural body temperature and serum C-reactive protein. According to our findings, implantation of ePTFE-based endografts results in a less pronounced inflammatory response, in comparison to those based on woven polyester.

REFERENCES

- [1] Blum U, Voshage G, Lammer J, Beyersdorf F, Tollner D, Kretschmer G, et al. Endoluminal stentgrafts for infrarenal abdominal aortic aneurysms. *N Engl J Med*. 1997;336:13-20.
- [2] Galle C, De Maertelaer V, Motte S, Zhou L, Stordeur P, Delville JP, et al. Early inflammatory response after elective abdominal aortic aneurysm repair: a comparison between endovascular procedure and conventional surgery. J Vasc Surg. 2000;32:234-46.
- [3] Chang CK, Chuter TA, Niemann CU, Shlipak MG, Cohen MJ, Reilly LM, et al. Systemic inflammation, coagulopathy, and acute renal insufficiency following endovascular thoracoabdominal aortic aneurysm repair. J Vasc Surg. 2009;49:1140-6.
- [4] Rowlands TE, Homer-Vanniasinkam S. Pro- and anti-inflammatory cytokine release in open versus endovascular repair of abdominal aortic aneurysm. *Br J Surg.* 2001;88:1335-40.
- [5] Velazquez OC, Carpenter JP, Baum RA, Barker CF, Golden M, Criado F, et al. Perigraft air, fever, and leukocytosis after endovascular repair of abdominal aortic aneurysms. *Am J Surg.* 1999;178:185-9.
- [6] Norgren L, Swartbol P. Biological responses to endovascular treatment of abdominal aortic aneurysms. J Endovasc Surg. 1997;4:169-73.
- [7] Akowuah E, Wilde P, Angelini G, Bryan AJ. Systemic inflammatory response after endoluminal stenting of the descending thoracic aorta. *Interact Cardiovasc Thorac Surg*. 2007;6:741-3.
- [8] Kahn RA, Moskowitz DM, Marin M, Hollier L. Anesthetic considerations for endovascular aortic repair. Mt Sinai J Med. 2002;69:57-67.
- [9] Thompson MM, Nasim A, Sayers RD, Thompson J, Smith G, Lunec J, et al. Oxygen free radical and cytokine generation during endovascular and conventional aneurysm repair. *Eur J Vasc Endovasc Surg*. 1996;12:70-5.
- [10] Arnaoutoglou E, Papas N, Milionis H, Kouvelos G, Koulouras V, Matsagkas MI. Post-implantation syndrome after endovascular repair of aortic aneurysms: need for postdischarge surveillance. Interact Cardiovasc Thorac Surg. 2010;11:449-54.
- [11] Videm V, Odegard A, Myhre HO. lohexol-induced neutrophil myeloperoxidase release and activation upon contact with vascular stent-graft material: a mechanism contributing to the postimplantation syndrome? J Endovasc Ther. 2003;10:958-67.
- [12] Gerasimidis T, Sfyroeras G, Trellopoulos G, Skoura L, Papazoglou K, Konstantinidis K, et al. Impact of endograft material on the inflammatory response after elective endovascular abdominal aortic aneurysm repair. Angiology. 2005;56:743-53.
- [13] Swartbol P, Truedsson L, Norgren L. Adverse reactions during endovascular treatment of aortic aneurysms may be triggered by interleukin 6 release from the thrombotic content. J Vasc Surg. 1998;28:664-8.
- [14] Espinosa G, Marchiori E, Silva LF, de Araujo AP, Riguetti C, Baquero RA. Initial results of endovascular repair of abdominal aortic aneurysms with a self-expanding stent-graft. *J Vasc Interv Radiol*. 2002;13:1115-23.
- [15] Lee TH, Marcantonio ER, Mangione CM, Thomas EJ, Polanczyk CA, Cook EF, et al. Derivation and prospective validation of a simple index for prediction of cardiac risk of major noncardiac surgery. *Circulation*. 1999;100:1043-9.
- [16] van Prehn J, van der Wal MB, Vincken K, Bartels LW, Moll FL, van Herwaarden JA. Intra- and interobserver variability of aortic aneurysm volume measurement with fast CTA postprocessing software. J Endovasc Ther. 2008;15:504-10.
- [17] van Keulen JW, van Prehn J, Prokop M, Moll FL, van Herwaarden JA. Potential value of aneurysm sac volume measurements in addition to diameter measurements after endovascular aneurysm repair. J Endovasc Ther. 2009;16:506-13.
- [18] Swartbol P, Truedsson L, Parsson H, Norgren L. Tumor necrosis factor-alpha and interleukin-6 release from white blood cells induced by different graft materials in vitro are affected by pentoxifylline and iloprost. J Biomed Mater Res. 1997;36:400-6.
- [19] Medtronic Inc. Talent Abdominal Stent Graft, product brochure. Available at: http://www.medtronicendovascular.com/system/files/TalentAAA_Brochure.pdf Accessed May 31st 2011
- [20] Medtronic Inc. Endurant Stent Graft System, instructions for use. Available at: http://www.medtronicendovascular.com/system/files/EndurantIFU.pdf Accessed May 31st 2011

- [21] W.L. Gore & Associates. Excluder AAA Endoprosthesis, instructions for use. Available at: http://www.goremedical.com/resources/dam/assets/AH0313-ML4_EN_US.pdf Accessed May 31st 2011
- [22] Dotter CT, Buschmann RW, McKinney MK, Rosch J. Transluminal expandable nitinol coil stent grafting: preliminary report. *Radiology*. 1983;147:259-60.
- [23] Stoeckel D, Pelton A, Duerig T. Self-expanding nitinol stents: material and design considerations. *Eur Radiol*. 2004;14:292-301.
- [24] Task Force for Preoperative Cardiac Risk A, Perioperative Cardiac Management in Non-cardiac S, European Society of C, Poldermans D, Bax JJ, Boersma E, et al. Guidelines for pre-operative cardiac risk assessment and perioperative cardiac management in non-cardiac surgery. *Eur Heart J*. 2009;30:2769-812.
- [25] Dunkelgrun M, Boersma E, Schouten O, Koopman-van Gemert AW, van Poorten F, Bax JJ, et al. Bisoprolol and fluvastatin for the reduction of perioperative cardiac mortality and myocardial infarction in intermediate-risk patients undergoing noncardiovascular surgery: a randomized controlled trial (DECREASE-IV). Ann Surg. 2009;249:921-6.
- [26] Voute MT, Winkel TA, Poldermans D. Optimal medical management around the time of surgery. *Heart*. 2010;96:1842-8.
- [27] Voute MT, Winkel TA, Poldermans D. Safety of fluvastatin in patients undergoing high-risk noncardiac surgery. Expert Opin Drug Saf. 2010;9:793-800.
- [28] Schouten O, Boersma E, Hoeks SE, Benner R, van Urk H, van Sambeek MR, et al. Fluvastatin and perioperative events in patients undergoing vascular surgery. *N Engl J Med*. 2009;361:980-9.

Blood thinners Statins Beta-blockers Vitamin D status New-onset arrhythmias Carotid stent characteristics Aortic stentgraft composition



Laparoscopic sac fenestration after EVAR

Need we intervene for type II endoleaks?

Published as:

"Treatment of post-implantation aneurysm growth by laparoscopic sac fenestration: long-term results"

European Journal of Vascular and Endovascular Surgery 2012;44:40-44

Michiel T. Voûte Frederico M. Bastos Gonçalves Joke M. Hendriks Roderik Metz Marc R.H.M. van Sambeek Bart E. Muhs Hence J.M. Verhagen

ABSTRACT

Objectives:

Sac growth after endovascular aneurysm repair (EVAR) is an important finding, which may influence prognosis. In case of a type II endoleak or endotension, clipping of sidebranches and subsequent sac fenestration has been presented as a therapeutic alternative. The long-term clinical efficacy of this procedure is unknown.

Methods:

The study included eight patients who underwent laparoscopic aortic collateral clipping and sac fenestration for enlarging aneurysms following EVAR. Secondary interventions and clinical outcome were retrieved from hospital records. Sac behavior was evaluated measuring volumes on periodical CTA imaging using dedicated software.

Results:

Follow-up had a median length of 6.6 (range 0.6-8.6) years. During this time, only three patients successfully achieved durable aneurysm shrinkage (n=2) or stability (n=1). The remaining patients suffered persistent (n=2) or recurrent sac growth (n=3), all regarded as failure of fenestration. A total of six additional interventions were performed, comprising open conversion (n=2), relining (n=1) and implantation of iliac extensions (n=3). All additional interventions were successful at arresting further sac growth during the remainder of follow-up.

Conclusions:

Despite being a less invasive alternative to conversion and open repair, the long-term outcome of sac fenestration is unpredictable and additional major procedures were often necessary to arrest sac growth.

MANUSCRIPT

INTRODUCTION

Endovascular aneurysm repair (EVAR) of abdominal aortic aneurysm (AAA) has developed since 1991 [1] and is now frequently the preferred method of treatment. The ultimate goal of EVAR is to prevent death from aneurysm rupture by excluding the aneurysm sac from the circulation, thereby relieving it from pressure. After EVAR, most aneurysms will stabilize or shrink in diameter. Some aneurysms, however, will continue to expand.[2-4]

Continued sac expansion after EVAR can have several explanations, but endoleaks and graft porosity (endotension) are frequently cited as culprits. In the case of sac growth, most physicians propose additional treatment to prevent the aneurysm from rupturing or to prevent aortic dilatation near the proximal or distal sealing zones, giving rise to possible migration and/or type I endoleaks. When an endoleak is associated with growth, a secondary endovascular procedure or conversion to open repair is usually performed. When no endoleak is found, the solution is more challenging, as the cause of continued aneurysm expansion is frequently unclear.

Previously, laparoscopic fenestration of the aneurysm sac was suggested as treatment for patients with an enlarging aneurysm sac after EVAR, with clipping of aortic sac collaterals.[5] Although the early results were promising, long-term durability of this treatment remains unknown. The aim of the current study was to evaluate the longterm effects of this treatment on sac behavior, to provide guidance in future decision making.

METHODS

Patient selection

From June 1999 to October 2005, a total of 143 AAA patients underwent an EVAR procedure in our hospital. During follow-up, sac growth was observed in 34 patients (23.8%). Type II endoleaks were detected in 21 cases (14.7%). These were either observed or treated with percutaneous interventions, such as coil-embolisation, glue injections and endoscopic clipping of lumbar arteries, depending on sac behavior. In case of a growing aneurysm sac where no endoleak was detected or when an endovascular approach of type II endoleak was technically unsuccessful or failed to arrest growth, an alternative approach was proposed. Laparoscopic fenestration of the aneurysm sac was then performed, which was preceded by clipping of patent Inferior Mesenteric Artery (AMI) and lumbar arteries. In order to evaluate the effect of fenestration on sac behavior, all patients who underwent this procedure were included. The sole exclusion criterion for this study was the lack of a minimum two postfenestration imaging studies, as that would make observations on sac behavior impossible. The study was conducted in agreement with the Institutional Medical Ethics Committee guidelines.

Fenestration procedure

The technical details of this intervention were described previously.[5] In summary, all visible lumbar arteries were clipped endoscopically through a retroperitoneal approach, and a patent AMI was clipped laparoscopically. Cleared from all patent side- branches, the aneurysm was then fenestrated. During this phase of the operation, the operators could check for residual back-bleeding and suture any remaining type II endoleaks. Also, the sac contents were removed at this time and an omentum slip was inserted whenever technically possible in the sac to prevent immediate closure of the fenestration, reduce exposure of the bare endograft to the small intestines and possibly facilitate resorption of hygroma in the early stages after fenestration.

In one case, the procedure was converted to open suturing of all patent side-branches and fenestration of the sac. The primary operator during all procedures was the same, experienced vascular surgeon (J.H.), who was assisted by an experienced laparoscopic surgeon.

Efficacy of fenestration

At the time of these procedures, sac growth was a phenomenon that was aggressively treated. Therefore, the preferential outcome of this treatment at the time was to achieve sac stability or shrinkage. Primary endpoint of the current study is therefore persistent or recurrent sac growth, which is considered failure of treatment. Aneurysm related death and additional vascular interventions were recorded as secondary endpoints. Information on survival and the cause of death was retrieved from hospital records.

Analysis of sac behavior

Measurement of the aneurysm sac was performed on computer tomography angiography (CTA) images. The first CTA, within 48h after the fenestration, was considered the baseline for future follow-up. CTAs were then performed approximately every six or twelve months, according to institutional protocol. All hospital records were reviewed for additional interventions and rationale behind treatment decisions. Sac behavior was scored by two complementary methods.[6,7] First, the single largest diameter of the aneurysm sac was measured. Second, the total sac volume was quantified on each CTA and plotted in time-related curves, regarding the first measurement after fenestration as baseline. All measurements were performed on a workstation with dedicated software (3Mensio Vascular v4.2; 3Mensio Medical Imaging B.V., Bilthoven, The Netherlands) and using center-lumen line (CCL) reconstruction. Volume measurements were obtained according to a standardized and previously validated protocol.[8] Sac growth was defined as >5% increase in volume compared to baseline or in a 12 month interval. All data was subsequently analyzed using the Statistical Package for the Social Sciences, version 17.0 (SPSS Inc., Chicago, Illinois, USA).

RESULTS

Study population

In the presented time window, a total of 9 patients with a growing aneurysm after

EVAR underwent aneurysm sac fenestration. One patient died of non-Hodgkin lymphoma three months after the procedure, having received only one CTA after the procedure, and was therefore excluded from the current study. From the remaining 8 patients (7 men) one patient suffered from a common iliac artery aneurysm rather than an AAA, but was similarly treated by EVAR and later fenestration for continued growth. In one patient, the endoscopic procedure was converted to an open fenestration procedure, as described.⁵ At the time of fenestration, the 8 patients had a median age of 65.2 (range 55.1-74.3) years. Patient baseline characteristics are detailed in Table 1. There was no perioperative mortality.

Table 1. Descriptive statistics for the study population						
Baseline characteristics	All patients (n=8)					
Age in years, median (range)	65.2 (55.1-74.3)					
Female gender, n (%)	1 (12.5)					
Ischemic heart disease, n (%)	2 (25)					
Diabetes mellitus, n (%)	1 (12.5)					
History of stroke, n (%)	0					
Congestive heart failure, n (%)	1 (12.5)					
Renal dysfunction, n (%)	1 (12.5)					
Cardiac arrhythmias, n (%)	1 (12.5)					
Hypertension, n (%)	4 (50)					
History of smoking, n (%)	3 (37.5)					
COPD, n (%)	0					
COPD denotes chronic obstructive pulmonary disease						

Procedural details

Four patients were treated with an Excluder AAA Endoprosthesis (W.L. Gore and associates, Flagstaff, AZ, USA), one of which was the low permeability design introduced in 2004 (Table 2). The remaining implanted grafts were three Zenith AAA Endovascular Grafts (Cook Medical, Bloomington, IN, USA) and one Ancure Graft (Guidant, Menlo Park, CA, USA). Fenestration took place at a median of 1.7 (range 0.5-5.8) years after EVAR. Pre-fenestration sac diameters measured on CTA had a median of 73.2mm (range 56.5-91.0mm). The indication for fenestration was persistent or recurrent sac growth for all cases. In 50% a type-II EL could be detected as the possible culprit (Table 2). Attempts to treat these first with glue injections and coil embolizations had been unsuccessful. Upon reviewing the imaging studies in preparation of the procedures, no intense inflammatory component was observed nor was this noticed during the operation. During the procedures, the operators concurred in having achieved proper exposure and the ability to clip all side-branches. As confirmation, in only one case residual back-bleeding was observed upon opening the aneurysm sac, which was sutured from within. An omentum slip to leave in the fenestration was available in 5 out of 8 patients.

ble 2. Details on EVAR follow-up prior to fenestration							
Case	Implanted graft type	Time since EVAR (years)	Sac diameter (mm)	Detected endoleak			
1	Excluder OD	1.6	74.7	none			
2	Excluder OD	0.7	68.0	Type II			
3	Zenith	1.7	83.5	none			
4	Zenith	2.6	69.9	Type II			
5	Ancure	5.8	71.6	Type II			
6	Zenith	2.4	56.5	none			
7	Excluder OD	1.6	84.3	none			
8	Excluder LP	0.5	91.0	Type II			

Aneurysm sac behavior

Follow-up had a median length of 6.6 (range 0.6-8.6) years. During this time, only three patients experienced durable aneurysm sac shrinkage (n=2) or stability (n=1) and were considered a success. In these three cases where sac growth was successfully arrested, two cases suffered progression of disease leading to dilatation of a common iliac artery (Case #2 and #5). Although this prompted the endovascular extension of one of the distal sealing zones (Table 3), this was not regarded as failure of fenestration.

The remaining five cases suffered persistent sac growth (n=2) or recurrent growth after initial shrinkage (n=3), all regarded as failure of fenestration. The two cases with persistent sac growth comprised one patient with a persistent type II endoleak despite clipping and fenestration, who was converted after 6 months (Case #4), and another patient without detectable endoleaks but an original design Excluder in situ (Case #7). This patient was presumed to suffer from endotension, but refused additional treatment until over 5 years after fenestration, when relining of the endograft finally arrested sac growth.

The three cases with recurrent sac growth included one patient that showed shrinkage during the first 7 years, but on the latest CTA suddenly had growth of the aneurysm sac (Case #3) suggesting re-pressurization, and one patient with a persistent type II endoleak who showed shrinkage at first but recurrent growth within 15 months, spurring conversion (Case #8). In the final case, primary indication for EVAR was a combination of a large iliac aneurysm and a small abdominal aortic aneurysm (Case #6). Sac shrinkage was observed in the first two years after fenestration, but eventually volume and diameter increased again until, finally, contrast was observed in the iliac aneurysm sac, resulting in an extension of the distal dealing zone.

Table 3. Details on fenestration follow-up and outcomes									
Case	Baseline volume	Midterm volume	Latest volume	Follow-up (years)	Sac growth	Status endoleak	Additional intervention		
1	152	81	81	8.6	No	n/a	None		
2	97	97	96	8.6	No	Treated	Iliac extension		
3	263	244	289	8.0	Yes	n/a	None		
4	239	239	263	0.6	Yes	Persistent	Conversion		
5	239	164	151	6.3	No	Persistent	Iliac extension		
6	47	37	43	7.0	Yes	n/a	Iliac extension		
7	188	387	254	6.0	Yes	n/a	Relining		
8	432	381	431	1.3	Yes	Persistent	Conversion		
Volum Sac gro	Volumes are abdominal aneurysm sac volumes in ml. Sac growth was defined as >5% volume change compared to baseline or in a 12 month interval.								

No technical aspects of the procedures or observation made during surgery could be identified as playing part in the success rate of fenestrations. As mentioned earlier, no (untreated) back-bleeding was observed during the fenestration that could eventually predispose a patient to a residual or recurrent type II endoleak. Furthermore, the impossibility to mobilize an omentum slip for insertion in the fenestration was no predictor for outcome (arresting growth in two, conversion in one).

In summary, six patients underwent additional interventions after fenestration. Two patients were converted to open repair, both suffering from persisting type II endoleaks and early sac (re-)growth. One patient was relined for persistent sac growth, in the presence of an original design Excluder endoprosthesis. Additionally, three patients underwent implantation of iliac extensions, one of which suffered from recurrent iliac sac growth and the other two from common iliac artery dilatation due to progression of disease. All secondary interventions after fenestration were successful at arresting further sac growth during the remaining duration of follow-up.

DISCUSSION

EVAR has become the preferred method of treatment in many AAA patients, especially when the aortic anatomy is favorable. Despite the early survival advantage, EVAR is associated with greater aneurysm-related complications and therefore most agree on the need for life-long follow-up with imaging studies in order to evaluate migration, stent integrity, endoleaks and aneurysm size.[9-11] Post-implantation growth has received particular attention because it is observed with relative frequency and suggests continued pressurization of the aneurysm sac, and therefore failure of treatment (despite relative rarity in clinical consequences).[12] After EVAR, the majority of patients have either a gradual decline or stabilization of their aneurysm dimensions over the years.[13]

When growth occurs, however, a plausible explanation should be sought and treatment promptly offered. While it may be the accepted standard of care that patients with type I or III endoleaks require rapid intervention, opinions vary over the implications of type II endoleaks, especially in cases where the diameter of the aneurysm stabilizes or only grows slowly.[14,15] Within the last decade, studies report that selective surveillance of a type II endoleak is a safe course.[16] Controversially, Jones et al. reported that persistent type II endoleak increases the risk for rupture and the need for conversion, [17] while data from the EUROSTAR registry suggested that it actually seems to protect the patient against rupture.[14] When the current patients were diagnosed with a growing aneurysm after EVAR, endotension and type II endoleaks were aggressively treated. In 2002, Veith et al. reported on a summit with twenty-seven interested leaders who reached concensus that growing aneurysms without detection of endoleaks should be treated surgically or by repeated EVAR procedure.[17] Concerning type II endoleaks, Steinmetz et al. reported that if no sac growth is seen no additional intervention is necessary.[15] However, general opinion among the leaders previously mentioned was that persistent type II endoleaks required treatment, either with coil/glue embolization[18] or laparoscopic clipping.[19]

With that historical backdrop, a series of 9 patients with growing aneurysms without detectable endoleaks or with persistent type II endoleaks were treated by laparoscopic clipping of side branches and aneurysm sac fenestration. Although the short-term results were promising⁵, the current study is the first to show that long-term results are sub-optimal in a large proportion of patients, raising doubt over the applicability of this previously described technique. The ultimate goal of the clipping and fenestration procedure was to halt sac growth. Durable aneurysm sac stability was only achieved in three patients, two of which underwent additional procedures for progression of the disease in the common iliac arteries. Out of the other five cases, two were converted to open repair, one was relined, one was extended at the distal sealing zone and one was diagnosed with recurrent sac growth on the latest scan. In general, the two-step procedure was not particularly successful in achieving its goal of durable sac stability.

The first step in the procedure was to clip all lumbars and other possible side-branches to treat or prevent type II endoleaks. Noticeably, out of four cases presenting with a type II endoleak prior to fenestration in our study, the endoleak persisted in three, despite the subjectively good view on lumbar arteries during this procedure. The only successful elimination of a type II endoleak was achieved in the one patient that was converted, and therefore clipping of collaterals and sac fenestration was performed as an open procedure, reducing the endoscopic success rate of clipping to nil. Interestingly, an open aneurysm sac with a subsequently demonstrable endoleak had no clinical consequences in our series. Although minimally invasive clipping of lumbar side-branches has been frequently performed, right-sided lumbar arteries are technically difficult to expose and clip.[20] In some cases, endoscopic clipping may be unsuccessful, resulting in residual type II endoleaks.[21] This could have contributed to the failure of arresting type II endoleaks durably, in the current study. An alternative approach is primary fenestration and subsequent sewing of back-bleeding lumbars from within the sac.[22,23]

In the current study, fenestration was performed after clipping of the side-branches, allowing for visual control by scanning for residual back-bleeding, as previously described by Dion et al. in 2001.[24] Only in one case back-bleeding was still observed, and this was sutured from within the sac. Although sac contents were thoroughly evacuated after fenestration, the residual type II endoleaks could have been masked by mural thrombus or other debris, missed at the time of surgery. This illustrates that laparoscopic fenestration is a demanding procedure and, even in the hands of experienced vascular and laparoscopic surgeons can lead to underexposure of the inside of the sac, and thus incomplete removal of thrombus and assessment of back-bleeding side-branches.

The most logical indication for fenestration would therefore be endotension as a result of increased graft porosity. Transudate of fluid through the graft fabric is well described, particularly after implantation of the original Excluder endograft (W.L. Gore and associates, Flagstaff, AZ, USA).[18] Releasing the hygroma would theoretically result in arrested growth and prolonged success. This idea has also been defended by others, both with open[19] or with percutaneous sac fenestration. In our series, two patients implanted with the Original Design Excluder continued to exhibit growth without detectable endoleaks prior to fenestration. After fenestration, sac stability was observed in one, but sac growth persisted in the other, who later underwent successful relining with a low-permeability graft. This sac growth could be explained by healing of the fenestration, resulting in the recurrence of hygroma, allowing re-pressurization. Goodney et al. and Kougias et al. have published on their experience with relining, with similar good results at short-term.[21,21] This alternative solution, although promising, still lacks long-term data, but is generally accepted as first line treatment in case of a growing sac with an original Excluder endograft in situ, or when graft integrity is thought compromised at a specific location. [22,23] Importantly, standard CTA is not the most sensitive technique for type-II endoleak visualization, and definite diagnosis of endotension of often only possible after opening the aneurysm sac and visualizing no bleeding aortic collaterals.[25] Therefore, it is theoretical to reserve this technique for endotension cases.

The current report is limited by its observational design and by the small number of patients. Also, the indication for treatment was individualized and no strict criteria were observed, with potential selection bias. For the purpose of demonstrating the safety and efficacy of the technique, however, these limitations - albeit important - can be accepted in order to prevent others to subject their patients to this ineffective treatment as well.

CONCLUSIONS

In conclusion, the results after fenestration are quite variable and, more importantly, largely unpredictable. Sac growth was observed after fenestration in five out of eight cases, spurring additional interventions in the majority. Therefore, we cannot recommend fenestration as primary treatment for sac growth. Other techniques may hold more promise when minimally invasive interventions fail, risk of rupture is considered high and the patient is too frail for aortic cross-clamping and endograft explantation.

REFERENCES

- [1] Parodi JC, Palmaz JC, Barone HD. Transfemoral intraluminal graft implantation for abdominal aortic aneurysms. *Ann Vasc Surg*. Nov 1991;5(6):491-499.
- [2] Farner MC, Carpenter JP, Baum RA, Fairman RM. Early changes in abdominal aortic aneurysm diameter after endovascular repair. *J Vasc Interv Radiol*. Feb 2003;14(2 Pt 1):205-210.
- [3] Dubenec SR, White GH, Pasenau J, Tzilalis V, Choy E, Erdelez L. Endotension. A review of current views on pathophysiology and treatment. *J Cardiovasc Surg (Torino)*. Aug 2003;44(4):553-557.
- [4] Baum RA, Stavropoulos SW, Fairman RM, Carpenter JP. Endoleaks after endovascular repair of abdominal aortic aneurysms. *J Vasc Interv Radiol*. Sep 2003;14(9 Pt 1):1111-1117.
- [5] van Nes JG, Hendriks JM, Tseng LN, van Dijk LC, van Sambeek MR. Endoscopic aneurysm sac fenestration as a treatment option for growing aneurysms due to type II endoleak or endotension. *J Endovasc Ther.* Aug 2005;12(4):430-434.
- [6] van Keulen JW, van Prehn J, Prokop M, Moll FL, van Herwaarden JA. Potential value of aneurysm sac volume measurements in addition to diameter measurements after endovascular aneurysm repair. J Endovasc Ther. Aug 2009;16(4):506-513.
- [7] Lee JT, Aziz IN, Haukoos JS, Donayre CE, Walot I, Kopchok GE, Lippmann M, White RA. Volume regression of abdominal aortic aneurysms and its relation to successful endoluminal exclusion. J Vasc Surg. Dec 2003;38(6):1254-1263.
- [8] van Prehn J, van der Wal MB, Vincken K, Bartels LW, Moll FL, van Herwaarden JA. Intra- and interobserver variability of aortic aneurysm volume measurement with fast CTA postprocessing software. J Endovasc Ther. Oct 2008;15(5):504-510.
- [9] Lederle FA, Freischlag JA, Kyriakides TC, Padberg FT, Jr., Matsumura JS, Kohler TR, Lin PH, Jean-Claude JM, Cikrit DF, Swanson KM, Peduzzi PN, Open Versus Endovascular Repair Veterans Affairs Cooperative Study G. Outcomes following endovascular vs open repair of abdominal aortic aneurysm: a randomized trial. JAMA. Oct 14 2009;302(14):1535-1542.
- [10] De Bruin JL, Baas AF, Buth J, Prinssen M, Verhoeven EL, Cuypers PW, van Sambeek MR, Balm R, Grobbee DE, Blankensteijn JD, Group DS. Long-term outcome of open or endovascular repair of abdominal aortic aneurysm. N Engl J Med. May 20 2010;362(20):1881-1889.
- [11] United Kingdom ETI, Greenhalgh RM, Brown LC, Powell JT, Thompson SG, Epstein D, Sculpher MJ. Endovascular versus open repair of abdominal aortic aneurysm. N Engl J Med. May 20 2010;362(20):1863-1871.
- [12] Schanzer A, Greenberg RK, Hevelone N, Robinson WP, Eslami MH, Goldberg RJ, Messina L. Predictors of abdominal aortic aneurysm sac enlargement after endovascular repair. *Circulation.* Jun 21 2011;123(24):2848-2855.
- [13] Rhee RY, Eskandari MK, Zajko AB, Makaroun MS. Long-term fate of the aneurysmal sac after endoluminal exclusion of abdominal aortic aneurysms. J Vasc Surg. Oct 2000;32(4):689-696.
- [14] van Marrewijk CJ, Fransen G, Laheij RJ, Harris PL, Buth J, Collaborators E. Is a type II endoleak after EVAR a harbinger of risk? Causes and outcome of open conversion and aneurysm rupture during follow-up. *Eur J Vasc Endovasc Surg.* Feb 2004;27(2):128-137.
- [15] Steinmetz E, Rubin BG, Sanchez LA, Choi ET, Geraghty PJ, Baty J, Thompson RW, Flye MW, Hovsepian DM, Picus D, Sicard GA. Type II endoleak after endovascular abdominal aortic aneurysm repair: a conservative approach with selective intervention is safe and cost-effective. J Vasc Surg. Feb 2004;39(2):306-313.
- [16] Silverberg D, Baril DT, Ellozy SH, Carroccio A, Greyrose SE, Lookstein RA, Marin ML. An 8-year experience with type II endoleaks: natural history suggests selective intervention is a safe approach. J Vasc Surg. Sep 2006;44(3):453-459.
- [17] Veith FJ, Baum RA, Ohki T, Amor M, Adiseshiah M, Blankensteijn JD, Buth J, Chuter TA, Fairman RM, Gilling-Smith G, Harris PL, Hodgson KJ, Hopkinson BR, Ivancev K, Katzen BT, Lawrence-Brown M, Meier GH, Malina M, Makaroun MS, Parodi JC, Richter GM, Rubin GD, Stelter WJ, White GH, White RA, Wisselink W, Zarins CK. Nature and significance of endoleaks and endotension: summary of opinions expressed at an international conference. J Vasc Surg. May 2002;35(5):1029-1035.
- [18] Baum RA, Carpenter JP, Cope C, Golden MA, Velazquez OC, Neschis DG, Mitchell ME, Barker CF, Fairman RM. Aneurysm sac pressure measurements after endovascular repair of abdominal aortic aneurysms. J Vasc Surg. Jan 2001;33(1):32-41.

- [19] Wisselink W, Cuesta MA, Berends FJ, van den Berg FG, Rauwerda JA. Retroperitoneal endoscopic ligation of lumbar and inferior mesenteric arteries as a treatment of persistent endoleak after endoluminal aortic aneurysm repair. J Vasc Surg. Jun 2000;31(6):1240-1244.
- [20] Kolvenbach R, Pinter L, Raghunandan M, Cheshire N, Ramadan H, Dion YM. Laparoscopic remodeling of abdominal aortic aneurysms after endovascular exclusion: a technical description. J Vasc Surg. Dec 2002;36(6):1267-1270.
- [21] Linsen MA, Daniels L, Cuesta MA, Wisselink W. Endoscopic type 2 endoleak repair following endovascular aortic aneurysm repair: acute results and follow-up experience. *Vascular*. May-Jun 2011;19(3):121-125.
- [22] Coggia M, Javerliat I, Di Centa I, Colacchio G, Cerceau P, Kitzis M, Goeau-Brissonniere OA. Total laparoscopic infrarenal aortic aneurysm repair: preliminary results. J Vasc Surg. Sep 2004;40(3):448-454.
- [23] Ferrari M, Sardella SG, Berchiolli R, Adami D, Vignali C, Napoli V, Serino F. Surgical treatment of persistent type 2 endoleaks, with increase of the aneurysm sac: indications and technical notes. *Eur J Vasc Endovasc Surg.* Jan 2005;29(1):43-46.
- [24] Dion YM, Gracia CR, Ben El Kadi HH. Totally laparoscopic abdominal aortic aneurysm repair. *J Vasc Surg*. Jan 2001;33(1):181-185.
- [25] Cornelissen SA, Prokop M, Verhagen HJ, Adriaensen ME, Moll FL, Bartels LW. Detection of occult endoleaks after endovascular treatment of abdominal aortic aneurysm using magnetic resonance imaging with a blood pool contrast agent: preliminary observations. *Invest Radiol.* Sep 2010;45(9):548-553.

Blood thinners Statins Beta-blockers Vitamin D status New-onset arrhythmias Carotid stent characteristics Aortic stentgraft composition Laparoscopic sac fenestration after EVAR

Need we intervene for type II endoleaks?

Published as:

"Intervention for Type II endoleaks? "Primum non nocere": appraisal for the conservative management of low-pressure endoleaks after endovascular aneurysm repair"

Journal of Cardiovascular Surgery (Torino). 2013;54:477-484

Barend M. Mees Michiel T. Voûte Frederico M. Bastos Gonçalves Luis Mota Capitão Hence J.M. Verhagen

ABSTRACT

Endoleaks are the most common problem following endovascular aneurysm repair. The importance of a type II endoleak has often been subject of discussion in scientific literature. Conflicting data on the natural history of type II endoleak have been published. There is no consensus on the threshold for treatment of type II endoleak and controversy exists about the optimal treatment modality. This paper discusses the evidence behind treating type II endoleak and investigates the need for treatment.

MANUSCRIPT

INTRODUCTION

Since first performed in 1987 by Volodos,[1] and first reported by Parodi in 1991,[2] endovascular aneurysm repair (EVAR) has developed into a viable alternative to conventional open surgical repair of abdominal aortic aneurysms (AAA) with advantages regarding perioperative morbidity and mortality.[3,4] Mid- and long-term results have demonstrated the durability of EVAR and equal survival compared to conventional open repair.[5-9] However, long-term follow-up has also revealed a significant amount of endograft-related complications, leading to re-interventions performed after EVAR.[7-9] Endoleaks are a common problem after EVAR; they account for 60% of endograft-related complications and are associated with up to half of all re-interventions.[5,10-12] Endoleak is defined as the persistence of blood flow outside the graft and within the aneurysm sac.[13,14] This persistent flow could prevent aneurysm sac thrombosis, resulting in a potential risk for continuous sac expansion and even rupture.

Endoleaks are categorized in four types according to their cause: type I, as a result of inadequate sealing of the graft proximally and/or distally; type II, by retrograde flow into the aneurysm sac via patent collaterals; type III, due to inadequate sealing of graft joints or graft fabric rupture; and type IV, caused by graft fabric porosity. There is little debate regarding the treatment of type I and III endoleaks, as these signify incomplete exclusion of the aneurysm sac from systemic arterial pressure and therefore ineffective aneurysm repair. In contrast, type II endoleaks are regarded as "low pressure" endoleaks and conflicting data on their natural history and association with adverse long-term outcome have led to controversy on the (need for) treatment of type II endoleaks, which is the focus of this paper.

Type II endoleak

Type II endoleaks are the most common endoleaks after EVAR. The reported incidence of any type II endoleak after EVAR varies from 9% to 29% in literature, depending on type and timing of imaging.[5,10,15-23] Type II endoleaks are related to retrograde flow into the aneurysm sac via collateral aortic branches, most commonly via lumbar arteries or the inferior mesenteric artery (IMA). A variety of demographic and anatomic factors associated with type II endoleaks have been reported. Demographic variables associated with a higher prevalence of type II endoleaks include older age, absence of peripheral arterial disease, and anticoagulation therapy.[21,24,25] Anatomic features frequently associated with type II endoleaks are longer infrarenal neck, larger AAA diameter, lower thrombus load in aneurysm sac, patent IMA, and larger number of patent lumbar arteries.[10,16,19,22,24-26] In contrast, protective factors for development of type II endoleaks are tobacco use, presence of COPD and renal insufficiency.[10,16,20,25]

Natural History

The natural history of type II endoleaks has been subject of research over the last decade. While type I and III endoleaks are grouped together as "high-pressure" leaks (and therefore need prompt treatment), a type II endoleak is generally regarded a "lowpressure" leak. From a physiological point of view, type II endoleaks could not remain patent if only supplied by a single vessel, because end arteries with no outflow inevitably thrombose. Indeed, the majority of type II endoleaks will spontaneously resolve. Approximately half of type II endoleaks, which are present on the completion angiogram of the EVAR, will thrombose spontaneously in the immediate postoperative course.[27,28] Subsequently, the majority of type II endoleaks, diagnosed on the first postoperative CT scan, will seal spontaneously. These type II endoleaks are described as "transient" type II endoleaks." Persistent" type II endoleaks are endoleaks, which remain present after six months post-EVAR and are reported in only 20% to 39% of patients with a previously diagnosed type II endoleak.[10,17,21,26,29] One has to realize that the presence of type II endoleak on imaging during follow-up after EVAR is highly dependent on operator, imaging modality and techniques used, which is subject of another review in this issue.[30] "Persistent" type II endoleaks may still seal spontaneously over time, as two recent Kaplan-Meier analyses reported a projected 35% rate of spontaneous resolution of "persistent" type II endoleak and 75% rate of spontaneous resolution of all type II endoleaks after five years follow-up.[20,22] Increase in aneurysm sac size has been reported in 5% to 50% of patients with a "persistent" type II endoleak.[17,20-22,26,29]

In summary, extrapolating from a conservatively estimated incidence of 20% of type II endoleaks on the first postoperative CT scan, at one year follow-up 4% of EVAR patients will have a "persistent" type II endoleak. Subsequently, the estimated incidence of a "persistent" type II endoleak with increasing sac size is 0.1% to 2% of all EVAR patients.[28]

"Transient" type II endoleaks are not associated with growing aneurysm sac size or other adverse outcomes.[10,31] "Persistent" type II endoleaks, however, have been suggested as potential predictors of adverse events during follow-up. On this topic, conflicting reports have been published. The majority of authors have not reported any major complications, conversion, rupture or death from type II endoleaks, regardless of sac growth.[17,20,29] In contrast, some papers have associated type II endoleaks with sac expansion and late conversion.[21-24] Type II endoleak has also been associated with rupture, but anecdotally.[21] In a recent systematic review of literature, only 14 described cases of rupture from a type II endoleak were found in nine studies. These studies represented a cohort of 2627 type II endoleaks, suggesting an incidence of rupture from a type II endoleak of 0.5% (14 of 2627 patients), which is the same as reported from the data from the EUROSTAR registry (1 in 191 patients).[16,29] This is, of course, assuming that the type II endoleak is the true cause of rupture in all cases, which was not reported undisputedly and may not correspond to reality. Both the EUROSTAR data as a systematic review have confirmed that rupture and open conversion rates were no different in patients with type II endoleak when compared to patients without type II endoleak. [16,28,32] In summary, only a small proportion of type II endoleaks may have a prognostic influence, and current evidence does not support the view that presence of a "persistent" endoleak will alter the clinical course of patients after EVAR if left untreated.

PRE-/ OR INTRAOPERATIVE TYPE II ENDOLEAK MANAGEMENT

Identification of patients at risk for developing type II endoleak has been the stimulus for some to adopt a prophylactic strategy of branch vessel management.[33] After preoperative IMA coil embolization a decrease in rate of type II endoleak from 48% to 17% has been reported.[34] However, others have not been able to duplicate these results.[10] A combination of preoperative IMA coil embolization and intrasac thrombin injection led to a non- statistically significant decrease in incidence from 26% to 14%.[35] Finally, intraoperative sac fibrin glue injection was shown to reduce the incidence of type II endoleak to 2-5%.[36,37] Although these figures seem promising, the patient populations were small, non-randomized and often non-controlled, and had limited follow-up. More importantly, results from these studies do not seem to differ from the natural history of untreated type II endoleaks and suggest that prophylactic embolization exposes the majority of EVAR patients to unnecessary risk, aside from increasing the overall cost and complexity of the procedure.

POSTOPERATIVE TYPE II ENDOLEAK MANAGEMENT

There is currently no consensus on the management of type II endoleak. Some authors have propagated a conservative approach to all type II endoleaks.[29] Others have supported an aggressive approach with intervention in all cases.[38] However, the majority of authors have suggested a selective approach in cases of type II endoleak persistence or in cases of type II endoleak in combination with sac growth.[20] Current clinical practice guidelines recommend a conservative approach for type II endoleak without sac growth and recommend intervention if the type II endoleak is associated with sac growth.[39]

Techniques

The most common technique of postoperative endoleak management is transarterial embolization of branch vessels using coils, glue or thrombin. Transarterial embolization of the IMA or lumbar arteries has been associated with encouraging initial technical success, defined as occlusion of the target artery.[40-42] However, during follow-up this technique has been shown to be less effective; failure and recurrence can occur in up to 80%.[15,41,43] Modification of the technique by embolizing both the feeding and draining arteries as the aneurysm sac itself is thought to be more effective, although comparative studies are missing.[43-45] Transarterial embolization is not without risks, as cases of aortoenteric fistula, transient paraplegia and colonic necrosis have been described.[44,46-48] In a recent review of nine years of experience in treatment of type II endoleak the success rate of transarterial embolization, defined as resolution of endoleak, was only 38%, while 27% of patients required blood transfusion, 10% had cardiac complications, and 14% had infectious complications.[49]

An alternative to the transarterial approach is translumbar embolization of the aneurysm sac.[50,51] A retrospective comparison between the modified transarterial embolization and translumbar embolization demonstrated equally success rates, with a major complication rate of 3.2%.[44] Translumbar embolization can also be performed via a translumbar transcaval or an ultrasound-guided transabdominal approach.[52-54] In a recent study the use of Onyx glue (ev3 Inc, Irvine, California) significantly improved endoleak resolution rates from 38% to 91%.[55] It's important to realize that both coil and glue embolization result in significant scatter on subsequent imaging techniques, and makes further reliable evaluation of the presence of endoleaks difficult or even impossible. (Figure 1)

Alternatively, transcatheter transcaval embolization of the aneurysm has been described, reporting clinical success, defined as absence of type II endoleak after one-year follow-up in 66-83% of patients.[56,57]



Panel A: Translumbar CT-guided embolization of aneurysm sac using Onyx for type II endoleak. Panel B: Follow-up CT-angiogram in the same patient demonstrates significant scattering, which impairs visualization of persistent or newly developed endoleak, as well as sac changes.

Several cases of technically successful laparoscopic retroperitoneal ligation of lumbar arteries or IMA have been reported, including a totally robotic ligation.[58-62] In current practice, laparoscopic ligation is often reserved for failures of endovascular techniques.[49] However, there are very limited data on the effectiveness of laparoscopic ligation. A recent study with long-term follow-up of laparoscopic ligation and sac fenestration reported recurrent aneurysm sac growth in 62% of patients. (Figure 2)[61]

Ultimately, open surgery with endograft removal or sacotomy followed by removal of thrombus from the aneurysm sac and ligation of IMA and lumbar arteries, is a final treatment of type II endoleak, when all above described techniques have failed. This

approach, also known as late conversion, is associated with a considerable morbidity and 10% perioperative mortality.[63]



IS THERE A NEED FOR TREATMENT OF TYPE II ENDOLEAK?

Firstly, the current available literature suggests the natural history of type II endoleak to be benign. The majority of type II endoleaks resolves spontaneously and the majority of persistent type II endoleaks does not lead to aneurysm sac expansion.[11,28,64] Moreover, rupture has not been unequivocally associated with isolated type II endoleak.[10,11,16,17,20,24,28,29,31,49,55,65] Conflicting data on the association between type II endoleak and adverse outcomes after EVAR could partly be explained by the fact that in several studies on natural history and treatment of type II endoleak, a significant proportion of patients, ranging from 21% to 36%, appeared to have a type I or III endoleak.[10,65,66] In these studies adverse outcomes may well have been caused by co-existing high-pressure (type I or III) endoleaks. Also, in the EUROSTAR registry (2463 patients), of all patients without discernible endoleak 7% developed an increase in aneurysm size two years after EVAR.[16] Finally, in a recent landmark review of over 10.000 EVAR patients the overall incidence of any endoleak was 31%. Of these patients 21% developed sac size enlargement, which was only 6% of total patients. However, in the total cohort the overall rate of sac size enlargement after five years follow-up was as high as 41%.[67] Both studies suggest therefore, that the presence of a type II endoleak may actually be a confounder.

Secondly, the results of treatment of type II endoleaks are disappointing. Success rates are inhomogeneously described: often only initial technical success is reported, while just a minority of papers mention type II endoleak resolution (immediate and over time) or, most importantly, freedom from sac growth. The latter, of course, is the only relevant successful outcome measure for the treatment of a type II endoleak. Small series of treatment of type II endoleaks have described inhomogeneous success rates of 20% to 100%.[34,40-42,51,56,68] However, the majority of these reports are anecdotal, non-controlled and non-randomized, have very different definitions for success and lack long-term follow up. More recent and larger series with a substantial follow-up report disappointingly low success rates, ranging from 22% arrest of sac growth to 38% resolution of endoleak, at a cost of a significant amount of reinterventions and risk for complications.[10,49,55,65,69] In a rough estimate using the above success rate of around 25%, the percentage of patients that would benefit (i.e. freedom of sac size growth) from treatment of type II endoleak would only be 25% of 4% (incidence of "persistent" endoleak), which is 1% of all patients after EVAR. Thus, the results of postoperative type II endoleak treatment seem to be no different from the natural history of a type II endoleak.

Finally, a recent meta-analysis has reviewed the available evidence to support any threshold for intervention on type II endoleak.[11] Only studies, which specifically reported sac outcome and specified a threshold for intervention, and with data on more than 100 EVAR, were included.[29,38,45,56,70-74] In total 231 patients were analyzed, 56 were treated at an aggressive threshold, 104 at a selective threshold, and 71 at a conservative threshold. The majority of type II endoleaks (194/231, 84%) demonstrated stable or shrinking sacs during follow-up regardless of treatment strategy and no ruptures were recorded. Meta-regression demonstrated no evidence that an aggressive or selective strategy reduced sac expansion, compared to using a conservative approach.

What to do with a patient with a persistent type II endoleak and aneurysm sac expansion?

In case a type II endoleak does not appear to behave benign (i.e. growing sac size), it should be regarded as a "sentinel" endoleak leading the clinician to an exhaustive search for all possible causes of inadequate exclusion of the aneurysm. Other causes of increasing sac size, such as type I endoleaks, type III endoleaks (including positional or intermittent endoleaks) should be treated or excluded. Focusing our attention on a type II endoleak associated with expansion may divert attention to the true (and occult) cause for growth. Balancing the benefit – risk ratio, a patient with type II endoleak and growing sac size, in absence of a type I or III endoleak and with a confirmed adequate distal and proximal seal of the endograft, may be treated conservatively, as long as closely followed. If either the patient of the physician is uncomfortable with this strategy, the only truly effective and definitive treatment remaining today is conversion to open repair, diminishing all reasons for sac growth after EVAR (of which type II endoleak is only one), at the price of high mortality and morbidity.

CONCLUSIONS

The importance and natural history of a type II endoleak following EVAR has often been subject of discussion in scientific literature. There is neither consensus nor evidence on the threshold for treatment of type II endoleak, but the strategy to reserve treatment for patients with a persistent type II endoleak and growing sac size, which reflects only around 1% of all EVAR patients, seems to have gained the most support between vascular surgeons. Controversy exists about the optimal treatment modality, while outcome is ill defined, effectiveness is disappointing, and (serious) complications resulting from treatment should not be neglected. Consequently, a high demand exists for a systematic review on this subject.

In our opinion, a type II endoleak should be regarded as a "sentinel" endoleak that demands a vigorous search for other (high-pressure) causes of aneurysm sac growth, which consequently require expeditious treatment. However, in absence of high-pressure endoleaks and with adequate seal of the endograft, a patient with isolated type II endoleak and growing sac size may be managed conservatively with close observation (provided informed consent) as rupture is unlikely to occur.

REFERENCES

- [1] Volodos NL, Karpovich IP, Shekhanin VE, Troian VI, Iakovenko LF. [A case of distant transfemoral endoprosthesis of the thoracic artery using a self-fixing synthetic prosthesis in traumatic aneurysm]. Grudn Khir. Nov-Dec 1988(6):84-86.
- [2] Parodi JC, Palmaz JC, Barone HD. Transfemoral intraluminal graft implantation for abdominal aortic aneurysms. *Ann Vasc Surg.* Nov 1991;5(6):491-499.
- [3] Prinssen M, Verhoeven EL, Buth J, et al. A randomized trial comparing conventional and endovascular repair of abdominal aortic aneurysms. N Engl J Med. Oct 14 2004;351(16):1607-1618.
- [4] Greenhalgh RM, Brown LC, Kwong GP, Powell JT, Thompson SG. Comparison of endovascular aneurysm repair with open repair in patients with abdominal aortic aneurysm (EVAR trial 1), 30day operative mortality results: randomised controlled trial. *Lancet.* Sep 4-10 2004;364(9437):843-848
- [5] Endovascular aneurysm repair versus open repair in patients with abdominal aortic aneurysm (EVAR trial 1): randomised controlled trial. *Lancet.* Jun 25-Jul 1 2005;365(9478):2179-2186.
- [6] Blankensteijn JD, de Jong SE, Prinssen M, et al. Two-year outcomes after conventional or endovascular repair of abdominal aortic aneurysms. N Engl J Med. Jun 9 2005;352(23):2398-2405.
- [7] Greenhalgh RM, Brown LC, Powell JT, Thompson SG, Epstein D, Sculpher MJ. Endovascular versus open repair of abdominal aortic aneurysm. N Engl J Med. May 20 2010;362(20):1863-1871.
- [8] De Bruin JL, Baas AF, Buth J, et al. Long-term outcome of open or endovascular repair of abdominal aortic aneurysm. N Engl J Med. May 20 2010;362(20):1881-1889.
- [9] Lederle FA, Freischlag JA, Kyriakides TC, et al. Outcomes following endovascular vs open repair of abdominal aortic aneurysm: a randomized trial. *JAMA*. Oct 14 2009;302(14):1535-1542.
- [10] El Batti S, Cochennec F, Roudot-Thoraval F, Becquemin JP. Type II endoleaks after endovascular repair of abdominal aortic aneurysm are not always a benign condition. J Vasc Surg. May 2013;57(5):1291-1297.
- [11] Karthikesalingam A, Thrumurthy SG, Jackson D, et al. Current evidence is insufficient to define an optimal threshold for intervention in isolated type II endoleak after endovascular aneurysm repair. *J Endovasc Ther.* Apr 2012;19(2):200-208.
- [12] Nordon IM, Karthikesalingam A, Hinchliffe RJ, Holt PJ, Loftus IM, Thompson MM. Secondary interventions following endovascular aneurysm repair (EVAR) and the enduring value of graft surveillance. *Eur J Vasc Endovasc Surg*. May 2010;39(5):547-554.
- [13] White GH, Yu W, May J, Chaufour X, Stephen MS. Endoleak as a complication of endoluminal grafting of abdominal aortic aneurysms: classification, incidence, diagnosis, and management. J Endovasc Surg. May 1997;4(2):152-168.
- [14] White GH, Yu W, May J. Endoleak--a proposed new terminology to describe incomplete aneurysm exclusion by an endoluminal graft. *J Endovasc Surg.* Feb 1996;3(1):124-125.
- [15] Chuter TA, Faruqi RM, Sawhney R, et al. Endoleak after endovascular repair of abdominal aortic aneurysm. J Vasc Surg. Jul 2001;34(1):98-105.
- [16] van Marrewijk C, Buth J, Harris PL, Norgren L, Nevelsteen A, Wyatt MG. Significance of endoleaks after endovascular repair of abdominal aortic aneurysms: The EUROSTAR experience. J Vasc Surg. Mar 2002;35(3):461-473.
- [17] Steinmetz E, Rubin BG, Sanchez LA, et al. Type II endoleak after endovascular abdominal aortic aneurysm repair: a conservative approach with selective intervention is safe and cost-effective. J Vasc Surg. Feb 2004;39(2):306-313.
- [18] Drury D, Michaels JA, Jones L, Ayiku L. Systematic review of recent evidence for the safety and efficacy of elective endovascular repair in the management of infrarenal abdominal aortic aneurysm. *Br J Surg.* Aug 2005;92(8):937-946.
- [19] Sheehan MK, Ouriel K, Greenberg R, et al. Are type II endoleaks after endovascular aneurysm repair endograft dependent? *J Vasc Surg*. Apr 2006;43(4):657-661.
- [20] Silverberg D, Baril DT, Ellozy SH, et al. An 8-year experience with type II endoleaks: natural history suggests selective intervention is a safe approach. J Vasc Surg. Sep 2006;44(3):453-459.
- [21] Jones JE, Atkins MD, Brewster DC, et al. Persistent type 2 endoleak after endovascular repair of abdominal aortic aneurysm is associated with adverse late outcomes. *J Vasc Surg.* Jul 2007;46(1):1-8.

- [22] Abularrage CJ, Crawford RS, Conrad MF, et al. Preoperative variables predict persistent type 2 endoleak after endovascular aneurysm repair. *J Vasc Surg.* Jul 2010;52(1):19-24.
- [23] Timaran CH, Ohki T, Rhee SJ, et al. Predicting aneurysm enlargement in patients with persistent type II endoleaks. *J Vasc Surg.* Jun 2004;39(6):1157-1162.
- [24] van Marrewijk CJ, Fransen G, Laheij RJ, Harris PL, Buth J. Is a type II endoleak after EVAR a harbinger of risk? Causes and outcome of open conversion and aneurysm rupture during followup. Eur J Vasc Endovasc Surg. Feb 2004;27(2):128-137.
- [25] Jonker FH, Aruny J, Muhs BE. Management of type II endoleaks: preoperative versus postoperative versus expectant management. *Semin Vasc Surg.* Sep 2009;22(3):165-171.
- [26] AbuRahma AF, Mousa AY, Campbell JE, et al. The relationship of preoperative thrombus load and location to the development of type II endoleak and sac regression. J Vasc Surg. Jun 2011;53(6):1534-1541.
- [27] Buth J, Laheij RJ. Early complications and endoleaks after endovascular abdominal aortic aneurysm repair: report of a multicenter study. *J Vasc Surg.* Jan 2000;31(1 Pt 1):134-146.
- [28] Gelfand DV, White GH, Wilson SE. Clinical significance of type II endoleak after endovascular repair of abdominal aortic aneurysm. Ann Vasc Surg. Jan 2006;20(1):69-74.
- [29] Rayt HS, Sandford RM, Salem M, Bown MJ, London NJ, Sayers RD. Conservative management of type 2 endoleaks is not associated with increased risk of aneurysm rupture. *Eur J Vasc Endovasc Surg.* Dec 2009;38(6):718-723.
- [30] Cornelissen SA, Prokop M, Verhagen HJ, Adriaensen ME, Moll FL, Bartels LW. Detection of occult endoleaks after endovascular treatment of abdominal aortic aneurysm using magnetic resonance imaging with a blood pool contrast agent: preliminary observations. *Invest Radiol.* Sep 2010;45(9):548-553.
- [31] Nolz R, Teufelsbauer H, Asenbaum U, et al. Type II endoleaks after endovascular repair of abdominal aortic aneurysms: fate of the aneurysm sac and neck changes during long-term followup. J Endovasc Ther. Apr 2012;19(2):193-199.
- [32] Buth J, Harris PL, van Marrewijk C, Fransen G. The significance and management of different types of endoleaks. *Semin Vasc Surg.* Jun 2003;16(2):95-102.
- [33] Bonvini R, Alerci M, Antonucci F, et al. Preoperative embolization of collateral side branches: a valid means to reduce type II endoleaks after endovascular AAA repair. J Endovasc Ther. Apr 2003;10(2):227-232.
- [34] Axelrod DJ, Lookstein RA, Guller J, et al. Inferior mesenteric artery embolization before endovascular aneurysm repair: technique and initial results. J Vasc Interv Radiol. Nov 2004;15(11):1263-126
- [35] Muthu C, Maani J, Plank LD, Holden A, Hill A. Strategies to reduce the rate of type II endoleaks: routine intraoperative embolization of the inferior mesenteric artery and thrombin injection into the aneurysm sac. J Endovasc Ther. Oct 2007;14(5):661-668.
- [36] Zanchetta M, Faresin F, Pedon L, Ronsivalle S. Intraoperative intrasac thrombin injection to prevent type II endoleak after endovascular abdominal aortic aneurysm repair. J Endovasc Ther. Apr 2007;14(2):176-183.
- [37] Pilon F, Tosato F, Danieli D, Campanile F, Zaramella M, Milite D. Intrasac fibrin glue injection after platinum coils placement: the efficacy of a simple intraoperative procedure in preventing type II endoleak after endovascular aneurysm repair. *Interact Cardiovasc Thorac Surg.* Jul 2010;11(1):78-82.
- [38] Solis MM, Ayerdi J, Babcock GA, et al. Mechanism of failure in the treatment of type II endoleak with percutaneous coil embolization. *J Vasc Surg*. Sep 2002;36(3):485-491.
- [39] Moll FL, Powell JT, Fraedrich G, et al. Management of abdominal aortic aneurysms clinical practice guidelines of the European society for vascular surgery. *Eur J Vasc Endovasc Surg.* Jan 2011;41 Suppl 1:S1-S58.
- [40] Baum RA, Carpenter JP, Tuite CM, et al. Diagnosis and treatment of inferior mesenteric arterial endoleaks after endovascular repair of abdominal aortic aneurysms. *Radiology.* May 2000;215(2):409-413.
- [41] Gorich J, Rilinger N, Sokiranski R, et al. Embolization of type II endoleaks fed by the inferior mesenteric artery: using the superior mesenteric artery approach. *J Endovasc Ther.* Aug 2000;7(4):297-301.

- [42] Haulon S, Willoteaux S, Koussa M, Gaxotte V, Beregi JP, Warembourg H. Diagnosis and treatment of type II endoleak after stent placement for exclusion of an abdominal aortic aneurysm. *Ann Vasc Surg.* Mar 2001;15(2):148-154.
- [43] Baum RA, Carpenter JP, Golden MA, et al. Treatment of type 2 endoleaks after endovascular repair of abdominal aortic aneurysms: comparison of transarterial and translumbar techniques. J Vasc Surg. Jan 2002;35(1):23-29.
- [44] Stavropoulos SW, Park J, Fairman R, Carpenter J. Type 2 endoleak embolization comparison: translumbar embolization versus modified transarterial embolization. *J Vasc Interv Radiol.* Oct 2009;20(10):1299-1302.
- [45] Kasirajan K, Matteson B, Marek JM, Langsfeld M. Technique and results of transfemoral superselective coil embolization of type II lumbar endoleak. *J Vasc Surg.* Jul 2003;38(1):61-66.
- [46] Bush RL, Lin PH, Ronson RS, Conklin BS, Martin LG, Lumsden AB. Colonic necrosis subsequent to catheter-directed thrombin embolization of the inferior mesenteric artery via the superior mesenteric artery: a complication in the management of a type II endoleak. J Vasc Surg. Dec 2001;34(6):1119-1122.
- [47] Forester ND, Parry D, Kessel D, Robertson I, Patel J, Scott DJ. Ischaemic sciatic neuropathy: an important complication of embolisation of a type II endoleak. *Eur J Vasc Endovasc Surg.* Nov 2002;24(5):462-463.
- [48] Bertges DJ, Villella ER, Makaroun MS. Aortoenteric fistula due to endoleak coil embolization after endovascular AAA repair. *J Endovasc Ther*. Feb 2003;10(1):130-135.
- [49] Gallagher KA, Ravin RA, Meltzer AJ, et al. Midterm outcomes after treatment of type II endoleaks associated with aneurysm sac expansion. *J Endovasc Ther*. Apr 2012;19(2):182-192.
- [50] Gorlitzer M, Mertikian G, Trnka H, et al. Translumbar treatment of type II endoleaks after endovascular repair of abdominal aortic aneurysm. *Interact Cardiovasc Thorac Surg.* Oct 2008;7(5):781-784.
- [51] Baum RA, Cope C, Fairman RM, Carpenter JP. Translumbar embolization of type 2 endoleaks after endovascular repair of abdominal aortic aneurysms. J Vasc Interv Radiol. Jan 2001;12(1):111-116.
- [52] Stavropoulos SW, Carpenter JP, Fairman RM, Golden MA, Baum RA. Inferior vena cava traversal for translumbar endoleak embolization after endovascular abdominal aortic aneurysm repair. J Vasc Interv Radiol. Sep 2003;14(9 Pt 1):1191-1194.
- [53] Ellis PK, Kennedy PT, Collins AJ, Blair PH. The use of direct thrombin injection to treat a type II endoleak following endovascular repair of abdominal aortic aneurysm. *Cardiovasc Intervent Radiol.* Sep-Oct 2003;26(5):482-484.
- [54] Kasthuri RS, Stivaros SM, Gavan D. Percutaneous ultrasound-guided thrombin injection for endoleaks: an alternative. *Cardiovasc Intervent Radiol.* Jan-Feb 2005;28(1):110-112.
- [55] Abularrage CJ, Patel VI, Conrad MF, Schneider EB, Cambria RP, Kwolek CJ. Improved results using Onyx glue for the treatment of persistent type 2 endoleak after endovascular aneurysm repair. J Vasc Surg. Sep 2012;56(3):630-636.
- [56] Mansueto G, Cenzi D, Scuro A, et al. Treatment of type II endoleak with a transcatheter transcaval approach: results at 1-year follow-up. J Vasc Surg. Jun 2007;45(6):1120-1127.
- [57] Scali ST, Vlada A, Chang CK, Beck AW. Transcaval embolization as an alternative technique for the treatment of type II endoleak after endovascular aortic aneurysm repair. J Vasc Surg. Mar 2013;57(3):869-874.
- [58] Wisselink W, Cuesta MA, Berends FJ, van den Berg FG, Rauwerda JA. Retroperitoneal endoscopic ligation of lumbar and inferior mesenteric arteries as a treatment of persistent endoleak after endoluminal aortic aneurysm repair. J Vasc Surg. Jun 2000;31(6):1240-1244.
- [59] Feezor RJ, Nelson PR, Lee WA, Zingarelli W, Cendan JC. Laparoscopic repair of a type II endoleak. J Laparoendosc Adv Surg Tech A. Jun 2006;16(3):267-270.
- [60] Lin JC, Eun D, Shrivastava A, Shepard AD, Reddy DJ. Total robotic ligation of inferior mesenteric artery for type II endoleak after endovascular aneurysm repair. Ann Vasc Surg. Mar 2009;23(2):255 e219-221.
- [61] Voute MT, Bastos Goncalves FM, Hendriks JM, et al. Treatment of post-implantation aneurysm growth by laparoscopic sac fenestration: long-term results. *Eur J Vasc Endovasc Surg.* Jul 2012;44(1):40-44.

- [62] Ferrari M, Sardella SG, Berchiolli R, et al. Surgical treatment of persistent type 2 endoleaks, with increase of the aneurysm sac: indications and technical notes. *Eur J Vasc Endovasc Surg.* Jan 2005;29(1):43-46.
- [63] Moulakakis KG, Dalainas I, Mylonas S, Giannakopoulos TG, Avgerinos ED, Liapis CD. Conversion to open repair after endografting for abdominal aortic aneurysm: a review of causes, incidence, results, and surgical techniques of reconstruction. J Endovasc Ther. Dec 2010;17(6):694-702.
- [64] Holt PJ, Karthikesalingam A, Patterson BO, et al. Aortic rupture and sac expansion after endovascular repair of abdominal aortic aneurysm. *Br J Surg.* Dec 2012;99(12):1657-1664.
- [65] Aziz A, Menias CO, Sanchez LA, et al. Outcomes of percutaneous endovascular intervention for type II endoleak with aneurysm expansion. *J Vasc Surg.* May 2012;55(5):1263-1267.
- [66] Funaki B, Birouti N, Zangan SM, et al. Evaluation and treatment of suspected type II endoleaks in patients with enlarging abdominal aortic aneurysms. J Vasc Interv Radiol. Jul 2012;23(7):866-872; quiz 872.
- [67] Schanzer A, Greenberg RK, Hevelone N, et al. Predictors of abdominal aortic aneurysm sac enlargement after endovascular repair. *Circulation.* Jun 21 2011;123(24):2848-2855.
- [68] Parry DJ, Kessel DO, Robertson I, et al. Type II endoleaks: predictable, preventable, and sometimes treatable? *J Vasc Surg*. Jul 2002;36(1):105-110.
- [69] Conrad MF, Adams AB, Guest JM, et al. Secondary intervention after endovascular abdominal aortic aneurysm repair. *Ann Surg.* Sep 2009;250(3):383-389.
- [70] Tuerff SN, Rockman CB, Lamparello PJ, et al. Are type II (branch vessel) endoleaks really benign? Ann Vasc Surg. Jan 2002;16(1):50-54.
- [71] Sampram ES, Karafa MT, Mascha EJ, et al. Nature, frequency, and predictors of secondary procedures after endovascular repair of abdominal aortic aneurysm. J Vasc Surg. May 2003;37(5):930-937.
- [72] Beeman BR, Murtha K, Doerr K, McAfee-Bennett S, Dougherty MJ, Calligaro KD. Duplex ultrasound factors predicting persistent type II endoleak and increasing AAA sac diameter after EVAR. J Vasc Surg. Nov 2010;52(5):1147-1152.
- [73] Liewald F, Ermis C, Gorich J, Halter G, Scharrer-Pamler R, Sunder-Plassmann L. Influence of treatment of type II leaks on the aneurysm surface area. *Eur J Vasc Endovasc Surg.* Apr 2001;21(4):339-343.
- [74] Nevala T, Biancari F, Manninen H, et al. Type II endoleak after endovascular repair of abdominal aortic aneurysm: effectiveness of embolization. *Cardiovasc Intervent Radiol*. Apr 2010;33(2):278-284.

Appendices



SUMMARY AND CONCLUSIONS

Medical considerations in vascular surgery

For optimal patient outcome after surgery, it is critical to accurately assess the preoperative cardiac risk of the individual patient. Undergoing surgery predisposes a patient to an increased risk of cardiac adverse events, caused by a variety of factors. One of these is the occurrence of thrombotic complications in patients at a higher risk of thrombosis, for instance after a myocardial infarction, stroke, coronary stent implantation or carotid surgery. The most important issue addressed in **Chapter 1** is the individual approach required to prepare a patient with an increased risk of thrombotic complications for surgery. Surgeons tend to discontinue aspirin and, especially, stronger antiplatelet agents, due to a fear of bleeding complications during surgery, thereby denying patients the optimal protection from thrombotic complications. Careful deliberation between a cardiologist and a surgeon should result in the best possible timing and preparation for surgery in patients receiving antiplatelet therapy. Furthermore, a timeline for discontinuation of anticoagulant agents and initiation of bridging therapy with low weight molecular heparin is provided, based on recent literature.

Additional to reducing the risk of thrombotic complications, several other pathways to perioperative myocardial damage are targeted by a wide variety of drugs. Some of these have effectively reduced the incidence of perioperative myocardial ischemia and cardiac death. For example, statin therapy is proven to be beneficial to patients, undergoing high-risk surgery. **Chapter 2** is a literature review on the safety of fluvastatin, a low-density lipoprotein cholesterol lowering drug, if prescribed in high-risk surgery patients. As reported, fluvastatin use is safe in patients undergoing high-risk surgery and, in fact, reduces the risk of perioperative myocardial ischemia. This protective effect is partly due to the non-lipid lowering effects, or pleiotropic effects, of statin therapy. These include a reduction of inflammatory markers, thereby promoting coronary plaque stability, thus reducing the risk of plaque rupture and myocardial malperfusion. Initiating statin therapy prior to high-risk surgery seems warranted, and statin discontinuation is discouraged. Therefore, statins with an extended release are preferred, to bridge the postoperative phase during which patients cannot take oral medication.

Complementary to plaque stability, attenuation of a surgery-induced stress response is important, to reduce the risk of a perioperative mismatch between myocardial oxygen demand and supply. This is best acquired by beta-blocker therapy, as frequently reported in literature. Historically there has been a reluctance to use β -blockers in a perioperative setting, especially in patients with peripheral arterial disease, because of the hypothetical side effects and worsening of outcome. In the last decade it has become apparent that β -blockers reduce the risk of perioperative myocardial ischemia, myocardial infarction and cardiac mortality, as discussed in **Chapter 3**. When started at

least one week prior to surgery, given in a low dose and titrated to heart rate, the benefits of β -blocker therapy still remain and side effects can be kept to a minimum.

In a continuous effort to improve cardiovascular outcome in (vascular surgery) patients, research is often directed at new risk markers, biomarkers or unexpected risk factors. This could improve risk stratification and lead to new targets for (secondary) prevention. In this light, serum vitamin D levels are gaining interest. In addition to its well-known effects on calcium homeostasis and bone metabolism, there is accumulating evidence that vitamin D deficiency has important extra-skeletal effects, related to the cardiovascular system. Clinical data suggest that vitamin D deficiency promotes atherosclerosis. However, it is not known whether this is a direct effect of vitamin D on the arterial wall, or the result of a vitamin D deficiency-associated increase in established cardiovascular risk factors, such as hypertension, obesity and diabetes. In **Chapter 4** we assessed the vitamin D status in a large population of patients with occlusive or aneurysmatic arterial disease, and related this to clinical cardiovascular risk factors as well as to direct markers for the severity of arterial disease. We demonstrate that low vitamin D status is an indicator for the severity of arterial disease, independent of traditional cardiovascular risk factors and irrespective of the type of vascular disease, i.e. occlusive or aneurysmatic disease.

Although there is still much debate about the requirement levels of vitamin D to this purpose, it might be hypothesized that primary and secondary preventive strategies to reduce vascular disease should focus on vitamin D status, in addition to blood pressure reduction, lipid and glucose control, weight loss, and lifestyle changes. Future studies could test the effects of vitamin D suppletion on cardiovascular disease and perhaps further elucidate the biology of vitamin D in relation to the arterial wall.

A final major cause for morbidity and mortality after vascular surgery is the occurrence of cardiac arrhythmias. The most frequent perioperative arrhythmia is new-onset atrial fibrillation (AF), a known major risk factor for postoperative stroke, myocardial infarction and pulmonary embolism. Some of these complications may be preventable as AF can be treated if diagnosed. Current standard Holter recording or periodical electrocardiography cannot always identify these (paroxysmal) arrhythmias. As printed in **Chapter 5**, the true incidence of paroxysmal AF after major vascular surgery seems to be much higher (over 25%) than estimated with standard practice Holter monitoring, leaving most cases currently undetected and untreated. An insertable cardiac monitor, evaluating rhythm disturbances continuously for weeks or months after surgery, detects patients at risk for thromboembolic complications reliably, opening a new treatment window for these patients. It is our recommendation that future research is aimed at developing a treatment strategy for this specific patient population.
Technical considerations in vascular surgery

Recent and well-embedded technical advancements in vascular surgery have offered a wide variety of therapeutic options in the individual patient. The introduction of endovascular techniques have expanded the arsenal of treatment modalities for the vascular surgeon. While some of these developments seemed instantly applicable and brilliant at the time, some lessons are learned long after the first patients were treated with these new techniques. The development of the carotid artery stent, for instance, facilitated treatment of carotid artery stenosis in patients with difficult anatomy or severe comorbidities. A multitude of manufacturers launched several carotid stents onto the market, all with different properties. Proper stent selection however, can only be preceded by objective comparison of these properties, which has not been performed on many occasions. In Chapter 6 an effort is made to compare an underestimated, but critical feature of self-expandable carotid stents: the radial force. The results show that there are significant differences in radial force between different stent types, and that the length of the lesion plays an important role in this. Clinical success of carotid artery stenting may be improved if the proper stent is selected in each individual patient, and radial force is an important property to take into account. In future studies, the influence of radial force on patency rates in different types of lesions should be investigated in the clinical setting.

Another major technical development in vascular surgery is the introduction of endovascular stent grafts to treat aortic aneurysms. Endovascular aneurysm repair (EVAR) has become widely adapted, and this too caused the introduction of a variety of stentgrafts. A major feature in the composition of different devices is the graft material, which most commonly is either woven polyester of a polymer known as ePTFE. As the use of stentgrafts increased, so did the number of reports of flu-like symptoms in patients after implantation. In **Chapter 7** the incidence of the post-implantation syndrome is reported in over one hundred and fifty patients treated by EVAR. About one third of patients developed post-implantation syndrome, comprising fever and inflammation, and a strong association with woven polyester was observed even after multivariable adjustment. Although long-term consequences of post-implantation syndrome remain uncertain, these results may influence the development of future devices.

Another aspect of EVAR is the management of side-branches. With open repair most side branches, such as lumbar arteries, are ligated. In EVAR they are left untreated, allowing for possible back-bleeding or so-called type II endoleaks. A controversial intervention in the case of sac re-growth after EVAR is endoscopic clipping of side-branches and fenestration of the growing aneurysm sac. In **Chapter 8**, long-term follow-up of eight patients that underwent such a procedure is presented. Outcome was largely variable and unpredictable. In some cases, endoleaks persisted or recurred and in some cases recurrent sac growth led to later open surgical repair. Therefore, this procedure does not seem recommendable in the treatment sac growth after endovascular repair.

Ours was not the only endeavour to come up with the optimal treatment strategy for type II endoleaks. Despite many publication with often conflicting data on the natural history of type II endoleak, there is no consensus on the threshold for treatment of type II endoleak and controversy exists about the optimal treatment modality. In the literature, reviewed in **Chapter 9**, the strategy to preserve treatment for patients with a persistent type II endoleak and also a growing aneurysm sac size, seems to be the most supported among vascular surgeons. We make the argument that sac growth should instigate a vigorous search for a more high-pressure endoleak than just a type II endoleak, often resulting in an expeditious treatment of the true cause for sac growth. In absence of such a cause, the surgeon and the patient could reside in a conservative management with close observation, as sac rupture is unlikely to occur.

NEDERLANDSE SAMENVATTING

Medische overwegingen in de vaatchirurgie

Voor een optimale uitkomst na chirurgie is het van essentieel belang om nauwkeurig het preoperatieve cardiale risico in de individuele patiënt in te schatten. Het ondergaan van een operatie stelt de patiënt bloot aan een verhoogd risico op schade aan het hart, veroorzaakt door verschillende factoren.

Een daarvan is het optreden van trombotische complicaties in patiënten met een verhoogd risico op trombose, bijvoorbeeld na een doorgemaakt hartinfarct, een beroerte, een stentplaatsing in een kransslagader of een operatie van de halsslagader. Het belangrijkste van wat behandeld wordt in **Hoofdstuk 1** is de individuele benadering die nodig is om een patiënt met een verhoogd tromboserisico voor te bereiden op chirurgie. Chirurgen hebben de gewoonte om aspirine en, vooral, sterkere plaatjesremmers te staken, uit angst voor bloedingscomplicaties tijdens de operatie, waarmee de patiënt de optimale bescherming van trombotische complicaties wordt ontzegd. Zorgvuldig overleg tussen een cardioloog en de chirurg zou moeten resulteren in de best mogelijke timing en voorbereiding op de operatie in patiënten die plaatjesremmers slikken. Daarnaast wordt een tijdslijn voor het tijdelijk staken van bloedverdunners en het starten van overbruggingstherapie met laaggewicht moleculaire heparine verschaft, gebaseerd op recente literatuur.

Naast het reduceren van trombotische complicaties worden verschillende andere oorzaken van peroperatieve schade van de hartspier behandeld met een breed scala aan medicatie. Sommigen daarvan hebben succesvol de incidentie van peroperatieve hartschade en hartdood verminderd. Zo is bijvoorbeeld het gebruik van cholesterolremmers, statines genaamd, gunstig gebleken voor patiënten die een hoogrisico operatie ondergaan. Hoofdstuk 2 is een literatuurstudie naar de veiligheid van fluvastatine, een cholesterol verlagend medicijn, indien voorgeschreven bij patiënten voor hoog-risico operaties. Zoals beschreven, is fluvastatine gebruik veilig bij patiënten die een hoog-risico operatie ondergaan, en reduceert het zelfs het risico op peroperatieve hartschade, dankzij de niet-cholesterol gerelateerde effecten, ookwel pleitrope effecten, van deze medicijnen. Statines lijken ontstekingsmediatoren te remmen, waardoor de stabiliteit van verkalkingen in de kransslagaderen wordt bevorderd, en daarmee het risico wordt verlaagd op het losscheuren van deze laesies en op een verslechterde doorbloeding van de hartspier. Het starten van statines in aanloop naar een hoog-risico operatie lijkt aan te raden, en het staken van statines wordt afgeraden. Daarom genieten langwerkende statines de voorkeur, om de periode te overbruggen dat patiënten na een operatie geen orale medicatie kunnen innemen.

Naast stabiliteit van bestaande aderverkalkingen is het afzwakken van de operatiegerelateerde stress-response belangrijk, om het risico op een discrepantie tussen vraag en aanbod van zuurstof in de hartspier te verlagen. Dit is het best te bereiken met bètablokker therapie, zoals al vaak in de literatuur beschreven. Historisch gezien was er een aarzeling om bètablokkers te gebruiken rond operaties, vooral in patiënten met perifeer vaatlijden, vanwege hypothetische bijwerkingen en verslechterde uitkomsten. In de laatste tien jaar is steeds meer duidelijk geworden dat bètablokkers het risico verkleinen op peroperatieve hartschade, hartinfarcten en hartdood, zoals besproken wordt in **Hoofdstuk 3**. Wanneer men tenminste een week voor de operatie start, laag doseert en op geleide van de hartslag titreert blijven de voordelen van bètablokkade bestaan, terwijl de bijwerkingen tot een minimum worden beperkt.

In een voortdurende strijd om cardiovasculaire uitkomsten bij (vaatchirurgische) patiënten te verbeteren is onderzoek vaak gericht op nieuwe risicomarkers, biomarkers of onverwachte risicofactoren. Hierdoor zou risicoinschatting kunnen verbeteren en tot nieuwe aanknopingspunten leiden voor (secundaire) preventie. Vanuit dat oogpunt neemt de interesse in de bloedspiegel van vitamine D toe. Naast de welbekende effecten op de calciumhuishouding en botmetabolisme is er opstapelend bewijs dat een vitamine D tekort belangrijke niet-skeletale effecten heeft, gerelateerd aan het hart- en vaatstelsel. Klinische gegevens suggereren dat een vitamine D tekort bijdraagt aan aderverkalking. Het is alleen niet bekend of dit een direct effect is van vitamine D op de vaatwand, of het resultaat van een stijging van risicofactoren die aan vitamine D tekorten gerelateerd zijn, zoals een hoge bloeddruk, overgewicht en suikerziekte. In Hoofdstuk 4 hebben we de vitamine D status geïnventariseerd in een grote populatie patiënten met vernauwd of verwijd vaatlijden, en dit vervolgens gerelateerd aan bekende klinische risicofactoren en enkele directe tekenen van de ernst van het vaatlijden. We tonen aan dat een laag vitamine D gehalte een indicator is voor de ernst van vaatlijden, onafhankelijk van traditionele risicofactoren, en ongeacht het type vaatlijden, verwijd of vernauwd.

Hoewel er nog veel discussie gaande is over de vereiste bloedspiegel voor vitamine D wat betreft deze toepassing, zou kunnen worden verondersteld dat primaire en secundaire preventie om hart- en vaatziekten terug te dringen zich moeten focussen op de vitamine D status, naast bloeddruk, cholesterol en suiker control, gewichtsverlies en leefstijl veranderingen. Studies in de toekomst zouden de effecten van vitamine D supplementen op hart- en vaatziekten kunnen testen, en misschien de biologie achter vitamine D uitdiepen, in relatie tot de vaatwand.

Een laatste grote oorzaak voor morbiditeit en mortaliteit na vaatchirurgie is het van hartritmestoornissen. meest frequente optreden De peroperatieve hartritmestoornis is nieuw-ontstane boezemfibrilleren (AF), een bekende grote risicofactor voor postoperatieve beroertes, hartinfarcten en longembolieën. Sommige van deze complicaties zouden kunnen worden voorkomen, aangezien AF kan worden behandeld als het wordt gediagnosticeerd. De huidige standaard van Holter opnames of periodieke hartfilmpjes kunnen niet altijd deze (intermitterende) ritmestoornissen identificeren. Zoals in Hoofdstuk 5 staat gedrukt is de daadwerkelijke incidentie van AF na vaatchirurgie veel hoger (meer dan 25%) dan wordt geschat met de reguliere Holter opnames, waardoor de meeste gevallen onopgespoord en onbehandeld blijven. In implanteerbare hartritmemonitor, die gedurende weken of maanden continue ritmestoornissen kan evalueren, kan op betrouwbare wijze patiënten identificeren met een verhoogd risico op trombo-embolische complicaties, waardoor nieuwe behandelmogelijkheden voor hen beschikbaar zijn. Wij kunnen aanbevelen dat toekomstig onderzoek zich richt op het ontwikkelen van een behandelstrategie voor deze specifieke patiëntengroep.

Technische overwegingen in de vaatchirurgie

Recente en breed ingevoerde technische ontwikkelingen in de vaatchirurgie hebben een breed scala aan therapeutische opties opgeleverd in de individuele patiënt. De introductie van endovasculaire technieken hebben het arsenaal van behandelwijzen voor de vaatchirurg uitgebreid. Hoewel sommige van deze ontwikkelingen gelijk toepasbaar en een uitkomst leken indertijd, worden sommige lessen pas geleerd, lang nadat de eerste patiënten met deze nieuwe technieken werden behandeld. De ontwikkeling van een stent voor in de halsslagader, bijvoorbeeld, vergemakkelijkte de behandeling van halsslagader stenose in patiënten met een moeilijke anatomie of ernstige co-morbiditeit. Een veelvoud aan fabrikanten hebben diverse stents op de markt gebracht, allen met verschillende eigenschappen. Adequate stent-selectie kan echter alleen maar worden voorafgegaan aan objectieve vergelijking van deze eigenschappen, iets wat niet vaak is gebeurd. In Hoofdstuk 6 is gepoogd om een onderschatte, maar kritieke eigenschap van zelf-ontplooiende stents te vergelijken: de radiale kracht. De resultaten laten zien dat er significante verschillen bestaan tussen verschillende stenttypes, en dat de lengte van de laesie een belangrijke rol hierin speelt. Het klinisch succes van stenten van de halsslagader zou kunnen worden verbeterd als de juiste stent wordt geselecteerd voor de individuele patiënt, en de radiale kracht is een belangrijke eigenschap om rekening mee te houden. In toekomstige studies zou de invloed van de radiale kracht op de doorgankelijkheid in verschillende typen laesies moeten worden onderzocht in klinisch verband.

Een andere belangrijke technische ontwikkeling in de vaatchirurgie is de introductie van endovasculaire stentgrafts voor de behandeling van het aneurysma van de aorta. Endovasculair aneurysma herstel (EVAR) is wijdverbreid, en hiermee is ook een breed scala van stentgraft geïntroduceerd. Een belangrijke eigenschap in de samenstelling van de verschillende producten is de kunststof prothesewand, welke meest gebruikelijk van gewoven polyester of van een polymeer genaamd ePTFE wordt gemaakt. Terwijl het gebruik van stentgrafts toenam, steeg ook het aantal beschreven griepachtige symptomen in patiënten na implantatie. In **Hoofdstuk 7** wordt de incidentie van het post-implantatie syndroom gerapporteerd in meer dan honderd en vijftig patiënten die middels EVAR zijn behandeld. Ongeveer een derde van de patiënten ontwikkelde het post-implantatie syndroom, een samenstelling van koorts en inflammatie, en een sterke associatie werd waargenomen met geweven polyester, ook na multivariabele correctie. Hoewel de lange termijn consequenties van het post-implantatie syndroom onduidelijk zijn kunnen deze resultaten de ontwikkeling van toekomstige stentgrafts beïnvloeden. Ena ander aspect van EVAR is de aanpak van zijtakken. Tijdens open chirurgie worden de meeste zijtakken, zoals naar de ruggenwervels, onderbonden. Bij EVAR blijven deze onbehandeld, met de kans op het terug lekken van bloed, de zogenoemde type II *endoleaks*. Een controversiële ingreep in het geval van groei van de aneurysmazak na EVAR is het endoscopisch klemmen van de zijtakken en het openknippen of *fenestreren* van de groeiende aneurysmazak. In **Hoofdstuk 8** wordt de lange termijn follow-up beschreven van acht patiënten die een dergelijke procedure hebben ondergaan. De uitkomst was zeer wisselend en onvoorspelbaar. Bij een aantal patiënten

persisteerde een *endoleak* of keerde het terug, en in een aantal patiënten was hernieuwde groei van de aneurysmazak de aanleiding voor later herstel middels een open buikoperatie. Derhalve lijkt deze procedure niet aan te raden in de behandeling van groei van een aneurysma na een endovasculaire behandeling.

Onze aanpak was niet de enige poging om tot de optimale behandelstrategie te komen voor type II *endoleaks*. Ondanks de vele publicaties, met veelal conflicterende data over de ontstaansgeschiedenis van type II *endoleaks*, is er geen consensus over de drempel voor behandeling van een type II *endoleak* en er bestaat controverse over de optimale behandelwijze. In de literatuur, welke in **Hoofdstuk 9** wordt besproken, lijkt de strategie om behandeling te reserveren voor patiënten die een persisterend type II *endoleak* én ook nog groei van het aneurysma hebben de meeste steun te hebben onder vaatchirurgen. Wij onderbouwen dat groei van een aneurysma zou moeten lijden tot een zorgvuldige zoektocht naar een meer hoge druk *endoleak* dan slechts een type II, wat meestal lijdt tot een spoedige behandeling van de ware oorzaak van de groei. In afwezigheid van een dergelijke verklaring zouden de chirurg en de patiënt kunnen berusten in een afwachtende houding met nauwkeurige observatie, aangezien een ruptuur van het aneurysma onwaarschijnlijk is.

LIST OF PUBLICATIONS

Peer reviewed articles for this thesis

- 1. Voûte MT, Winkel TA, Poldermans D. Optimal medical management around the time of surgery. *Heart. 2010;96:1842-1848.*
- 2. Voûte MT, Winkel TA, Poldermans D. Safety of fluvastatin in patients undergoing high-risk noncardiac surgery. *Expert Opin Drug Saf. 2010;9:793-800.*
- Ravensbergen NJC, Voûte MT, Poldermans D. Safety of perioperative beta-blocker use: how do β-blockers compare in terms of side effects? *Expert Opin Drug Saf* 2011;10:545-558
- 4. van de Luijtgaarden KM, **Voûte MT**, Hoeks SE, Bakker EJ, Chonchol M, Stolker RJ, Rouwet EV, Verhagen HJM. Vitamin D deficiency may be an independent risk factor for arterial disease. *Eur J Vasc Endovasc Surg.* 2012;44:301-306
- 5. **Voûte MT**, Winkel TA, Hoeks SE, Plokker HWM, Stolker RJ, Verhagen HJMM. The incidence of new-onset paerioperative arrhythmias in major vascular surgery may be much higher than expected. *Submitted*.
- 6. **Voûte MT**, Hendriks JM, Van Laanen JH, Pattynama PM, Muhs BE, Poldermans D, Verhagen HJM. Radial force measurements in carotid stents: influence of stent design and length of the lesion. J Vasc Interv Radiol. *2011;22:661-666.*
- Voûte MT, Bastos Gonçalves FM, Van de Luijtgaarden KM, Klein Nulent CGA, Hoeks SE, Stolker RJ, Verhagen HJM. Stentgraft composition plays a material role in postimplantation syndrome. *J Vasc Surg. 2012;56:1503-1509*
- Voûte MT, Bastos Gonçalves FM, Hendriks JM, Metz R, Van Sambeek MRHM, Muhs BE, Verhagen HJM. Treatment of post-implantation aneurysm growth by laparoscopic sac fenestration: long-term results. *Eur J Vasc Endovasc Surg.* 2012;44:40-44
- Mees B, Voûte M, Bastos Gonçalves F, Mota Capitão L, Verhagen H. Intervention for type II endoleaks? "Primum non nocere": appraisal for the conservative management of low-pressure endoleaks after endovascular aneurysm repair. J Cardiovasc Surg (Torino). 2013;54:477-484

Peer reviewed articles outside this thesis

- Bakker EJ, Valentijn TM, van de Luijtgaarden KM, Hoeks SE, **Voûte MT**, Goncalves FB, Verhagen HJM, Stolker RJ. Type 2 diabetes mellitus, independent of insulin use, is associated with an increased risk of cardiac complications after vascular surgery. *Anaesth Intensive Care. 2013;41:584-590.*
- Valentijn TM, Hoeks SE, Bakker EJ, **Voûte MT**, Chonchol M, van de Luijtgaarden KM, Verhagen HJM, Stolker RJ. Influence of aortic valve calcium on outcome in patients undergoing peripheral vascular surgery. *Am J Cardiol.* 2012;110:1195-1199
- Voûte MT, Bastos Gonçalves F, Verhagen HJM. Commentary on 'ADSORB: a study on the efficacy of endovascular grafting in uncomplicated acute dissection of the descending aorta'. *Eur J Vasc Endovasc Surg. 2012;44:37*
- Bastos Gonçalves F, Voûte MT, Hendriks JM, Verhagen HJM. Muscle over mind? Eur J Vasc Endovasc Surg. 2012;43:613
- Bastos Gonçalves F, Jairam A, Voûte MT, Moelker AD, Rouwet EV, ten Raa S, Hendriks JM, Verhagen HJM. Clinical outcome and morphologic analysis after endovascular aneurysm repair using the Excluder endograft. J Vasc Surg. 2012;56:920-928
- Bastos Gonçalves FM, **Voûte MT**, Hoeks SE, Chonchol MB, Boersma HE, Stolker RJ, Verhagen HJM. Calcification of the abdominal aorta as an independent predictor of cardiovascular outcome: a meta-analysis. *Heart.* 2012;98:988-994
- Voûte MT, Bastos Gonçalves FM, Verhagen HJM. Comments regarding "The Wonders of New Available Post-Analysis CT Software in the Hands of Vascular Surgeons". *Eur J Vasc Endovasc Surg. 2012;43:407*
- Mastenbroek MH, Hoeks SE, Pedersen SS, Scholte Op Reimer WJ, Voute MT, Verhagen HJM. Gender Disparities in Disease-specific Health Status in Postoperative Patients with Peripheral Arterial Disease. *Eur J Vasc Endovasc Surg.* 2012;43:433-440
- Bakker EJ, Ravensbergen NJ, Voûte MT, Hoeks SE, Chonchol MB, Klimek M, Poldermans D. A randomized study of perioperative esmolol infusion for haemodynamic stability during major vascular surgery; rationale and design of DECREASE-XIII. *Eur J Vasc Endovasc Surg. 2011;42:317-23.*
- Van Kuijk JP, Flu WJ, Valentijn TM, Chonchol M, Voûte MT, Kuiper RJ, Verhagen HJM, Bax JJ, Poldermans D. Preoperative left ventricular dysfunction predisposes to postoperative acute kidney injury and long-term mortality. J Nephrol. 2011;24:764-70
- Ravensbergen NJ, Voûte MT, Poldermans D. Safety of perioperative β-blocker use: how do β-blockers compare in terms of side effects? *Expert Opin Drug Saf.* 2011;10:545-558.

- Winkel TA, Rouwet EV, van Kuijk JP, **Voûte MT**, de Melis M, Verhagen HJM, Poldermans D. Aortic surgery complications evaluated by an implanted continuous electrocardiography device: a case report. *Eur J Vasc Endovasc Surg. 2011;41:334-336*.
- Winkel TA, **Voûte MT**, de Melis M, Hoeks SE, Schouten O, Kessels R, Verhagen HJM, Poldermans D. Sudden death during follow-up after new-onset ventricular tachycardia in vascular surgery patients. *Vasc Surg. 2011;53:732-737;*
- Winkel TA, Schouten O, Hoeks SE, **Voûte MT**, Chonchol M, Goei D, Flu WJ, Van Kuijk JP, Lindemans J, Verhagen HJM, Bax JJ, Poldermans D. Prognosis of vascular surgery patients using a quantitative assessment of troponin T release: is the crystal ball still clear? *Eur J Vasc Endovasc Surg. 2010;40:739-746.*
- Van Kuijk JP, Voûte MT, Flu WJ, Schouten O, Chonchol M, Hoeks SE, Boersma EE, Verhagen HJM, Bax JJ, Poldermans D. The efficacy and safety of clopidogrel in vascular surgery patients with immediate postoperative asymptomatic troponin T release for the prevention of late cardiac events: Rationale and design of the Dutch Echocardiographic Cardiac Risk Evaluation Applying Stress Echo-VII (DECREASE-VII) trial. Am Heart J. 2010;160:387-393.
- Winkel TA, Schouten O, **Voûte MT**, Hoeks SE, Welten GM, Bax JJ, Verhagen HJM, Poldermans D. The effect of statins on perioperative events in patients undergoing vascular surgery. *Acta Chir Belg. 2010;110:28-31.*
- Flu WJ, Van Kuijk JP, **Voûte MT**, Kuiper R, Verhagen HJM, Bax JJ, Poldermans D. Asymptomatic low ankle-brachial index in vascular surgery patients: A Predictor of Perioperative myocardial damage. *Eur J Vasc Endovasc Surg. 2010;39:62-69.*
- Van Kuijk JP, Flu WJ, Witteveen OP, Voûte MT, Bax JJ, Poldermans D. The influence of statins on the expansion rate and rupture risk of abdominal aortic aneurysms. *J Cardiovasc Surg (Torino). 2009;50:599-609.*
- Van Kuijk JP, Flu WJ, Voûte MT, Poldermans D, Schouten O. Asymptomatic perioperative cardiac damage: long-term prognosis. *Future Cardiol. 2009;5:417-420.*

Book chapter

• Voûte MT, Akkersdijk GP, Hendriks JM, Vrancken Peeters MPFM, Pattynama PMT, Verhagen HJM. Thrombolysis for acute and chronic arterial occlusion. *Innovative Cardiovascular Procedures, Branchereau A, Jacobs MJHM, ed. 2009*

Oral presentations

- 2014 Vascular Rounds, Rotterdam
- 2011 Vascular Annual Meeting, Chicago (two presentations)
- 2011 Vaatdagen, Noordwijkerhout
- 2010 Stafdag Heelkunde Erasmus MC, Rotterdam (winner 'Zilveren Kreeft')
- 2010 Vascular Annual Meeting, Boston
- 2010 Chirurgendagen, Veldhoven (four presentations)
- 2010 Vaatdagen, Noordwijkerhout
- 2009 Stafdag Heelkunde Erasmus MC, Rotterdam
- 2009 European Society of Vascular Surgery Annual Meeting, Oslo
- 2009 Vascular Annual Meeting, Denver
- 2008 Stafdag Heelkunde Erasmus MC, Rotterdam
- 2008 Najaarsvergadering Nederlandse Verenging voor Vaatchirurgie, Zeist

ACKNOWLEDGEMENTS

"Praise the bridge that carried you over." - George Colman

Professor Hence Verhagen, de laatste keer dat ik dat tegen je zeg, beloofd! Vanaf mijn eerste stapjes in het onderzoek tot de afronding van dit proefschrift heb ik met je mogen samenwerken, en daar ben ik dankbaar voor. Het is niet altijd gemakkelijk geweest, maar we hebben weten door te zetten. Je bijdrage aan dit proefschrift is essentieel geweest, maar je vriendschap net zo belangrijk. Ik hoop onze samenwerking nog jaren te continueren.

Professor Robert Jan Stolker, heel hartelijk dank voor de manier waarop je hebt geholpen met de meest belangrijke projecten uit dit proefschrift. We hebben in korte tijd intensief samengewerkt, en ik heb veel van je opgestoken.

Professor Frans Moll, de man om wie het allemaal begonnen is! Als je me als 14-jarige niet in alle vroegte naar het St. Antonius had meegenomen om een fem-pop, een rood schort en een rokerig koffiekamertje te zien, dan was ik wellicht nooit gevallen voor de chirurgie.

Professor Huug Tilanus, het is me dan toch gelukt. Ik weet dat je me hier al tien jaar over achter de broek aan zit, maar ik zal uiteindelijk promoveren. Dank voor je aanstekelijke enthousiasme voor de chirurgie, en voor je oprechte betrokkenheid bij mij persoonlijk.

Overige leden van de kleine en grote promotiecommissie, dank voor uw wetenschappelijke bijdrage en kritische beoordeling van dit proefschrift. Het is mij een eer met u hierover van gedachten te wisselen.

Mijn paranimfen, Frederico en Koen die een belangrijke rol vertolkten in de voorbereiding op mijn verdediging, niet alleen inhoudelijk maar ook logistiek. Veel dank mannen! Oh, and Fred, just because you refuse to learn Dutch: thanks!

Gezien bovenstaande uitspraak hoort hier ook een dankwoord aan Don Poldermans, vanwege zijn charme, klinisch inzicht en ongelooflijke drive. Don, ik betreur je absentie, maar je weet dat ik jou én Virginie dankbaar ben.

De verschillende co-auteurs zonder wie dit proefschrift niet tot stand zou zijn gekomen, en in het bijzonder dr. Thijs Plokker voor zijn cruciale bijdrage aan het ontbrekende puzzelstukje van dit proefschrift.

Alle 'DECREASE' collega's, maar ook mijn dierbare kamergenoten en buren in de Z-flat, mede-onderzoekers tegen wil en dank en de overige onderzoekers en lab-ratjes van de heelkunde, heel veel dank voor de mooie jaren!

Joke Hendriks, net als jij ben ik ooit begonnen met een krakkemikkige proefopstelling maar een valide vraagstelling. Je was bij mijn co-schappen al een betrokken tutor en je hebt me tot nu toe altijd fantastisch geadviseerd.

Ook de medewerkers op de verschillende chirurgische afdelingen en de poliklinieken, niet alleen in het Erasmus MC maar ook het Maasstad Ziekenhuis en het IJsselland Ziekenhuis, dank ik voor de prettige samenwerking.

De chirurgen en arts-assistenten in het Maasstad Ziekenhuis, voor de gezelligheid, de leerzame tijd, en natuurlijk de kans die daar me is geboden om mijn eerste stappen in de kliniek te zetten.

Alle IJsvogels, en in het bijzonder dr. Dawson. Beste Imro, dank dat je in me hebt geïnvesteerd de afgelopen jaren. Ik hoop dat ik dit vertrouwen niet geschaad heb, en vind het geweldig dat je ook een rol op je hebt genomen bij mijn verdediging.

Ik dank ook mijn familie en vrienden, en in het bijzonder wil ik mijn lieve ouders en twee broers bedanken. Jullie zijn onmisbaar. Niet alleen in mijn persoonlijke leven en dat van mijn jonge gezin, ook om mee te sparren als ik in mijn werk een moeilijke horde moest nemen.

Tot slot mijn lieve Demet, die me energie geeft, mij scherp houdt, en me gelukkig maakt. Jij was niet alleen de afgelopen jaren een trouwe supporter, ook heb je me vaak vooruit geholpen met je eigenzinnige invalshoek en frisse blik. Dank voor alles wat jij en de kinderen voor me betekenen!

Canlarım, sizi çok seviyorum!

158 Appendices



PhD PORTFOLIO

Summary of PhD training and teaching activities

PhD student: Department: Research School:	Michiel Thomas Voûte Vascular Surgery / Anaesthesiology COEUR	PhD Period: 2009-2014 Promotores: H.J.M. Verhagen & R.J. Stolker	
1. PhD Training			
Courses		Year	ECTS
 CPO; Good Clinical Practice 		2010	1.5
 NIHES; Introduction to Data-analysis 		2010	1.0
 COEUR; Cardiovascular pharmacology 		2010	1.5
- NIHES; Regression Analysis for Clinicians		2011	1.9
Seminars & Worksh	nons		
- Journal club, Research meetings, Vascular Rounds		2009-2014	57
		2003 2014	0.8
COLON THE Day		2011	0.0
Presentations			
- National lectures		2009-2011	2.8
- International lectures		2009-2011	3.5
Symnosia & Meetin	105		
- National conferences		2009-2011	21
- International conferences		2009-2011	7.5
		2005 2011	7.5
2. Teaching			
Supervising		Year	ECTS
 BSc students at the Technical University Delft 		2009	3.0
 First aid for medical student – as examinator 		2010	0.5
- MSc Medical students		2010	1.2
Other activities			
- Organization COEUR PhD Day		2011	1.5

CURRICULUM VITAE

Michiel Thomas Voûte is op 4 oktober 1981 geboren te Amsterdam. Nadat hij op veertienjarige leeftijd een vaatoperatie met Professor Frans Moll bijwoonde was hij verkocht aan de vaatchirurgie. Na het behalen van zijn gymnasium diploma studeerde hij Spaanse Taal aan de Universiteit van Barcelona.

In 2000 begon hij met de studie Bouwkunde aan de Technische Universiteit in Delft. Dit leverde hem een rijk studentenleven in Delft op, waar hij ook zijn latere echtgenote leerde kennen. Overigens ontstond de klik pas nadat zij hem de hoofdrol zag vertolken in Oscar Wilde's "An Ideal Husband"...

Al snel stapte hij over naar Geneeskunde aan de Erasmus Universiteit Rotterdam. Tijdens zijn doctoraal onderzoek op de afdeling Vaatchirurgie zette hij een samenwerking op met de vakgroep Biomechanische Wetenschappen aan de TU Delft.

In 2009 behaalde hij zijn artsexamen en startte hij met zijn promotieonderzoek bij de afdelingen Vaatchirurgie (Prof.dr. H.J.M. Verhagen) en Anesthesiologie (Prof.dr. R.J. Stolker), wat resulteerde in dit proefschrift.

In 2012 deed hij klinische ervaring op in het Maasstad Ziekenhuis in Rotterdam, waarna hij op 1 januari 2013 is gestart met zijn opleiding tot chirurg in het IJsselland Ziekenhuis, onder supervisie van dr. I. Dawson en dr. B.P.L. Wijnhoven.

Naast zijn werk besteed Michiel zijn tijd het liefst aan golf en koken (vooral toetjes) en natuurlijk zijn vrouw Demet en hun kinderen. Bovendien werkt hij al jaren aan zijn eerste plaat, welke nog altijd niet op de radio is gedraaid...